Online ISSN : 2249-4588 Print ISSN : 0975-5853 DOI : 10.17406/GJMBR

# GLOBAL JOURNAL

OF MANAGEMENT AND BUSINESS RESEARCH: C

## Finance

Firm Value Effect

**Corporates Acquisition Programs** 



Capital Structure and Financial

Macroeconomic and Firm Specific

## Discovering Thoughts, Inventing Future

VOLUME 16 ISSUE 7 VERSION 1.0

they -

by Global Journal of Management and Business Research, USA



Global Journal of Management and Business Research: C Finance

## Global Journal of Management and Business Research: C

Volume 16 Issue 7 (Ver. 1.0)

**OPEN ASSOCIATION OF RESEARCH SOCIETY** 

## © Global Journal of Management and Business Research. 2016.

All rights reserved.

This is a special issue published in version 1.0 of "Global Journal of Science Frontier Research." By Global Journals Inc.

All articles are open access articles distributed under "Global Journal of Science Frontier Research"

Reading License, which permits restricted use. Entire contents are copyright by of "Global Journal of Science Frontier Research" unless otherwise noted on specific articles.

No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopy, recording, or any information storage and retrieval system, without written permission.

The opinions and statements made in this book are those of the authors concerned. Ultraculture has not verified and neither confirms nor denies any of the foregoing and no warranty or fitness is implied.

Engage with the contents herein at your own risk.

The use of this journal, and the terms and conditions for our providing information, is governed by our Disclaimer, Terms and Conditions and Privacy Policy given on our website <u>http://globaljournals.us/terms-and-condition/</u> <u>menu-id-1463/</u>

By referring / using / reading / any type of association / referencing this journal, this signifies and you acknowledge that you have read them and that you accept and will be bound by the terms thereof.

All information, journals, this journal, activities undertaken, materials, services and our website, terms and conditions, privacy policy, and this journal is subject to change anytime without any prior notice.

Incorporation No.: 0423089 License No.: 42125/022010/1186 Registration No.: 430374 Import-Export Code: 1109007027 Employer Identification Number (EIN): USA Tax ID: 98-0673427

## Global Journals Inc.

(A Delaware USA Incorporation with "Good Standing"; **Reg. Number: 0423089**) Sponsors: Open Association of Research Society Open Scientific Standards

#### Publisher's Headquarters office

Global Journals<sup>®</sup> Headquarters 945th Concord Streets, Framingham Massachusetts Pin: 01701, United States of America USA Toll Free: +001-888-839-7392 USA Toll Free Fax: +001-888-839-7392

## Offset Typesetting

Global Journals Incorporated 2nd, Lansdowne, Lansdowne Rd., Croydon-Surrey, Pin: CR9 2ER, United Kingdom

### Packaging & Continental Dispatching

Global Journals E-3130 Sudama Nagar, Near Gopur Square, Indore, M.P., Pin:452009, India

Find a correspondence nodal officer near you

To find nodal officer of your country, please email us at *local@globaljournals.org* 

#### *eContacts*

Press Inquiries: press@globaljournals.org Investor Inquiries: investors@globaljournals.org Technical Support: technology@globaljournals.org Media & Releases: media@globaljournals.org

Pricing (Including by Air Parcel Charges):

### For Authors:

22 USD (B/W) & 50 USD (Color) Yearly Subscription (Personal & Institutional): 200 USD (B/W) & 250 USD (Color)

## INTEGRATED EDITORIAL BOARD (COMPUTER SCIENCE, ENGINEERING, MEDICAL, MANAGEMENT, NATURAL SCIENCE, SOCIAL SCIENCE)

## John A. Hamilton,"Drew" Jr.,

Ph.D., Professor, Management Computer Science and Software Engineering Director, Information Assurance Laboratory Auburn University

## **Dr. Henry Hexmoor**

IEEE senior member since 2004 Ph.D. Computer Science, University at Buffalo Department of Computer Science Southern Illinois University at Carbondale

## Dr. Osman Balci, Professor

Department of Computer Science Virginia Tech, Virginia University Ph.D.and M.S.Syracuse University, Syracuse, New York M.S. and B.S. Bogazici University, Istanbul, Turkey

## Yogita Bajpai

M.Sc. (Computer Science), FICCT U.S.A.Email: yogita@computerresearch.org

## Dr. T. David A. Forbes

Associate Professor and Range Nutritionist Ph.D. Edinburgh University - Animal Nutrition M.S. Aberdeen University - Animal Nutrition B.A. University of Dublin- Zoology

## Dr. Wenying Feng

Professor, Department of Computing & Information Systems Department of Mathematics Trent University, Peterborough, ON Canada K9J 7B8

## **Dr. Thomas Wischgoll**

Computer Science and Engineering, Wright State University, Dayton, Ohio B.S., M.S., Ph.D. (University of Kaiserslautern)

## Dr. Abdurrahman Arslanyilmaz

Computer Science & Information Systems Department Youngstown State University Ph.D., Texas A&M University University of Missouri, Columbia Gazi University, Turkey

## Dr. Xiaohong He

Professor of International Business University of Quinnipiac BS, Jilin Institute of Technology; MA, MS, PhD,. (University of Texas-Dallas)

## **Burcin Becerik-Gerber**

University of Southern California Ph.D. in Civil Engineering DDes from Harvard University M.S. from University of California, Berkeley & Istanbul University

## Dr. Bart Lambrecht

Director of Research in Accounting and Finance Professor of Finance Lancaster University Management School BA (Antwerp); MPhil, MA, PhD (Cambridge)

## Dr. Carlos García Pont

Associate Professor of Marketing IESE Business School, University of Navarra

Doctor of Philosophy (Management), Massachusetts Institute of Technology (MIT)

Master in Business Administration, IESE, University of Navarra

Degree in Industrial Engineering, Universitat Politècnica de Catalunya

## Dr. Fotini Labropulu

Mathematics - Luther College University of Regina Ph.D., M.Sc. in Mathematics B.A. (Honors) in Mathematics University of Windso

## Dr. Lynn Lim

Reader in Business and Marketing Roehampton University, London BCom, PGDip, MBA (Distinction), PhD, FHEA

## Dr. Mihaly Mezei

ASSOCIATE PROFESSOR Department of Structural and Chemical Biology, Mount Sinai School of Medical Center Ph.D., Etvs Lornd University Postdoctoral Training,

New York University

## Dr. Söhnke M. Bartram

Department of Accounting and Finance Lancaster University Management School Ph.D. (WHU Koblenz) MBA/BBA (University of Saarbrücken)

## Dr. Miguel Angel Ariño

Professor of Decision Sciences IESE Business School Barcelona, Spain (Universidad de Navarra) CEIBS (China Europe International Business School). Beijing, Shanghai and Shenzhen Ph.D. in Mathematics University of Barcelona BA in Mathematics (Licenciatura) University of Barcelona

## Philip G. Moscoso

Technology and Operations Management IESE Business School, University of Navarra Ph.D in Industrial Engineering and Management, ETH Zurich M.Sc. in Chemical Engineering, ETH Zurich

## Dr. Sanjay Dixit, M.D.

Director, EP Laboratories, Philadelphia VA Medical Center Cardiovascular Medicine - Cardiac Arrhythmia Univ of Penn School of Medicine

## Dr. Han-Xiang Deng

MD., Ph.D Associate Professor and Research Department Division of Neuromuscular Medicine Davee Department of Neurology and Clinical Neuroscience Northwestern University

Feinberg School of Medicine

## Dr. Pina C. Sanelli

Associate Professor of Public Health Weill Cornell Medical College Associate Attending Radiologist NewYork-Presbyterian Hospital MRI, MRA, CT, and CTA Neuroradiology and Diagnostic Radiology M.D., State University of New York at Buffalo, School of Medicine and Biomedical Sciences

## **Dr. Roberto Sanchez**

Associate Professor Department of Structural and Chemical Biology Mount Sinai School of Medicine Ph.D., The Rockefeller University

## Dr. Wen-Yih Sun

Professor of Earth and Atmospheric Sciences Purdue University Director National Center for Typhoon and Flooding Research, Taiwan University Chair Professor Department of Atmospheric Sciences, National Central University, Chung-Li, Taiwan University Chair Professor Institute of Environmental Engineering, National Chiao Tung University, Hsinchu, Taiwan.Ph.D., MS The University of Chicago, Geophysical Sciences BS National Taiwan University, Atmospheric Sciences Associate Professor of Radiology

### Dr. Michael R. Rudnick

M.D., FACP Associate Professor of Medicine Chief, Renal Electrolyte and Hypertension Division (PMC) Penn Medicine, University of Pennsylvania Presbyterian Medical Center, Philadelphia Nephrology and Internal Medicine Certified by the American Board of Internal Medicine

## Dr. Bassey Benjamin Esu

B.Sc. Marketing; MBA Marketing; Ph.D Marketing Lecturer, Department of Marketing, University of Calabar Tourism Consultant, Cross River State Tourism Development Department Co-ordinator, Sustainable Tourism Initiative, Calabar, Nigeria

## Dr. Aziz M. Barbar, Ph.D.

IEEE Senior Member Chairperson, Department of Computer Science AUST - American University of Science & Technology Alfred Naccash Avenue – Ashrafieh

## PRESIDENT EDITOR (HON.)

Dr. George Perry, (Neuroscientist)

Dean and Professor, College of Sciences Denham Harman Research Award (American Aging Association) ISI Highly Cited Researcher, Iberoamerican Molecular Biology Organization AAAS Fellow, Correspondent Member of Spanish Royal Academy of Sciences University of Texas at San Antonio Postdoctoral Fellow (Department of Cell Biology) Baylor College of Medicine Houston, Texas, United States

## CHIEF AUTHOR (HON.)

**Dr. R.K. Dixit** M.Sc., Ph.D., FICCT Chief Author, India Email: authorind@computerresearch.org

## DEAN & EDITOR-IN-CHIEF (HON.)

## Vivek Dubey(HON.) MS (Industrial Engineering), MS (Mechanical Engineering) University of Wisconsin, FICCT Editor-in-Chief, USA editorusa@computerresearch.org Sangita Dixit M.Sc., FICCT Dean & Chancellor (Asia Pacific) deanind@computerresearch.org Suyash Dixit (B.E., Computer Science Engineering), FICCTT President, Web Administration and Development , CEO at IOSRD COO at GAOR & OSS

## **Er. Suyog Dixit** (M. Tech), BE (HONS. in CSE), FICCT SAP Certified Consultant CEO at IOSRD, GAOR & OSS Technical Dean, Global Journals Inc. (US) Website: www.suyogdixit.com Email: suyog@suyogdixit.com **Pritesh Rajvaidya** (MS) Computer Science Department California State University BE (Computer Science), FICCT Technical Dean, USA Email: pritesh@computerresearch.org Luis Galárraga J!Research Project Leader Saarbrücken, Germany

## Contents of the Issue

- i. Copyright Notice
- ii. Editorial Board Members
- iii. Chief Author and Dean
- iv. Contents of the Issue
- 1. The Firm Value Effect: Evidence from Egypt. 1-9
- 2. The French SMEs and their Financial Performance at the Launch of the Corporates Acquisition Programs. *11-17*
- 3. Macroeconomic and Firm Specific Determinants of Profitability of Insurance Industry in Ethiopia. *19-28*
- 4. Performance of Monthly Income Scheme in Mutual Fund Industry in India. 29-36
- 5. Capital Structure and Financial Performance of Insurance Industries in Ethiopia. *37-45*
- 6. Factors Affecting Loan Repayment Performance of Small Scale Enterprises Financed by Micro Finance Institutions: Study on Private Borrowers around Wolaita and Dawuro Zone. *47-55*
- v. Fellows
- vi. Auxiliary Memberships
- vii. Process of Submission of Research Paper
- viii. Preferred Author Guidelines
- ix. Index



Global Journal of Management and Business Research: c Finance

Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

## The Firm Value Effect: Evidence from Egypt

## By Omar Gharaibeh

Al Albayt University

*Abstract-* This paper investigates for a value effect in Egyptian firm returns using three different ways to determine value by sorting firms based on their past long-term returns (long-term contrarian), the book-to-market ratios (BE/ME), and the percentage changes in their BE/ME ratios (change). These three strategies are approaches commonly used to measure for value effect. Using sample period from January 1997 to April 2014, this study provides a strong evidence of an inter-firm value effect with three measures. The long-term return contrarian and BE/ME, produce significant abnormal raw returns of 2.18% and 2.01%, respectively. On the other hand, the percentage changes in their BE/ME provides weakly significant profits of 1.08% per month. This paper also shows that the value profits generated by all three alternative value strategies in Egyptian stock market can be explained by three-factor model.

Keywords: value effect, contrarian, three-factor model, egyptian stock market (EGX).

GJMBR - C Classification : JEL Code: D53

## THEFIRMVA LUEEFFECTEVIDENCEFROMEGYPT

Strictly as per the compliance and regulations of:



© 2016. Omar Gharaibeh. This is a research/review paper, distributed under the terms of the Creative Commons Attribution. Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

## The Firm Value Effect: Evidence from Egypt

Omar Gharaibeh

Abstract- This paper investigates for a value effect in Egyptian firm returns using three different ways to determine value by sorting firms based on their past long-term returns (long-term contrarian), the book-to-market ratios (BE/ME), and the percentage changes in their BE/ME ratios (change). These three strategies are approaches commonly used to measure for value effect. Using sample period from January 1997 to April 2014, this study provides a strong evidence of an interfirm value effect with three measures. The long-term return contrarian and BE/ME, produce significant abnormal raw returns of 2.18% and 2.01%, respectively. On the other hand, the percentage changes in their BE/ME provides weakly significant profits of 1.08% per month. This paper also shows that the value profits generated by all three alternative value strategies in Egyptian stock market can be explained by threefactor model.

Keywords: value effect, contrarian, three-factor model, egyptian stock market (EGX).

#### I. INTRODUCTION

he empirical literature on the value effect has shown that BE/ME ratio can be used to predict future returns (Clifford S Asness, Moskowitz, & Pedersen, 2013; Chen, 2011; Dempsey, 2010; Fama & French, 1993). Studies that have examined the value effect have proved the persistence of this effect at the level of company, industry and international index level (Clifford S Asness, et al., 2013; Chen, 2011; Chou, Ho, & Ko, 2012; Dempsey, 2010; Fama & French, 1993; Gharaibeh, 2016; Lakonishok, Shleifer, & Vishny, 1994).

Although most previous empirical research studies on monthly value effect employ data either from developed stock markets or emerging stock markets, few from these previous studies have addressed the Arabic stock markets. Egypt is one of the most important Arabic stock markets. Egyptian stock market constitutes an increasing share of the Arabic stock portfolio Therefore, to the best of our knowledge; no such work has yet been done on the Egyptian stock market in any international literature. This paper mainly aims to investigate value effect in an Arabic stock market of developing country, namely Egypt.

In addition to the traditional methods used in previous studies to calculate the value effect which are long-term contrarian strategy and BE/ME ratio, this study is the first to suggest using the percentage change in the BM ratio as a third new method for identifying value. The results of this paper are easily summarized in three points. First, the current study shows the very existence of value effect in Egypt stock market. Second, among the alternative three value strategies, this paper reveals that long-term contrarian and BE/ME strategies provide the highest monthly average returns. In particular, previous two strategies produce abnormal raw returns of 2.18% and 2.01% respectively, while change BE/ME strategy generate only abnormal profits of 1.08% per month. Lastly, this paper finds that all three alternative value effects used in Egypt stock market can be explained by three factor model.

The rest of the current study is organized as follows. Section 2 reviews that literature related to the value effect, while Section 3 describes the data and outlines the portfolio construction for three alternative value strategies. Section 5 provides the main empirical results, and finally Section 6 concludes the chapter.

#### II. LITERATURE REVIEW

Pioneering work by Fama and French (1993) which is the three-factor model has attracted the attention of many academic researchers and practitioners, as it found that the CAPM does not provide an adequate explanation of realized returns. Employing Fama and French's (1993) procedure to construct risk factors, Simlai (2009) re-investigated whether the size and book-to-market factors affect on the performance of portfolio returns. Simlai (2009) found that both size and book-to-market ratios have a key role in interpreting the variation in stock returns over the period from July 1926 to June 2007.

Lakonishok, Shleifer, and Vishny (1994)(LSV) investigated the relative performance of value strategies and showed that they outperform the market. Their finding supported the result of Fama and French (1992) that value strategies provide high returns. However, Whilst Fama and French (1992) consider the profitability of value strategies by explaining that these strategies are fundamentally riskier, Lakonishok et al. (1994) regard their profitability as being the result of stock mispricing.

Dempsey (2010) investigate the role of the BE/ME ratio in the formation of stock returns. He investigated whether the BE/ME ratio should take into account "risk-based", not a "mispricing" explanation for share prices in the Australian markets. His work was motivated by the explanation of stock return performance suggested by the Fama and French threefactor model, and applied Peterkort and Nielsen's (2005) approach to explain the relationship between the BM variable and stock return. Dempsey (2010) confirms

Author: Al Albayt University. e-mail: omar\_k\_gharaibeh@yahoo.com

the previous results that stock returns are strongly related to the firm's book-to-market equity ratio. Furthermore, strong evidence suggests that this relationship stems from the BE/ME ratio's absorption of the conclusion of company leverage as a risk factor. In spite of the distinctive characteristics of the Australian stock market, these previous results are substantially consistent with the U.S. results of Fama and French (1993) and Peterkort and Nielsen (2005).

Chen (2011) examined the reason why the book-to-market effect increased in small stocks and decreased in large stocks. His analysis found that firms with short life expectations have high idiosyncratic volatility. Chou, Ho, and Ko (2012) claim that the bookto-market effect in the U.S. equity market is mostly an intra-industry phenomenon. In more recent study, Asness, Moskowitz and Pedersen (2013) examine value strategy returns for global stocks, currencies, equity indices, government bonds and commodities. They provide evidence of value effect in each asset class.

Hasan, Alam, Amin, & Rahaman (2015) examine whether the size and value effects can explain the inter-firm returns in Dhaka Stock Exchange (DSE) in Bangladesh. They show strong evidence of size and value effects. Small firms along with high BE/ME firms tend to provide higher average monthly returns than big firms along with low BE/ME firms. Hasan, Alam, Amin, & Rahaman (2015) also show that cross-section of expected return in DSE can be explained by three-factor model.

Using 18 emerging stock markets during the period 1990 – 2013, Cakici, Tang, & Yan (2016) examine the presence of value effect. Egypt market is not addressed in their study; they show that the value effect is existence in 17 emerging markets except Brazil. During the global financial crisis, Cakici, Tang, & Yan (2016) point out that value premium move increasingly and positively together across-market.

Next section describes the dataset and methodology used in this study, and then this study expands upon each of these results in some detail.

#### III. DATA AND METHODOLOGY

#### a) Data

This paper considers monthly stock returns, firm size (ME), and the firm book-to-market ratio (BE/ME) for 104 Egyptian firms of all firms listed in the Egyptian Exchange (EGX) for the period of January 1997 to May 2014. At present, a total 104 firms of different sectors are listed in EGX till May 2014. Monthly stock price data are downloaded from Data Stream. The current study use Egyptian Treasury bill rate (monthly average) as the proxy for risk free rate and collected from Jordan central Bank. MSCI index is used as the proxy for market portfolio and data are collected from Data Stream. Following Fama and French (1992), Egyptian firm's BM

ratio for June of year t is the book value of equity for the last fiscal year end in t-1 divided by the market value of equity as of December of t-1. A firm's annual BM ratio for June of year t is the average of the BM ratios of the firms. In the BM monthly portfolio sorts that follow, this annual firm BM ratio is used for the following 12 months. Table 1 details descriptive statistics over the period January 1997 through May 2014 for the Egyptian firms, demonstrating average monthly returns, standard deviation, Skewness and Kurtosis for each firm. Table 1 shows big difference in the mean and standard deviation of average returns. The South Valley Cement has the biggest monthly average (over 4% per month). In contrast, the Maridive & Oil Services has the lowest average at -104. The Egyptian firms have an average monthly return of 1.34% and an average standard deviation of 15.63%.

#### b) Portfolio Construction

This paper applies three alternative measures to determine value for each firm: the long-term return reversal by employing contrarian strategies, the firm's BM ratio, its 60-month past return, and the percentage change in its BE/ME ratio over the last 24, 36, 48 or 60 months. Using percentage change over the last 24, 36, 48, and 60 months allows testing the sensitivity of this new method to measure value to the same formation period. As a result this paper investigates three alternative value strategies: the long-term contrarian strategy, the BE/ME strategy and the change BE/ME strategy. The construction methodology for these strategies is presented in the next sections.

The portfolios for the three value strategies are formed as follows. At the beginning of each month *t*, the 104 firms are sorted based on their past BE/ME ratios (for the value strategy), on their 60-month past returns (for the contrarian strategy), and on the percentage changes in their BE/ME ratios over the past *J* months for J = 24, 36, 48 or 60months (for the change strategies). The high BE/ME, long-term winner and high change equal-weighted portfolios (denoted HV, LW and HC, respectively) contain the 25% of firms with the highest values for their respective sorting variables in the same way, the low BE/ME, long-term loser and low change portfolios (LV, LL and LC, respectively) contain the 25% of firms with the lowest values for their respective sorting variables.

The zero cost BE/ME strategy (HV-LV) is based on buying the high BE/ME portfolio and selling the low BE/ME portfolio. The zero cost long-term contrarian strategy (LL-LW) is longs the long-term loser portfolio and shorts the long-term winner portfolio. The zero cost change strategy (HC-LC) is buying the high change portfolio and selling the low change portfolio. Portfolios are held for *K*-month holding periods, while K = 1, 3, 6, 9and 12 months. For the long-term contrarian strategy, the current study keeps a 12-month gap between the end of the 60-month formation period and the beginning of the K-month holding period compatible with previous studies such as Fama and French (1996), Figelman (2007), Grinblatt and Moskowitz (2004) and Malin and Bornholt (2013). The reason for employing this process is that Fama and French (1996) show that omitting the first 12-month after the end of the formation period enhances the performance of long-term contrarian strategy because it avoids any long-term reversals being compensated by the short-term continuation of returns.

This process is compatible with DeBondt and Thaler's (1985) finding that the first 12-month of the holding period did not earn significant contrarian profits. For all other strategies in this paper, the current study adopts the common practice used in momentum studies of omitting 1-month between the end of the formation period and the beginning of the holding period. Whereas a gap of zero or 1-month makes no significant difference to the outcomes, a small gap makes achievement of trading strategies easier in the real world. In addition it avoids any concerns about microstructure biases.

#### Table 1 : Descriptive Statistics

Table 1 reports the descriptive statistics for 104 firm returns from January 1997 until April 2014, obtained from Datastream. The first column is the name of the firm. This is followed by the average monthly returns, the standard deviation of monthly returns, book-to-market ratios and finally the "Skew" is the skewness, and the "Kurt" is the kurtosis for each firm.

Firm Names	Average	SD	BE/ME	Skew	Kurt
South Valley Cement	4.53	26.24	1.27	3.90	27.58
Six of Oct.Dev.& Inv.	2.99	26.32	0.87	3.33	16.46
Egyptian Kuwaiti Holding	2.99	23.75	0.56	6.46	60.98
Egyptians Housing Dev.	2.94	24.30	0.98	3.51	16.83
Egyptians Abroad Invs.	2.88	25.19	1.10	2.21	7.38
Samad Misr -Egyfert	2.86	18.25	0.73	1.19	2.65
Faisal Islamic Bank Of Egypt Egp	2.86	17.50	1.67	4.30	28.94
Giza General Contracting	2.72	22.50	0.62	1.56	5.26
Global Telecom	2.71	18.63	0.68	2.14	10.30
United Arab Shipping	2.69	25.60	-0.85	1.93	6.21
El Ezz Porcelain (Gemma)	2.66	18.33	1.21	1.11	1.77
Orascom Construction Ind	2.62	11.87	0.38	0.00	1.01
Arab Ceramic	2.60	15.65	0.46	1.26	2.89
Misr Beni Suef Cement	2.55	11.41	0.57	1.00	3.00
United Housing & Dev.	2.44	16.58	0.27	1.16	3.70
Ezz Steel	2.41	18.91	0.69	0.84	1.51
Cairo Poultry	2.37	14.87	0.88	0.96	3.89
Heliopolis Housing	2.26	18.00	0.13	1.93	8.05
Misr Duty Free Shops	2.15	24.04	0.63	6.18	60.43
Acrow Misr	2.13	18.47	1.06	2.98	19.36
Piraeus Bank Egypt Dead - 19/03/10	2.13	19.82	0.78	2.65	11.43
Alexandria Cement	2.08	18.03	0.60	1.56	4.92
Helwan Cement Dead - 02/02/10	2.07	14.39	0.55	3.08	15.38
Egyptian Gulf Bank	2.04	16.11	0.68	2.75	31.47
Qatar National Bank Alahly	2.03	13.83	0.59	-0.12	9.81
Egyp.Co.for Mobl.Svs. (Mobinil)	2.02	15.25	0.21	1.88	7.31
Medinet Nasr Housing	1.98	17.31	0.26	1.37	5.73
Egyptian Electric Cable	1.89	28.26	3.34	8.64	103.10
Coml.Intl.Bank (Egypt)	1.86	11.25	0.57	0.74	1.71
Kafr El-Zait Pesticides	1.82	16.87	0.80	1.57	4.91
Orascom Hotels And Dev.	1.82	19.50	0.53	2.06	8.43
Sinai Cement	1.80	11.34	0.77	0.79	1.43
Nozha Intl.Hospital	1.78	14.94	0.68	1.81	11.46
Misr Cement (Qena)	1.72	8.86	0.34	2.18	9.33
Vodafone Egypt Telecom	1.71	15.08	0.25	2.73	28.29
Egyptian Finl.& Indl.	1.67	14.71	1.23	1.29	4.31
Development & Engr.	1.66	21.29	1.09	2.81	15.06
Orascom Hotel Holdings (Ohh)	1.62	19.02	1.36	2.25	13.40
Olympic Gp.Finl.Invs. Dead - 27/01/13	1.62	15.28	1.06	1.47	5.81
El Ahli Inv.& Dev.	1.62	20.67	0.78	2.32	11.05
Housing & Dev.Bank	1.56	16.88	1.48	1.88	8.48
El Ezz Aldk.Steel Alexa.	1.51	13.55	0.76	1.46	7.49

Eastern Tobacco	1.51	11.11	0.63	1.60	7.13
Elswedy Electric	1.41	12.93	0.63	0.18	1.55
National Dev.Bank	1.38	17.43	0.56	1.89	7.96
Alexandria Flour Mills	1.32	19.60	0.79	2.66	13.14
Arab Cotton Ginning	1.29	20.37	1.61	0.65	4.65
El Watany Bank Of Egypt	1.25	13.98	0.85	1.55	5.61
Bisco Misr	1.22	9.48	0.43	1.72	8.85
Alexandria Spng.& Wvg.	1.21	16.77	1.78	0.48	3.28
South Cairo & Giza Mls.& Bkrs.	1.21	18.35	1.03	2.28	9.17
Extracted Oils Derivatre	1.20	17.56	0.87	2.26	11.93
Middle Egypt Flour Mills	1.19	17.04	1.10	2.16	7.61
Abou Kir Fertilizers	1.16	10.75	0.33	2.17	16.09
National Cement	1.12	17.44	0.35	2.10	8.78
Egyptian Intl.Pharms. (Epico)	1.11	7.91	0.85	1.03	4.84
Credit Agricole Egypt	1.10	18.33	0.58	3.03	24.63
Upper Egypt Flour Mills	1.09	14.92	0.84	2.39	10.91
Ajwa For Food Inds.	1.09	23.40	0.68	5.50	46.60
Egypt Aluminium	1.09	14.01	1.21	1.37	4.15
Oriental Weavers	1.06	10.43	1.07	0.47	1.05
Export Dev.Bk.Of Egypt	1.02	17.40	1.37	2.98	29.69
Pyramisa Hotels	1.02	11.91	1.63	1.48	6.21
Mena Tourism & RIst.Inv.	1.01	18.39	0.85	1.34	3.87
Ameriyah Cement Dead - 22/06/10	0.97	11.36	0.47	1.81	7.73
El Nasr Clothes & Text. (Kabo)	0.92	16.55	1.83	1.01	3.54
Ntrl.Gas & Mng.Project (Egypt Gas)	0.92	13.93	1.17	3.08	21.74
Cairo Pharmaceuticals	0.90	11.42	1.11	3.41	30.51
Misr Chemical Industries	0.90	15.63	0.97	1.16	3.69
Namaa For Dev.&.Reit.Co. Dead -	0.88	19.59	1.08	1.43	8.12
Suez Cement	0.83	10.37	0.78	1.14	3.35
Ahli United Bank Egypt Dead -	0.82	10.27	0.53	2.52	21.14
Nile Cotton Ginning	0.78	19.49	0.99	2.26	19.09
Delta Insurance	0.78	12.34	0.93	1.30	5.23
Egyptian Strch.& Glucose	0.77	16.74	0.82	0.85	7.07
East Delta Flour Mills	0.75	12.92	0.94	1.84	8.99
Egypt American Bank Dead - 30/08/07	0.74	10.26	0.57	1.40	6.55
Blom Bank Egypt Dead - 16/10/10	0.74	15.97	1.10	0.23	10.24
Alexandria For Pharmacy	0.73	10.71	0.96	1.46	13.75
Mid.& Ws.Delt.Flr.Mls.	0.61	12.40	0.75	1.44	6.22
General Silos & Storage	0.60	16.84	0.96	3.70	25.92
Nile Pharmaceuticals	0.60	11.26	0.88	1.61	7.01
Torah Cement	0.58	10.55	0.51	0.71	2.88
Misr For Hotels (Hilton)	0.55	13.35	1.74	1.58	5.62
Palm Hills Devs.Sae	0.54	19.00	1.15	0.39	1.06
Raya Hldg.For Tech.& Comms.	0.54	14.50	1.27	0.56	2.58
Delta Industries (Ideal) Dead -	0.54	15.63	0.87	0.33	7.97
Paint & Chmid.(Pachin)	0.52	10.31	0.79	0.63	2.74
Misr Oil	0.48	13.37	0.80	1.14	5.38
Sidi Kerir Petrochem.	0.47	9.92	0.32	0.20	0.77
Amreyah Pharms.Inds Dead -	0.45	10.19	0.78	5.53	58.23
Memphis Pharmaceuticals	0.41	11.89	0.78	1.65	10.75
North Cairo Mills	0.41	15.95	0.81	2.65	15.22
Talaat Moustafa Group	0.39	13.81	2.30	0.30	0.83
Misr Intl.Bank (Mibank) Dead -	0.38	10.82	0.95	2.62	10.38
Delta Sugar	0.30	13.04	0.46	-0.17	17.41
Telecom Egypt	0.22	8.63	0.99	0.46	-0.03
Misr Conditioning (Miraco)	0.18	14.67	0.51	-0.76	13.79
Alexandria Mrl.Oils	0.16	8.86	0.46	-0.04	0.15
Suez Canal Bank	0.15	13.41	1.51	1.46	10.51
Egyptian Media Prdn.City	-0.24	16.14	1.72	0.80	1.60
Naeem Holding	-0.37	14.77	2.22	0.77	2.58
Al Arafa Inv.& Cnsl.	-0.85	8.78	1.31	0.03	-0.22
Maridive & Oil Services	-1.04	12.04	0.43	-0.40	0.42
Average	1.43	15.63			

#### IV. Results

This section analyses the findings of the various value strategies. The section includes a discussion of raw and risk-adjusted results. this section reports the average monthly holding period returns for the long, short and long-short portfolios of the long-term contrarian strategy in Table 2, the BE/ME strategy in Table 3 and the pure change BE/ME strategy in Table 4 when applied to the sample of 104 Egypt firms. Columns 3 through 7 in each Table list the equal-weighted average monthly returns in percentages for the *K*-month holding periods (K = 1, 3, 6, 9 and 12 months).

profits (LL-LW) are statistically significant over all *K*month holding periods if J = 24, 36, 48, or 60 months. Table 2 demonstrates significant long-term contrarian LL-LW profits. For example, for the 60-month (five-year) formation period case with a 6-month holding period (*K* = 6) case, the difference between the average monthly returns of the LL portfolio and the LW portfolio is large 2.18% per month and it is statistically significant (t-stat 2.84). In summary, there are large and significant longterm contrarian profits generated for long formation periods of 24, 36, 48 and 60 months.

#### a) Value strategies' results

Except for the J = 24 case over K = 1, the longterm contrarian results in Table 2 show that the strategy

#### Table 2 : Profitability of Long-Term Contrarian at Egypt Firms

Table 2 provides the average monthly holding period returns in percentages of the selling, buying, and selling minus buying portfolios of the long-term reversal strategy for 104 Egypt firms. Portfolios are constructed as follows: At the beginning of each month t, the 104 firms are sorted derived from their past J-month formation period returns for J = 24, 36, 48, and 60 months. The long-run loser equal-weighted portfolio (LL) comprises of the 25 % of portfolios with the lowest returns, and the long-term winner equal weighted portfolio (LW) comprises of the 25 % of portfolios with the largest returns. The strategy LL-LW buying the long-run loser portfolio and sells the long-run winner portfolio to be held for K = 1, 3, 6, 9, or 12 months. The *t*-statistics depends on the Newey and West (1987) adjustment for autocorrelation up to lag 11.

			Hold	ing Period Returns		
J	Portfolio	K=1	K=3	K=6	K=9	K=12
24	LW	0.88	0.89	0.87	0.82	0.91
		(1.58)	(1.62)	(1.55)	(1.45)	(1.59)
	LL	2.06	2.10	2.06	2.61	2.57
		(2.74)	(2.78)	(2.73)	(3.94)	(3.84)
	LL-LW	1.18	1.21	1.19	1.79	1.66
		(1.94)	(2.01)	(2.05)	(3.98)	(3.76)
36	LW	1.07	1.08	1.03	1.02	1.06
		(1.86)	(1.88)	(1.75)	(1.72)	(1.76)
	LL	2.77	2.48	2.41	2.43	2.39
		(3.93)	(3.6)	(3.52)	(3.52)	(3.43)
	LL-LW	1.70	1.39	1.39	1.40	1.33
		(3.17)	(2.79)	(2.88)	(2.97)	(2.87)
48	LW	1.23	1.33	1.28	1.23	1.16
		(1.89)	(2.04)	(1.96)	(1.88)	(1.72)
	LL	2.56	2.66	2.58	2.72	2.79
		(3.42)	(3.49)	(3.37)	(3.49)	(3.46)
	LL-LW	1.33	1.33	1.30	1.49	1.63
		(2.44)	(2.47)	(2.49)	(2.85)	(3)
60	LW	0.52	0.67	0.68	0.66	0.72
		(0.56)	(0.72)	(0.72)	(0.69)	(0.73)
	LL	2.82	2.99	2.86	2.86	2.86
		(3.15)	(3.37)	(3.32)	(3.25)	(3.22)
	LL-LW	2.31	2.31	2.18	2.19	2.14
		(2.78)	(2.85)	(2.84)	(2.83)	(2.56)

The BE/ME strategy results in Table 3 show clearly that the strategy profits (HV-LV) are statistically

significant over all K-month holding. For example, for the 6-month holding period (K=6) case, the difference

between the average monthly returns of the HV portfolio and the LV portfolio is large 2.01% per month (t-stat 4.35), which is statistically significant. In general, the holding period returns in Table 3 give strong evidence of BE/ME effect at the Egypt firm level.

Table 4 shows that the pure change strategy produces statistical significant and sometimes weakly significant profits for all K holding periods when the percentage change in the BM ratio is measured over 24, 36, 48 or 60 months. For example, when the percentage change in the BM ratio is calculated over the past 60 months, the high change portfolio (HC) provides an

average return of 2.36% per month while the low change portfolio (LC) produces an average return of only 1.29% per month with a six-month holding period. The difference of 1.08% per month between HC and LC is weakly significant (*t*-stat 1.65), and is economically large. On the other hand, measuring the percentage change in BE/ME ratios over 24, 36 or 48 months generates statistical significant profits and consistent results, with only the six-month holding period providing statistical significant profits (1.62%, 1.16% and 1.23%) per month (*t*-stat 2.89, 2.02 and 1.97), respectively.

#### Table 3 : Profitability of BE/ME at Egypt Firms

Table 3 provides the average monthly holding period returns in percentages of the buying, selling, and buyingselling portfolios for the BE/ME strategy applies to 104 Egypt firms. At the beginning of each month t from November 1994 to April 2014, the 104 firms are ranked based on their BE/ME, and are assigned to one of four portfolios. The high BE/ME equal-weighted portfolios (HV) comprises of the 25% of firms with the highest values, while the low BE/ME comprises of the 25% of firms with the lowest values. HV-LV refers to the buying the fourth portfolio and selling first portfolio. All reported returns are equally weighted. The strategy LL-LW longs the long-term loser portfolio and shorts the long-term winner portfolio to be held for K = 1, 3, 6, 9, or 12 months. The t-statistics are based on the Newey and West (1987) adjustment for autocorrelation up to lag 11.

	Holding Period Returns						
Portfolio	K=1	K=3	K=6	K=9	K=12		
HV	2.42	2.45	2.25	2.26	2.52		
	(3.48)	(3.65)	(3.36)	(3.3)	(3.7)		
LV	0.08	0.06	0.24	0.35	0.57		
	(0.12)	(0.09)	(0.39)	(0.56)	(0.92)		
HV-LV	2.50	2.40	2.01	1.91	1.95		
	(4.94)	(5.07)	(4.35)	(4.17)	(4.28)		

In short, the results in Table 2, 3 and 4 suggest that the three alternative measures of value provide high levels of profitability. In Table 1 and 2, strategy profits for the long-term contrarian and BE/ME strategies are significant and very similar for all holding periods. For example, the long-term contrarian strategy earns a significant 2.18% per month (*t*-stat 2.84) and the BE/ME strategy earns 2.01% per month (*t*-stat 4.35) with sixmonth holding periods (K=6). For change BE/ME strategy, although Table 4 shows that the change value

strategy provides weakly significant for the same period, it is still economically large. The change value generates monthly returns 1.08% per month (t-stat 1.65).

The post-formation behaviors of the value strategies' profits are also illustrated in Figure 1. Figure 1 depicts the post-formation cumulative returns of the long-term contrarian strategy (LL-LW) with J = 60, the BE/ME strategy (HV-LV), and the change BE/ME strategy (HC-LC) with J = 60 for the 60 months following the end of the formation period.

#### Table 4 : Profitability of Change BE/ME at Egypt Firms

This table reports the average monthly holding period returns in percentages of the long, short, and long-short portfolios for change strategy applied to 104 Firms. Portfolios are constructed as follows: At the beginning of each month t, the 104 industries are ranked based on their percentage changes in their BM ratios over the past *J* months for J = 24, 36, 48 and 132 months. The high change portfolios HC contains the 25% of firms with the largest change values, while the low change BM portfolio LC contains the 25% of firms with the lowest change values. The change BM strategy (HC LC) portfolios are held for K = 1, 3, 6, 9 or 12 months.

		Holding Period Returns					
J	Portfolio	K=1	K=3	K=6	K=9	K=12	
24	HC	2.84	3.02	2.74	2.76	2.71	
		(3.69)	(3.98)	(3.69)	(3.77)	(3.66)	
	LC	0.90	1.10	1.31	1.41	1.38	

		(1.38)	(1.7)	(2.03)	(2.15)	(2.1)
	HC-LC	2.06	1.87	1.62	1.40	1.49
		(3.25)	(3.14)	(2.89)	(2.7)	(2.96)
36	HC	2.57	2.53	2.48	2.33	2.25
		(3.44)	(3.46)	(3.39)	(3.18)	(3.07)
	LC	1.10	0.97	1.32	1.30	1.13
		(1.49)	(1.31)	(1.86)	(1.86)	(1.61)
	HC-LC	1.47	1.57	1.16	1.03	1.12
		(2.42)	(2.59)	(2.02)	(1.78)	(1.98)
48	HC	2.41	2.58	2.42	2.41	2.42
		(3.16)	(3.45)	(3.25)	(3.2)	(3.18)
	LC	1.23	1.16	1.18	1.10	1.30
		(1.63)	(1.54)	(1.56)	(1.48)	(1.74)
	HC-LC	1.18	1.41	1.23	1.32	1.12
		(1.74)	(2.18)	(1.97)	(2.22)	(1.9)
60	HC	2.35	2.34	2.36	2.57	2.74
		(2.77)	(2.9)	(2.91)	(3.16)	(3.29)
	LC	1.11	1.09	1.29	1.49	1.36
		(1.51)	(1.44)	(1.72)	(1.93)	(1.77)
	HC-LC	1.24	1.26	1.08	1.08	1.38
		(1.82)	(1.83)	(1.65)	(1.67)	(2.15)

Given the Figure 1, while the value strategies graph suggests a slowing in the cumulative returns towards the end of the 60 months we note that all alternative three value strategies generate positive cumulative returns. Long-term contrarian strategy provides the highest cumulative returns, then comes the BE/ME strategy. The change BE/ME strategy comes in the last strategy among alternative value strategies.





This graph presents the cumulative returns of the long-term return reversal portfolio LL-LW (with J = 60 months), BE/ME strategy HV-LV and change BE/ME (with J = 60 months) using non-overlapping portfolio (K = 1) for the 60 months after the end of the formation period.

#### b) Risk adjustments

To find whether the profits of these strategies could be considered a reward for bearing risk, the profits of the long-term contrarian, BE/ME and change

value strategies are risk-adjusted employing the Fama-French three-factor model. The three-factor regression model comprises of the market factor, a small minus big factor, and a value minus growth factor:  $R_{pt} - R_{ft} = \alpha_p + \beta_p (R_{mt} - R_{ft}) + s_p SMB_t + h_p HML_t + \varepsilon_{pt} , \quad (1)$ 

Where the dependent variable  $R_{pt} - R_{ft}$  is the monthly excess return of the strategy portfolio p,  $R_{pt}$  is the monthly return of portfolio p at time t, and  $R_{ft}$  represents the monthly risk-free rate at time t, represented by the one-month Egyptian T-Bill return. The independent variables or factors are as follows:

 $R_{mt} - R_{ft}$  is the Egyptian MSCI index's monthly excess market return for month *t*, while *SMB*<sub>t</sub> and *HML*<sub>t</sub> are the monthly size and book-to-market factors at time *t*, respectively.

The monthly return values for the three factors and one-month T-Bill risk-free rate covering the full sample period from January 1997 to May 2014 are downloaded from Data stream. The three-factor model covers the period from the period January 1997 to May 2014. The coefficients  $\beta_p$ ,  $s_p$  and  $h_p$  are the regression loadings corresponding to the factors of the models, while the intercept  $\alpha_p$  (or simply alpha) indicates to the risk-adjusted abnormal returns of the portfolios over the evaluation period. If alpha is statistically significant, then this is evidence of abnormal profits. The *t*-values corresponding to the regression coefficients are corrected for heteroskedasticity using White's (1980) test.

Table 5 reports the estimated regression coefficients of the three-factor model and the corresponding White-corrected *t*-statistics for the long, short and long-short portfolios for the long-term contrarian (J = 60), the BE/ME and the change value (J = 60) strategies with six-month holding periods (K = 6) in Panels A, B and C, respectively. Column 2 of Table 5 reports the monthly alphas of the three-factor model, while the last column lists the adjusted R<sup>2</sup>.

The alpha of the long-term contrarian long-short LL – LW portfolio in Panel A, B and C is small (0.013%, - 0.09 and -0.04 per month) and insignificant (*t*-stat 0.29, - 1.30 and -0.76), respectively.

In summary, the three alternative value results in Panels A, B and C of Table 5 reveal that there is value return in Egyptian firm returns that can be explained by the Fama-French three-factor model.. The insignificant long-term contrarian strategy's alpha is consistent with Fama and French's (1996) finding that the three-factor model can explain the reversal of long-term returns of individual U.S. stocks reported by DeBondt and Thaler (1985).

#### Table 5 : Risk-Adjusted inter-firm value Profits

This table presents the three-factor regression results for the contrarian, BE/ME and change BE/ME portfolios in Panel A, B and C respectively. These portfolios are described in Tables 2 and 3. The three-factor regression model is as follows:

$$R_{pt} - R_{ft} = \alpha_p + \beta_p (R_m - R_{ft}) + {}_{sp} SMB_t + {}_{hp} HML_t + \epsilon_{pt}$$

where  $R_{pt} - R_{ft}$  is the portfolio's excess return,  $R_{mt} - R_{ft}$  is the excess return on the market, and  $SMB_t$  and  $HML_t$  are the size and book-to-market factors. The t-statistics presented in parentheses are corrected for heteroskedasticity using White's (1980) test.

		Three-Factor N	lodel		
	α	β	S	h	Adj R2
Panel A:					
contrarian	0.013	0.041	-0.380	-0.093	28%
	(0.29)	(0.08)	(-5.39)	(-0.86)	
Panel B:					
BEME	-0.090	0.011	-0.017	1	100%
	(-1.3)	(1.36)	(-5.99)	(4.12)	
Panel C:					
CHBEME	-0.040	0.452	-0.205	0.632	12.8%
	(-0.76)	(0.77)	(-2.34)	(4.77)	

#### V. Conclusion

Arabic stock markets are clearly a significant part of the world portfolio today and therefore are important to the average investor. Finance literature has discovered important facts about value effect in US, as well as in the developed equity markets. Value effect is a lot less explored for emerging markets, especially Arabic market.

The current study provides results to fill this gap by considering stock returns in Egyptian stock market. Using sample period from January 1997 to April 2014, this paper has shown two main contributions: First, the result of this study provides strong evidence of value effect by using three alternative value strategies: long-term contrarian, BE/ME and change BE/ME strategies. More specific, the long-term contrarian and BE/ME value strategies provide abnormal returns more than 2% per month, while the change BE/ME value strategy generate abnormal returns more than 1% per month. Second, this paper constructs 4 portfolios based on each value strategy for Egypt stock market, and uses these portfolios as the returns in the three-factor model. This paper also finds that the size and value premium in addition to market risk premium have very strong power to explain cross-section of expected return in the Egyptian Exchange.

The participants of the stock market, e.g. investors and fund managers may be utilized using previous findings. The investors from developing countries like Egypt can achieve abnormal returns by using three alternative value measures. In addition, practitioners manage their portfolios and assess their assets more accurately through applying three-factor model. For future research, it would be attractive to examine whether volatility effect can shed some light on the Egypt value returns. None of the previous studies investigate the relationship between value returns with volatility effect in Egypt stock market.

#### **References** Références Referencias

- Asness, C. S. (1997). The interaction of value and momentum strategies. Financial Analysts Journal, 29-36.
- Asness, C. S., Moskowitz, T. J., & Pedersen, L. H. (2013). Value and momentum everywhere. The Journal of Finance, 68(3), 929-985.
- Cakici, N., Tang, Y., & Yan, A. (2016). Do the Size, Value, and Momentum Factors Drive Stock Returns in Emerging Markets? Value, and Momentum Factors Drive Stock Returns in Emerging Markets.
- 4. Chen, H. J. (2011). Firm life expectancy and the heterogeneity of the book-to-market effect. Journal of Financial Economics, 100(2), 402-423.
- Chou, P. H., Ho, P. H., & Ko, K. C. (2012). Do industries matter in explaining stock returns and asset-pricing anomalies? Journal of Banking & Finance, 36(2), 355-370.
- DeBondt, W. F. M., & Thaler, R. (1985). Does the stock market overreact? Journal of Finance, 40(3), 793-805.
- 7. Dempsey, M. (2010). The book-to-market equity ratio as a proxy for risk: evidence from Australian markets. Australian Journal of Management, 35(1), 7-21.
- 8. Fama, E. F., & French, K. R. (1992). The crosssection of expected stock returns. Journal of Finance, 427-465.

- 9. Fama, E. F., & French, K. R. (1993). Common risk factors in the returns on stocks and bonds. Journal of Financial Economics, 33(1), 3-56.
- 10. Fama, E. F., & French, K. R. (1996). Multifactor explanations of asset pricing anomalies. The Journal of Finance, 51(1), 55-84.
- Figelman, I. (2007). Stock return momentum and reversal. The Journal of Portfolio Management, 34(1), 51-67.
- 12. Gharaibeh, O. K. (2016). The Inter-Firm Value Effect in the Qatar Stock Market: 2005-2014. International Journal of Business and Management, 11(1).
- Grinblatt, M., & Moskowitz, T. (2004). Predicting stock price movements from past returns: The role of consistency and tax-loss selling. Financial Economics 71(3), 514-579.
- Hasan, M. B., Alam, M. N., Amin, M. R., & Rahaman, M. A. (2015). The Size and Value Effect to Explain Cross-Section of Expected Stock Returns in Dhaka Stock Exchange. International Journal of Economics and Finance, 7(1), 14.
- Lakonishok, J., Shleifer, A., & Vishny, R. W. (1994). Contrarian investment, extrapolation, and risk. The Journal of Finance, 49(5), 1541-1578.
- Malin, M., & Bornholt, G. (2013). Long-term return reversal: Evidence from international market indices. Journal of International Financial Markets, Institutions and Money, 25, 1-17.
- Newey, W. K. a., & West, K. D. (1987). A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix. Econometrica: Journal of the Econometric Society, 703-708.
- Peterkort, R. F., & Nielsen, J. F. (2005). Is the bookto-market ratio a measure of risk? Journal of Financial Research, 28(4), 487-502.
- 19. Simlai, P. (2009). Stock returns, size, and book-tomarket equity. Studies in Economics and Finance, 26(3), 198-212.
- 20. White, H. (1980). A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity. Econometrica: Journal of the Econometric Society, 48, 817-838.

## This page is intentionally left blank



GLOBAL JOURNAL OF MANAGEMENT AND BUSINESS RESEARCH: C FINANCE Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

# The French SMEs and their Financial Performance at the Launch of the Corporates Acquisition Programs

## By Omar Boufama

#### University, Skikda, Algeria

*Abstract-* This paper presents an econometric analysis of the profitability generated at the initiation of the corporates acquisition programs by French investors. These programs generally launched by the firms directors, need to have a high frequency of acquisition attempts, in order to cover fixed costs of the program and generate profits, which they increase the value of the firm and enrich the shareholders. The French "SME" succeed better than large companies in the launch of acquisition programs. The hypothesis of the announcement effect was been verified, and the two others, were rejected by the t-student test. In a subsample two "SME", Guerbet and Sartorius from Healthcare sector have a positive value. These results show clearly that the corporate acquisition programs are projects of value-creating investment for French "SME".

Keywords: acquisition program; announcement effect; economic impact; profitability.

GJMBR - C Classification : JEL Code: E44



Strictly as per the compliance and regulations of:



© 2016. Omar Boufama. This is a research/review paper, distributed under the terms of the Creative Commons Attribution. Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

## The French SMEs and their Financial Performance at the Launch of the Corporates Acquisition Programs

Omar Boufama

Abstract- This paper presents an econometric analysis of the profitability generated at the initiation of the corporates acquisition programs by French investors. These programs generally launched by the firms directors, need to have a high frequency of acquisition attempts, in order to cover fixed costs of the program and generate profits, which they increase the value of the firm and enrich the shareholders. The French "SME" succeed better than large companies in the launch of acquisition programs. The hypothesis of the announcement effect was been verified, and the two others, were rejected by the t-student test. In a subsample two "SME", Guerbet and Sartorius from Healthcare sector have a positive value. These results show clearly that the corporate acquisition programs are projects of value-creating investment for French "SME".

*Keywords:* acquisition program; announcement effect; economic impact; profitability.

#### I. INTRODUCTION

nowledge gleaned on the French case regarding corporates acquisition all are almost reserved to the big companies, especially for the valuation of the acquisition programs gain. Recent empirical research has shown that, the aim objective of merger and acquisition (M&A) operations for the acquiring firms is to operate an external expansion with the aim of getting bigger and developing. A competitive market of corporates acquisition implies that the earnings of the acquirers during no successful acquisitions have a negative price of bid; shareholders of the target firm will accept this price. Ruback (1983) tested this implication in a study using data on takeovers desired by several acquiring firms. The results confirm the hypothesis of market competition concerning acquisition operations.

From this perspective, the problem, which we try to deal and to analyze, is to demonstrate how to measure, analysis and explain the profitability of the acquisition operations within the framework of the acquisition programs launched by French firm's directors.

As it is noted in the research of Aktas, N., De Bodt, Roll, R.  $(2009)^1$ , why would firms undertake acquisitions if not to create value? The problem we

cleared up the possible ways being able to help to answer a fundamental question, which is why the directors launch programs of acquisition while the present value of the program is negative?

The second case is when the acquisition programs are positive, that is generate profits for the acquirers. In this case, our question, which is secondary, is to know why then the acquirers do not launch them in a continual way? This question is justified by the fact that there are very precise periods when the directors make this complex kind of investment.

From this perspective, our empirical study in which we evaluate the profitability of corporates acquisition programs, try to specify stocks price reactions at the announcement of M&A operations. It is a question particularly to verify if the partially anticipated events by the French acquirers generates consistent returns, which can be compared with those of the American case.

The actual financial literature about this subject largely try to calculate the gains associated to the acquisition operations for both, acquirers and target firms. Another point treated in this literature focus on the identification of the impact of these operations on the shareholders wealth.

Malatesta and Thompson (1985) provide evidence on the acquisition programs profitability for American case. Both researchers are the first ones who established an econometric model, allowing estimating the value of an acquisition program. Previously, Schipper and Thompson (1983) realized an empirical study in which they measure the impact of acquisitions activity on firm value by differentiating between specific merger events and programs of acquisition activity. Through a sample of conglomerate acquirers, they find significantly positive abnormal returns associated with the announcement of acquisitions programs and significantly negative returns associated with certain institutional laws.

In another type of studies published in the 11 volume of the "Journal of Financial Economics", Asquith, Bruner and Mullins (1983) examines the effect of mergers on the wealth of bidding firms' shareholders. They clarifies some interesting points about acquisitions activity. They feel that the bidding firms gain significantly during the twenty-one days leading to the

Author: University, Skikda, Algeria. e-mail: o.boufama@univ-skikda.dz

<sup>&</sup>lt;sup>1</sup>Aktas, N., De Bodt, E., Roll, R., 2009. Learning, hubris and corporate serial acquisitions. Journal of Corporate Finance 15, 543-561.

announcement of each of their first operation within four merger bids. These results have not a support to the capitalization hypothesis that acquirers' gains are capitalized at the beginning of merger programs.

In the same volume of the journal reserved to the wealth effects of mergers and the market for corporate control, Malatesta (1983) examines the net effects of the long-run sequence of events leading to merger, and of merger per se, on shareholder wealth. The author find that the long-run wealth effect of the event sequence culminating in merger is significantly negative for acquiring firms.

This study examines the profitability of corporates acquisition programs by distinguishing between types of effects. Specifically, an acquisition program has an announcement effect and an economic impact for each acquisition attempt made by the acquiring firm. The study determines the value of the both effects and the abnormal returns for the acquisition attempts on the non-event periods, for the total sample of big and SME French companies. The results shows that for the total sample which account 46 French acquirers, acquisition programs are losing. However, by considering the subsamples of the study, we find that the French SME perform better than, the big companies, and create value expressed in positive returns generated from their programs.

The next section explores theoretical considerations about the hypothesis studied within the framework of the estimation of corporates acquisition programs. I then describe the methodology used to evaluate reaction to acquisition programs announced by French acquirers. In this section, we try to present the mathematical development of the model that I apply on a sample of French acquirers in order to determine if corporates acquisition programs at the announcement date have a positive net present value or no. The following section is reserved to a discussion on the results of a number of SME, which are included in the initial sample. Through these results, I demonstrate that this type of firms are more profitable than, the other firms. The final section summarizes the results of the study.

### II. Theoretical Considerations on Acquisition Programs

The corporates acquisition programs were little handled by researchers because they are a very specific subject in finance. Indeed, directors must publicly announce the acquisition programs in order to be able to capitalize the gains, which are associated to it, at the time of their launch. The public announcement of this type of information, which is the launch of acquisitions programs, is very important for shareholders of acquiring firms. It allows their enrichment by the increase in share prices. In this case, the very important assumption implied that acquisitions operations as mergers and tender offers are comprised partly of individual events undertaken within structured acquisitions programs. However, the empirical studies shows that the results of acquiring firms are expected by acquirers to gain from acquisitions activity, although some of the evidence is without supports.

The effect of an acquisition program must be profitable to the acquiring firm and to these shareholders at the same moment of the launch of said program. For example, if a firm were to announce an event concerning the structure of its capital, by then the reaction of its shares should take place at the announcement of the event, not at any later date.

There is a difference between individual acquisitions and programs of acquisition activity. Generally, for individual acquisition operations, the methodology used in the estimation of the returns of the shareholders and the variation in the firm value of the both target and bidding companies is the "event study methodology". Fama, Fisher, Jensen and Roll (1969) developed this technique in order to capture the effect of an event on stock prices by calculating the abnormal returns caused by this event. Studies that have examined the gains of mergers and tender offers applies the method based on the estimation of abnormal stock returns at the time of the specific event and surrounding this event date.

One of the preliminary hypothesis on which is based the evaluation of an acquisition program is relative to the value-maximizing behavior. Within the same framework, several hypothesis have been put forward in empirical studies, which estimate the value of the acquisition programs. In Malatesta and Thompson (1985), the capitalization hypothesis were treated and they find that is consistent with constant announcement effects for successive acquisition attempts. The net present value of all the acquisition attempts made by the acquiring firm within the program is fully capitalized in firm value at the time of the initiation of the program. The frequency of acquisition attempts influences the net present value of the program, also the expected economic impact imputed to future attempts raise the acquisition program value, and the fixed costs are attributed to the program.

The estimation of the capitalized value of acquisitions programs is very important for two reasons. First, as Malatesta (1981) have pointed out, the fixed costs of an acquisition program will not affect the price reaction around an acquisition event. Further, if acquisition program requires any initial expenses, price reactions to individual event announcement as take-over or merger could indicate a positive return to those expenses. For that purpose, the price reaction to an acquisitions program will reflect the cost of the initial outlay as well as the expected return on it. The second and equally important reason is that once the expected value of an acquisitions program is capitalized, variations in the value of an acquiring firm surrounding event announcements will reflect only the surprise associated with the terms of the individual event. This surprise is measured relative to the expectations, which were capitalized at the time the acquisitions program was publicly announced.

The two main studies of the profitability of corporates acquisition programs within the framework of M&A operations are the ones of Shipper and Thompson (1983) and, Malatesta, and Thompon (1985).The results of the first study indicate that the acquiring firms anticipate gains from acquisitions activity. The second study approve this result. Indeed, Malatesta and Thompson (1985) find that the average estimated economic impact of an acquisition attempt exceeds 4 million dollars, but they cannot conclude that an acquisition program is desirable for firms in general.

#### III. Analysis of French Acquisitions Programs Profitability

Our questions relative to the profitability of the corporates acquisition programs were studied in the context of French M&A operations realized between 1997 and 2007. In this section, we describes our sample of firms involved in a process of acquisition operations in series called acquisition program, then we present the empirical methodology used and hypothesis. In the third part of this section, we present the results for the entire sample of 46 acquiring firms.

#### a) Data

The acquiring firms sample analyzed in this study comes primarily from the AMF<sup>2</sup> files, which, are on its Web site. During the observation period (1997-2011), the French acquirers made 1050 acquisition operations. These operations concern only four types of acquisition, which are takeover bid, public offer of exchange, mixed public offer and merger.

The basic selection criterion is concerning the number of acquisition attempts made by the acquirer. Every acquiring firm having made at least an acquisition attempt during the eleven years of the study period is held in the sample. We strictly apply this selection criterion because it is impossible in this case (French case) to verify the real public announcement of the acquisition programs in the financial press.

Two firms are concerned in every acquisition operation, the acquiring firm and the acquired (target) firm. Our empirical analysis touches only the French acquiring firms. In final our sample study covers over 11 years, from 1997 to 2007 and they contains 46 French frequently acquirers. These firms represents practically all the economic sectors of France. The entire 46 firm

<sup>2</sup> AMF : Autorité des Marchés Financiers, www.amf-france.org

sample which announced and subsequently carried out acquisition programs launched a total of 97 acquisition operations, they represents on average 2.11 attempts by firm during the study period. This represents 95% of all the acquisition attempts counted during observation period (1997-2011) as it shown in table 1 and figure 1. We consider the acquisition operations made by the acquiring firms as acquisition attempts since they can be successful or unsuccessful.

All the financial and stock exchange data concerning the 46 French acquirers about which we evaluate the profitability of their acquisition programs are obtained from *Datastream* database. This data ranges from stock price, dividend, CAC 40 stock index, market value and FRANCE TREASURY BILL 1 MONTH for interest rate. These data is gathered every month and we calculate the zero beta portfolio return after having calculate the monthly stock return for each firm. We use the zero beta return to calculate the stock and market risk premium and it instead of interest rate.

#### b) Empirical methodology and assumptions

The fundamental idea according to which Malatesta and Thompson (1985) have developed their econometric model for assessing profitability of an acquisition program is diverted from estimation of a classical investment. In the launch of an acquisition program, the acquiring firm has fixed costs that she hope to cover with future cash flows generated by their acquisition attempts. The net present value of the acquisition program for an acquiring firm j,  $NPV_j$  is given by

$$NPV_j = \frac{q_j \cdot v_j}{r} - F_j \tag{1}$$

$T_{\frown}$	h		1
12	())	-	
<i>i</i> a	$\sim$	$\mathbf{O}$	

Summary of the sample elements for empirical study

	The sample elements	Rough description					
Da	ta :	So	urce of data :				
-	Choice of firms	-	www.amf-france.org				
-	dates of the first announcement	-	www.amf-france.org				
-	Stock exchange and financial data	-	Datastream database				
-	Types of acquisition attempts	-	Take-over (cash, exchange, mixte) and				
The	e periods :		mergers				
_	Observation period	De	limitation of periods :				
-	Studyperiod	-	1/1997 to 12/2011				
-	Periodicity of the data	-	1/1997 to 12/2007				
-	Length of the study period	-	Monthly				
The	e acquisition attempts :	-	132 months				
_	Number of firms	Sta	itistical data :				
_	Number of firms without acquisition	-	46				
	attempts	-	0				
-	Sum of acquisition attempts during study	-	97				
	period	Sta	itistical data :				
Ave	erage of the attempts :	-	2.11				
-	Sum of attempts per firm	-	0.19				
-	Sum of annual attempts per firm						







Where  $F_j$  denotes the program's fixed cost that contains all kinds of expenses in order to prolong the program. In addition, for each individual future attempt, the economic impact is indicated by  $v_j$ . Acquisition attempts within the program have a constant frequency  $q_i$  per period, the risk free interest rate equal to r.

The approach that I use in our empirical study in order to determines if French acquisitions programs are profitable or not begin by the estimation of the stock rate of return for each firm within the sample and the market return

$$R_{it} = \frac{P_{it} - P_{it-1} + D_{it}}{P_{it-1}}$$
(2)

where  $R_{it}$  is the realized rate of return to stock *i*at time *t*,  $P_{it}$  and  $P_{it-1}$  are the stock price as they are taken from *Datastream* database at time *t* and *t* - 1 successively, and  $D_{it}$  represents the dividend on the stock at time *t*.

$$R_{mt} = \frac{I_{CAC \ 40_t} - I_{CAC \ 40_{t-1}}}{I_{CAC \ 40_{t-1}}} \tag{3}$$

in which  $R_{mt}$  is the return on the market at time t, and  $I_{CAC40_t}$ ,  $I_{CAC40_{t-1}}$  are the stock index CAC 40(Paris Stock Index) at time t and t - 1 successively.

These two calculations allow us to begin the estimation of the multiple regression coefficients of the principles four models. The basic model serving to verify the hypotheses of the study is in the form of a multiple regression with two independent variables, the first one is the market risk premium weighted by firm value  $V_{jt-1}$  at the time t-1. The second variable is a binary,  $d_{jt}$  equal to one for a firm *j*, which makes an acquisition attempt during period *t* and zero otherwise

$$(\tilde{r}_{jt} - r_t) V_{jt-1} = \alpha_j + \beta_j V_{jt-1} (\tilde{r}_{mt} - r_t) + \gamma_j d_{jt} + \tilde{e}_{jt}$$
(4)

where  $(\tilde{r}_{jt} - r_t)V_{jt-1}$  is the stock risk premium weighted by the firm value expressed by the market capitalization at the time t-1. The market risk premium  $V_{jt-1}(\tilde{r}_{mt} - rt$  is one of the two independent variables of the regression equation; this one also is weighted by the firm value. The second independent (explanatory) variable of the multiple regression is the binary variable  $d_{jt}$ , which is equal to one if the firm take an acquisition attempt and zero when the firm does not take attempt during given period.

The main objective of our study is to estimate the gains or the losses of the French corporates acquisition programs, which spread out over several years. This evaluation is completed by the calculation of three coefficients  $\alpha_j$ ,  $\gamma_j$  and  $\alpha_j + \gamma_j$  of the eq. (1). The analysis on which is based our empirical study is applied through four regressions, among which the three coefficients of two of them are expressed in excess euro returns, and the two other regressions are in excess rate of return form, as it is shown below

$$(\tilde{r}_{jt} - r_t)V_{jt-1} = \alpha_j + \beta_j V_{jt-1}(\tilde{r}_{mt} - r_t) + \gamma_j d_{jt} + \tilde{e}_{jt} \quad (5)$$

$$\left(\tilde{r}_{jt} - r_t\right) = \alpha_j + \beta_j \left(\tilde{r}_{mt} - r_t\right) + \gamma_j d_{jt} + \tilde{e}_{jt}$$
(6)

$$\left(\tilde{r}_{jt} - \tilde{r}_{zt}\right)V_{jt-1} = \alpha_j + \beta_j V_{jt-1}(\tilde{r}_{mt} - \tilde{r}_{zt}) + \gamma_j d_{jt} + \tilde{e}_{jt}$$
(7)

$$\left(\tilde{r}_{jt} - \tilde{r}_{zt}\right) = \alpha_j + \beta_j \left(\tilde{r}_{mt} - \tilde{r}_{zt}\right) + \gamma_j d_{jt} + \tilde{e}_{jt} \qquad (8)$$

Both equations in which appear the variable defining the firm value  $V_{jt-1}$  are the ones which give the results in excess euro returns, and the two others which are without this variable, their results are in excess rate of returns. Another specification concern both risk premium. Two equations eq (5) and eq. (6) are expressed in interest rate and the two others eq. (7) and eq. (8) use the zero beta return to calculate both risk premium, that of the stock and that of the market.

Three hypotheses are treated in this study. The first one concern the variations of the abnormal returns that should be negative in non-event period. The second one predicts that the expected economic impact is positive and the last one it is about the expected announcement effect that should be positive. An overview on these hypotheses is presented in table 2 below:

#### Table 2

1 l	- 1 + 1	! - ! I	- + + · · · ·
HVDDTDASAS	OT THE	empirical	STIION
		CITIDITIOU	JUGUV

Hypothesis	Coefficient	Coefficient sign	Definition of the hypotheses
$H_1: \alpha_j = -q_j v_j < 0$	$\alpha_j$	(-)	Expected abnormal return of an acquisition attempt for a non-event period
$H_2: \gamma_j = v_j > 0$	Υ <sub>j</sub>	(+)	Economic impact of an acquisition attempt
$H_3: \alpha_j + \gamma_j = (1 - q_j)v_j > 0$	$\alpha_j + \gamma_j$	(+)	Announcement effect of an acquisition attempt

The first hypothesis  $H_1$  implies that the non-event period is the non-announcement period this means that is the time when the acquirer do not launch acquisition attempt. In every period, the frequency of acquisition attempt is positive  $q_j$  and because the expected economic impact of an acquisition attempt  $v_j$  ispositive, then the variation of the abnormal returns in nonannouncement periods  $\alpha_j$  must be negative. In the second hypothesis  $H_2$ , since  $\gamma_j$  is equal to  $v_j$  it follows that  $\gamma_j$  that measure the economic impact must be positive. The announcement effect of acquisition attempts is tested by the hypothesis  $H_3$ , which states that the announcement effect must be positive if the expected economic impact is positive.

#### c) Results

For each of 46 companies constituting our sample, the parameters of every model 1, 2, 3 and 4 are considered by the ordinary least squares approach (OLS) applied to 132 months of study period (01/01/1997 to 31/12/2007).

Table 3 reports the main results of our study. In this table, I present for the four models, the average estimation of the coefficients, the average of t-student test and the significant percent different from zero of the coefficients.

In the panel A, the model 1 test the first hypothesis that  $\alpha_j < 0$ , I find that only one case on 46, this result was obtained and which is significant at the level of 0.05. This unique negative alpha is the one of the firm Unibel, an SME company specializes in the food

industry. On the other hand, in 32 cases we reject the null hypothesis in favor of the alternative  $\alpha_j > 0$  for a significance level of 0.1. The number of non-significant cases is 13 cases on 46 studied what represents approximately 28% of the total sample.

The average of  $\tilde{\alpha}_j$  for the entire sample is 169.033 million  $\in$  what is against the hypothesis which predicts a negative sign. The explanation of this result according to which the French financial market is not efficient is not plausible, but there are other factors, which can explain this, as for example the competitiveness in the French corporates acquisition or what the launch of acquisition programs is expensive to the French acquiring firms.

The results given in panels B, C and D confirm the result of the basic model shown in panel A for both types, results expressed in excess euro returns or results in excess rate of return form.

#### IV. The Acquisitions Programs Profitability of French SME

All the results on the profitability of the French corporates acquisition programs presented above are relative to the entire sample of French companies. Bouzgarrou and Navatte (2012)<sup>3</sup> find positive significant CARs (Cumulative Abnormal Retrns) for French acquirers of private targets and negative insignificant CARs for French acquirers of listed targets. This result

my help us to complete our study. Indeed, to seize well the firms, which are more successful than others are, we distributed the 46 companies making up our sample study on subsamples. This work allowed us to constitute eight subsamples of industry: real estate, industrial services, chemistry and oil, technology, banks, media, distribution and health. For example, in the sector of the real estate, we have Vinci, Bouygues and especially Saint-Gobain which is a leader in his domain and very successful. Note that, in our sample there is also a small and medium enterprises "SME" as by example in the sub-sample of the Healthcare sector. This sub-sample consists of two companies only, Sartorius and Guerbet.

Indeed, for this sector and for both firms together the sign of both coefficients  $\gamma_j$  and  $\alpha_j + \gamma_j$  is positive. This is in line with the signs waited in both hypotheses, that of the economic impact and the announcement effect. The value of these two parameters, are estimated at 3.57 and 5.28 million  $\in$ . Thus, there is at least one sector among eight branches of industry, which verified both hypotheses  $H_2$  and  $H_3$ . This is also an obvious result of the performance of the French EMS within the framework of French corporates acquisition programs.

Guerbet and Sartorius, together has significant results and their acquisition programs are slightly profitable but their financial performance are better than other big firms. Sartorius is listed at the SBF 120 stock index and Guerbet at the CAC SMALL 90 index.

#### Table 3

#### Summary of results for the four models

For each firm *j* the rates of return is indicated by  $\tilde{r}_{jt}$  and the market return by  $\tilde{r}_{mt}$ . Riskless interest rate is noted by and in some cases, this interest rate is replaced by the zero beta portfolio return denoted by  $\tilde{r}_{zt}$ . The b  $d_{jt}$  take two value: (1) if firm *j* 

firm j at period t - 1 is expressed by  $V_{jt-1}$ .

For each of 46 acquiring firms, estimates of  $\alpha$ ,  $\gamma$  and  $\alpha + \gamma$ 

from 1/1/1997 through 31/12/2007, while  $\alpha$  measures the acquisition attempts abnormal return on nonperiods,  $\gamma$  measures the economic impact of acquisition attempts and  $\alpha + \gamma$  measures the announcement effect.

Coefficient	Number of Estimates	Average Estimate	Percent Positive	Average <i>t-statistic</i>	Percent s (0.1 level of s	ignificant significance)
	( <b>J</b> )		(%)		(+)	(-)
Panel A Model 1	$(\widetilde{r}_{jt}-r_t)V_{jt-1}$	$\alpha_{j} = \alpha_{j} + \beta_{j} V$	$r_{jt-1}(\widetilde{r}_{mt}-r_t)+$	$\gamma_j d_{jt} + \widetilde{e}_{jt}$		
$ \begin{array}{c} \alpha \\ \beta \\ \gamma \\ \alpha + \gamma \end{array} $	46 46 46 46	169.033ª 1.20 -40.901ª 128.131ª		2.48 <sup>d</sup> 7.55 <sup>d</sup> -0.25 <sup>d</sup> 0.04	69.57 97.83 2.17 58.7	2.17 0.00 10.87 37.00

<sup>3</sup> Bouzgarrou, H., Navatte, P., 2012. Short Term Wealth Creation Sustainability of French Acquirers of Unlisted Versus Listed Firms. Bankers, Markets & Investors N° 121, November-December, 47-58.

Panel B Model 2: $(\tilde{r}_{jt} - r_t) = \alpha_j + \beta_j (\tilde{r}_{mt} - r_t) + \gamma_j d_{jt} + \tilde{e}_{jt}$							
α	46	0.0240	97.8	3.02°	79.09	2.17	
β	46	0.8332	100.0	6.71 <sup>b</sup>	96.65	0.00	
γ	46	-0.0101	45.7	-0.11°	4.35	2.17	
$\alpha + \gamma$	46	0.0140	58.7	0.07	56.5	41.30	
Panel C Model 3: $(\widetilde{r}_{jt} - \widetilde{r}_{zt})V_{jt-1} = \alpha_j + \beta_j V_{jt-1}(\widetilde{r}_{mt} - \widetilde{r}_{zt}) + \gamma_j d_{jt} + \widetilde{e}_{jt}$							
α	46	147.01ª	82.6	1.80 <sup>d</sup>	56.52	2.17	
β	46	0.92	100.0	9.13 <sup>b</sup>	97.83	0.00	
γ	46	-69.09ª	41.3	-0.27 <sup>d</sup>	2.17	10.87	
$\alpha + \gamma$	46	77.92ª	56.5	-0.03	54.35	41.30	
Panel D Model 4: $(\tilde{r}_{jt} - \tilde{r}_{zt}) = \alpha_j + \beta_j (\tilde{r}_{mt} - \tilde{r}_{zt}) + \gamma_j d_{jt} + \tilde{e}_{jt}$							
α	46	0.0218	91.3	2.41 <sup>d</sup>	69.56	2.17	
β	46	0.8732	100.0	9.18 <sup>b</sup>	100.00	0.00	
γ	46	-0.0107	43.5	-0.13 <sup>n·s</sup>	4.35	4.35	
$\alpha + \gamma$	46	0.0111	54.3	0.04	52.17	45.65	

<sup>a</sup> In million €<sup>b</sup> Significant at the 1% level

° Significant at the 5% level

<sup>d</sup> Significant at the 10% level

<sup>n</sup>·s No Significant

#### V. Summary and Conclusions

By examining acquisitions programs, launched by frequently French acquiring firms from half of the 90s until the middle of the 2000s, I find that only the announcement effect hypothesis appears to apply to French acquisitions programs. Indeed, our results are a little limited by the weakness of the frequency of the acquisition attempts launched by the French acquirers.

Finally, proved evidence through the four regressions results supports partially the hypothesis that acquisition programs can be perceived as profitable investment projects for the French acquirers essentially those of "SME". This conclusion is justified by the individual results of firms. Otherwise, the sample which consists of 46 French companies considered as frequent acquirers on French acquisition market indicate clearly that firm's acquisition programs during the last decade were destructive of value. It means they do not maximize the value of the firm nor the stockholder's wealth.

#### References Références Referencias

- 1. Aktas, N., De Bodt, E., Roll, R., 2013. Learning from repetitive acquisitions: Evidence from the time between deals. Journal of Financial Economics 108, 99-117.
- Aktas, N., De Bodt, E., Roll, R., 2009. Learning, hubris and corporate serial acquisitions. Journal of Corporate Finance 15, 543-561.
- 3. Arvanitis, S., Stucki, T., 2014. Do mergers and acquisitions among small and medium-sized

enterprises affect the performance of acquiring firms? International Small Business Journal, January, 1-34.

- 4. Asquith, P., Bruner, R., Mullins, D., 1983. The gains to bidding firms from merger. Journal of Financial Economics 11, 121-139.
- Bouzgarrou, H., Navatte, P., 2012. Short Term Wealth Creation Sustainability of French Acquirers of Unlisted Versus Listed Firms. Bankers, Markets & Investors N° 121, November-December, 47-58.
- 6. Malatesta, P., Thompson, R., 1985. Partially anticipated events: a model of stock price reactions with an application to acquisitions. Journal of Financial Economics 14, 237-250.
- Malatesta, P., 1983. The wealth effect of merger activity and the objective functions of merging firms. Journal of Financial Economics 11, 155-181.
- 8. Schipper, K., Thompson, R., 1983. Evidence on the capitalized value of merger activity for acquiring firms. Journal of Financial Economics 11, 85-119.
- 9. Weston, F., Copeland, T., 1992. Managerial Finance. Ninth Edition. The Dryden Press International Edition.
- 10. Web site of "Autorité des Marchés Financiers": www.amf-france.org.

## This page is intentionally left blank



GLOBAL JOURNAL OF MANAGEMENT AND BUSINESS RESEARCH: C FINANCE Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

# Macroeconomic and Firm Specific Determinants of Profitability of Insurance Industry in Ethiopia

## By Demis Hailegebreal

Jimma University

*Abstract-* This study was conducted on the determinants of profitability of Ethiopian insurance industry. The study attempts to examine the firm specific factors which are age of company, size of company, leverage ratio, liquidity ratio, premium growth, technical provision, underwriting risk, solvency, re-insurance dependency and tangibility of assets and macroeconomic factors; GDP and Inflation on profitability of Ethiopian insurance industry. Nine insurance companies from the total of 17insurance companies established before 2008 were included in the study. Secondary data that was collected from the financial statements (Balance sheet and income statements) of insurance companies; and annual reports of National bank of Ethiopia are the major sources of data for this study. This study found that under writing risk, technical provision, leverage and inflation have negative and significant effect whereas premium growth, age of the company, solvency ratio and GDP have statically positive and significant relationship with the profitability of Ethiopian insurance industry.

Keywords: Profitability of Insurance industry, Macro-economic and Firm Specific Determinants, Insurance Industry in Ethiopia.

GJMBR - C Classification : JEL Code: E44



Strictly as per the compliance and regulations of:



© 2016. Demis Hailegebreal. This is a research/review paper, distributed under the terms of the Creative Commons Attribution. Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

## Macroeconomic and Firm Specific Determinants of Profitability of Insurance Industry in Ethiopia

Demis Hailegebreal

Abstract- This study was conducted on the determinants of profitability of Ethiopian insurance industry. The study attempts to examine the firm specific factors which are age of company, size of company, leverage ratio, liquidity ratio, premium growth, technical provision, underwriting risk, solvency, reinsurance dependency and tangibility of assets and macroeconomic factors; GDP and Inflation on profitability of Ethiopian insurance industry. Nine insurance companies from the total of 17 insurance companies established before 2008 were included in the study. Secondary data that was collected from the financial statements (Balance sheet and income statements) of insurance companies; and annual reports of National bank of Ethiopia are the major sources of data for this study. This study found that under writing risk, technical provision, leverage and inflation have negative and significant effect whereas premium growth, age of the company, solvency ratio and GDP have statically positive and significant relationship with the profitability of Ethiopian insurance industry. However, the study found that liquidity, re-insurance dependency, tangibility of assets and company size have no significant effect on the profitability of insurance industry in Ethiopia.

Keywords: Profitability of Insurance industry, Macroeconomic and Firm Specific Determinants, Insurance Industry in Ethiopia.

#### I. INTRODUCTION

nsurance companies are playing vital role through saving, pooling of funds for huge investment, risk sharing and protection from suffering from risk for economic growth of developed and developing countries. That is, insurance companies, by channeling funds form savers to those have shortage of funds but have business ideas, and transferring risks from insurers to the insureds, can facilitate fund mobilization, saving and investment in particular country.

Previous study, Naveed et al (2011), stated that the effectiveness of insurance companies and transfer of risk can have an influence on economic growth and institutional insolvencies can result in systemic risk which have adverse results in the country in general and in insurance industry in particular. Therefore, the vital role that financial institutions such as insurance companies remain in financing and insuring economic activities and contribute to the stability of the financial system in particular and the stability of the economy of a particular country in general, is part of protected and repaired system of the economy. In doing so, insurance industry in general and insurance company in particular should be financially sound and examining the determinants of financial soundness of insurance industry, is a big concern of the business research.

Hence, the issue needs empirical analysis so as to sort out what are the major factors influencing profitability of insurance industry in Ethiopia too and this will help concerned bodies to focus on the relevant factors. The sound financial performance of insurance industry is important and studies by different previous researchers focus on what factors affecting the financial performance of the sector. Likewise, this study examined firm specific and macro-economic determinants of profitability of Ethiopian insurance industry.

#### a) Problem statements

Insurance companies can support businesses and individuals through channeling funds and indemnifying the losses of other sectors in the economy. Moreover, insurance companies offer economic and social benefits in the society through loss prevention, anxiety reduction, fear reduction and increasing employment.

According to Hifza (2011), profitability is the ultimate goal of wealth maximization of financial management. The best soundness of any industry in general and any firm in particular plays the role of rising the market value of that specific firm attached with the role of leading towards the growth of the whole industry which finally leads to the overall success of the economy. measuring the financial performance of insurance industry is important and researchable area in the business research as the sector is not only providing the means of saving money and transferring risk but also helps channeling funds from surplus economic units to deficit economic units so investment activities in a country can be promoted.

The insurance industry is part of protected and repair scheme of an economy and successful operation of the industry can set energy for other industries and development of a country. For doing so, the insurance industry is expected to be financially sound enough or profitable in operation. Therefore, not only measuring the financial performance of insurance companies but also comprehensible imminent on factor influencing financial performance in the industry is the problem that

Author: Lecturer, Department of Banking and Finance, Jimma University, Ethiopia. e-mail: dhgebreal@gmail.com

must be examined. The Ethiopian economy is highly supported by insurance industry followed from the banking industry. As a result, the financial soundness of insurance industry in Ethiopia is not a compromising issue and examining the factors that can have an influence on the industry is highly researchable area.

The above critical issues motivated the researcher to put some sort of contribution on what factors have an effect on the financial performance of insurance companies. While taking importance of the issue of factors determining the profitability of insurance industry, the researcher tried to examine macroe-conomic and firm specific factors that have influence on the profitability Ethiopian insurance industry.

#### b) Hypothesis

*H1:* age has significant and positive effect on Ethiopian insurance industry's profitability.

*H2:* size has positive and significant effect on insurance industry's profitability in Ethiopia.

H3: leverage has negative and significant effects on profitability of Ethiopian insurance industry.

*H4:* Tangibility of assets of insurance companies and their profitability are positively related.

*H5:* Liquidity ratio and profitability of insurance companies are negatively related.

*H6:* solvency has positive and significant effect on profitability of Ethiopian insurance industry.

*H7:* premium growth has significant and positive effect on insurance industry's profitability in Ethiopia.

*H8:* re-insurance dependency significantly and negatively affects the profitability of Ethiopian insurance industry.

*H9:* underwriting risk has a negative and significant effect on Ethiopian insurance industry's profitability.

*H10:* technical provision has a negative and significant effect on profitability of Ethiopian insurance industry.

*H11:* GDP has positive and significant effect on profitability of insurance industry in Ethiopia.

*H12:* Inflation has negative and significant effect on Ethiopian insurance industry profitability.

#### c) Research objectives

The main objective of this study is to identify the major factors that affect the profitability of insurance industry in Ethiopia. Specifically, this study is designed to measure the extent to which these determinants exert impact on insurance companies' profitability; to determine the relationship between these factors and profitability in insurance companies and to make policy recommendations regarding the key drivers of profitability of insurance companies in Ethiopia based on the empirical findings.

#### d) Significant of the study

This research will help the policy makers and managers of insurance industry in Ethiopia to consider major determinants of insurance industry in Ethiopia. Despite the role of insurance for the overall growth of Ethiopian economy (that is affected by the performance or profitability of the industry), only few researches are conducted on the area. As far as the researcher's knowledge is concerned, there have not been empirical studies addressed the performance of insurance industry in Ethiopia. Thus, this research is aimed at filling this gap; motivate other researches to the area and providing appropriate recommendation.

#### e) Scope of the study

This study will be delimited on the firm specific determinants of profitability of insurance industry in Ethiopia from the fiscal year of 2008 to 2013. Both public and private and life and non life insurance companies will be included in the study.

#### II. Empirical Literature

**Renbao Chen et.al (2004)** investigated that "higher profit provide both the means (larger obtainability of money from retained earned or from the capital market) and the incentive (a high rate of return) for new investment". This shows that insurance companies are needed to profitable or financially sound so as to support other industries in the economy.

Hifza (2011), stated that insurance companies plays vital role in promoting commercial and infrastructural businesses by encouraging financial and social stability; mobilizing and channeling savings; supporting trade, commerce and entrepreneurial activities and improves the standard of the lives of individuals and the overall wellbeing in a country.

Renbao Chen et.al (2004) summarized firm specific factors affecting profitability of property or liability insurance of general insurers and life or health insurance and provide valuable guidelines for insurers financial soundness as life/health insurance companies are different from property/liability insurers in terms of business, investment, exposure and length of liabilities. According to **Renbao Chen et.al 2004**, Life insurers perform a function of financial intermediation though general insurers act as risk takers.

Hamdan 2008) stated that return on assets (ROA), return on equity (ROE) and return on invested capital (ROIC) are used for the measurement of insurance companies profitability. Accordingly, ROA is the measure of financial performance of the company using its total assets. This is an indication of how effective management is in using the total assets to generate earnings whereas ROE measures a company's profitability which tells how much a company generates earnings with the money shareholders have invested. ROIC is a measure used to measure a company's

effectiveness in sharing the capital under its control in profitable business. This shows how well a company is in using its capital to generate returns. Comparing a company's ROIC with its weighted average cost of capital (WACC) indicates whether spent capital is used efficiently or not.

William H. Greene and Dam Segal (2004) in contrast, argued that the financial performance of insurance companies is normally expressed in net premium earned, underwriting profit, annual income, return on asset, return on equity. This can be categorized as profit performance and investment performance. Nevertheless, too many researchers in the field of insurance and their profitability stated that the key indicator of a firms' profitability is ROA. Philip Hardwick and Mike Adams (1999), Hafiz (2011) are among others, who have suggested among others, ROA can be the best proxy of profitability and better to use it.

Swiss Re (2008) indicated that profits are determined with underwriting performance (which are influenced by product pricing, risk selection, claims management, and marketing and administrative expenses) and investment performance.

Previous studies, Hifza(2011), SylwesterKozak (2011), Hamadan (2008) Swiss Re (2008) and Jay Angoff Roger Brown (2007) investigated factors of non life insurers profitability while Naveed Ahmed, Zulfqar Ahmed, Ahmad Usman (2011), Adams M., Hardwick P. and Zou H., (2008) Desheng Wu Z., Sandra V. &Lianga, (2007), Wright, K. M. (1992), investigated determinants of life and health insurance companies. Thus, most of the researchers focused on internal factors that influence profitability and most of the factors considered are age of company, size of company, leverage ratio, volume of capital, tangibility of assets and liquidity ratio.

Hamadin (2008) stated that internal factors can be financial statement variables and non-financial statements variables. The financial statement variables are factors which are directly driven from items in a balance sheet and profit & loss accounts of the insurance companies whereas the non-financial statement variables are those factors which are not directly shown on the financial statements accounts. Accordingly, Hamadin (2008), Kozak (2011), Swiss Re (2008), Brown (2007), Adams M., Hardwick P. and Zou H., (2008), Naveed et al (2011), Desheng Wu Z., Sandra V. & Lianga (2007), Wright, K. M. (1992), Flamini et.al (2009) and Hafiz (2011) used size, leverage, liquidity, tangibility of assets, and volume of capital.

Athanasoglou *et al.* (2005), investigated newly established banks are not particularly profitable in their first years of operation, as they give higher focus on boosting their market share, rather than on improving profitability. Similarly, Yuqi li (2007) stated that older banks expected to be more profitable due to their longer tradition and they have good reputation. Several studies have been conducted to evaluate the influence of company size and age on its profitability. However, the results on the relation of profitability with size of the company are somewhat different. For example, empirical study by Hardwick and Adams (1999) shows that there is an opposite relation between profitability and firm size. Brown (2007) found that age has a positive and significant effect on firms' profitability as measured by ROA.

Contrarily, Hamadan (2008) found that no significant statistical relation between age and profitability of insurance companies however, size has positive and significant effect on profitability. Also, Malik (2011) found that age has significantly and positive effect on company profitability.

Flamini et. al (2009) specified that **s**ize is used to show that larger companies are offering better economies of scale and providing a higher profits than smaller firms.

Accordingly, a positive relationship can exist between size and profitability by many insurance area researchers. However, for extremely large firms, the effect of size could be negative due to bureaucratic and other reasons (Yuqi Li, 2007). Therefore, the relationship between size and profitability may be expected to be non-linear. Athanasoglou *et al.* (2005) and Yuqi Li (2007) found positive relationship between size and profitability.

Liquidity is the probability of firm to pay liabilities which include operating expenses and payments for losses/benefits, reveals large current assets are held and idle if the ratio becomes high which could be examined in favorable investments. Naveed Ahmed et.al (2011) found that profitability (ROA) has no significant relationship with liquidity. Similarly, several researches evaluated the performance of the insurance companies. However, Chen and Wong (2004) found that liquidity is the important factor influencing of financial soundness of companies with a negative relationship.

Renbao Chen and Kie Ann Wong, (2004), Hamadan Ahamed Ali Al-Shami, (2008), Hifza Malik, (2011), Sylwester Kozak, (2011), Swiss Re, (2008) and Flamini et.al, (2009) found that negative and strong relationship between leverage and profitability of firms.

Tangibility of assets in insurance companies in most researches is determined by the proportion of fixed assets to total assets. Naveed Ahmed et.al (2011) investigated that size, profitability, age, risk, growth and tangibility are selected as explanatory variables and ROA as the measure of profitability. The study revealed that leverage, size and risk are major factors of performance of long term insurance whilst tangibility of asset has significant effect on ROA contrary, Hafiz Malik (2011) and Yuqi Li (2007) investigated that tangibility of asset has positive and significant effect on profitability of insurers. Chen-Ying Lee (2014) and Ana-Maria and Ghiorghe (2014) proved in their study that the financial leverage, company size, growth of gross written premiums, underwriting risk, and solvency margin are the most significant determinants of insurance industry's profitability.

In his study 'the effect of firm specific factors and macroeconomics on profitability property-liability insurance industry in Taiwan found that underwriting risk and reinsurance usage are the most determinants of profitability in Taiwan property-liability insurance industry. This study also found that economic growth is the most important determinants of profitability of insurance industry in Taiwan. Similarly, **Doreen (2013)** found there is a strong and positive relationship between GDP and insurance companies' profitability.

Gatzlaff (2009) conducted research entitle dimensions of property-liability insurer performance and found that operational performance was negatively related to underwriting risk and premium growth. Lee and Lee (2012), Olajumoke Olaosebikan (2012) and Hsu-Hua and Chen (2012) found in their study that reinsurance, underwriting risks, and liquidity ratio have significant influence on firm performance.

Ana-Maria and Ghiorghe (2014) conducted a research on the Determinants of Financial Performance in the Romanian Insurance Market and the result shows that the financial leverage, company size, growth of gross written premiums, underwriting risk, and solvency margin are the most significant determinants of Romanian insurance industry's profitability.

Pervan and Pavić (2010), Doreen (2013) and Y. Shiu (2004) in their study 'determinants of insurance companies' profitability in Croatia' found an inverse and significant effect of inflation on profitability of insurance industry.

**B. Charumathi (2012)** found that profitability of life insurers is positively and significantly influenced by the size and liquidity whereas, leverage, premium growth and negatively and significantly influenced the profitability of Indian life insurers. However, the study indicated that no significant relationship between underwriting risk and profitability.

#### III. Research Methodology

#### a) Research design

Depending on the nature of the research problem and the research perspective, a research method could be based on the philosophy of quantitative or qualitative or a combination of these two approaches.

According to Creswell (2003), quantitative research uses a review of the existing literature to deductively develop theories and hypotheses to be tested; the research problem is translated to specific variables and hypotheses.

Similarly, Creswell (2003) described qualitative approach as it uses the philosophical assumption of social constructivism world view that provides an understanding of social reality based on the subjective interpretation. Besides, the third approach is mixed research approach that seeks a pragmatic knowledge claim philosophy that consists of both quantitative and qualitative approaches.

Thus, in order to achieve the objectives stated in the previous section, bearing in mind the nature of research problem and the research outlook, this study mainly employed mixed research approach.

#### b) Target Population and sample size

Currently, as of 2016, there are 17 insurance companies in Ethiopia (www.nbe.gov.et). All private and public insurance companies established before the year 2008 were selected as a sample purposely. Accordingly, nine insurance companies were included in this study.

#### c) Types and Sources of Data

The data used for this study was secondary data which was collected from the audited financial statements of each insurance companies and NBE from the fiscal year of 2004 to 2014.

d) Variables with its measurement

	Variables	Measures	Sign	Sources
Dependent	Profitability (ROA)	Net profit before tax/total assets		
	Underwriting risk	claim incurred / premium earned	-	Ana-MariaandGhiorghe (2014)
Independent	Reinsurance dependence	premium ceded/total asset	-	Hsu-Hua and Chen (2012) and Olajumoke (2012)
	Solvency margin(SM)	net assets/net written premiums	+	Ana-Maria and Ghiorghe (2014), Shiu (2004), (B. Charumathi (2013).
	Company size	Natural logarithm of total assets		Hardwick and Adams (1999), Swiss Re (2008), Malik (2011)
	Liquidity(LQ)	Current Assets / Current Liabilities	-	Chen and Wong, (2004),

Premium growth(PG)	PG = (GWP (t) - GWP (t-1)) / GWP	+	EmineÖner Kaya (2015), Maria and
	(t-1)		Ghiorghe (2014)
Tangibility of assets	fixed assets/Total assets	+	HafizMalik (2011)
Leverage	Total liability/Total assets	-	Chen and Kie Ann Wong, (2004),
			Hamadan Ahamed Al-Shami (2008)
Technical provisions	claims outstanding/equity ratio	-	
Inflation(I)	I = (Inft - Inft - 1)/Inft - 1,	-	Chen-Ying Lee, 2014), and Doumpos
			et al., 2012),
Growth rate of GDP	(GDPt-GDPt - 1)/GDPt - 1	+	Gustina and Abdullah (2012), Beck
	·		and Webb (2003).

Sources: empirical literatures

e) Model specification

#### $ROA = C + \beta X_{it} + \varepsilon$ (adapted from Hifza, 2011)

Where ROA is return on assets,  $X_{it}$  is dependent variables for insurance "i" at time "t", C is constant,  $\beta$  is the coefficient and  $\epsilon$  is the error term.

a) Testing multi-colinearity problem

#### IV. Result and Discussion

The data collected from annual reports of each insurance company was analyzed with the help of software (stata 12.0) and then was interpreted in the following section.

VIF	Variable	1/VIF	
3.64	napw	0.275095	
2.57	coe	0.389662	· · · ·
2.15	ciep	0.464082	
2.12	cacl	0.470945	Summer of the local division of the local di
2.00	pcta	0.500610	
1.63	age	0.614648	
1.49	Pg	0.671438	
1.45	gdp	0.688739	
1.45	lev	0.691366	No. 1
1.38	inf	0.726781	
1.21	tangiblity	0.823763	
1.13	CS	0.882480	
1.85	Mean VIE	16	

Sources: author computation

The above table presents the multi-collinearity among the independent variables. According to Morgan et al, (2004), the variance inflation factor (VIF) above 10 or the tolerance value (1/VIF) below 0.1 is an indication that there is a problem of multi-collinearty among the variables. The above table shows that there is no VIF greater than 10 and 1/VIF below 0.1; in turn reveals any of the independent variable included in this study is not explained by the other. Hence all variables can be retained in the model of this study.

b) Testing hetroskedasticity problem

Table 4.2 : Testing hetroskedasticity problem

Breusch-Pagan / Cook-Weisberg test for heteroskedasticity Ho: Constant variance Variables: fitted values of roa

chi2(1)	=	0.85
Prob > chi2	=	0.3572

Sources: author computation

Hetroskedastisty can be tested with Breush-Pagan test showing whether there is inconsistency (hetroskedastist) or consistency (homoskedastisty) in the variance of the error terms. The hetroskedastisty problem can be happened if the p-value of the test is below 0.05. However, the above table shows that the pvalue is above 0.05 which is 0.3572 meaning model is free from hetroskedastisty problem.

#### c) Hypothesis testing

ROA	Coef.	Std.Err	t	p>t	[95% Conf.	Interval ]	
CIEP	2006741	.0342456	-5.86	0.000*	268752	1325962	
NAPW	.0456464	.0163123	2.80	0.006*	.0780742	0132186	
CACL	.0049059	.0175293	0.28	0.780	0299413	.0397531	
CS	1.35e-07	1.98e-07	0.68	0.024*	-2.59e-07	5.28e-07	
PG	.0946792	.0224599	4.22	0.000*	.0500304	.139328	
PCTA	0315101	.0664695	-0.47	0.637**	163647	.1006268	
COE	0394648	.0174286	-2.26	0.026*	0741118	0048178	
						.0603932	
					.(	.0039419	
LEV	0348023	.0147505	-2.36	0.021*	0641253	0054794	
GDP	4.518175	.5641255	8.01	0.000*	5.639619	3.39673	
INF	0654974	.0391418	-1.67	0.098**	1433087	.0123139	
cons	.7131401	.0750059	9.51	0.000*	.5640332	.862247	
—							
Number o	Number of obs = 99						
F(12, 86	6)= 13.04						
Prob>F=	.000	0					
R-square	= 0.645	4					
Adj-R-squ	uare= 0.595	9					
'							

Table 4.3 : Hypothesis testing

\*significant at 1% level, \*\* significant at 10% level

Sources: author computation

As Morgan et al (2004) stated the adjusted Rsquare in the above table indicates how well the model variance explained. The adjusted R-square nearest to 1 is an indication that the model is strongly explained by the variables included in the study whereas the Adjusted R-square nearest to 0 is indicating that the model is not strongly explained by the variables used in the study. The above table shows that the adjusted R-square is 59.59% (0.5959), meaning 59.59% (0.5959) of profitability of insurance industry in Ethiopia is explained by under writing risk (CIEP), solvency ratio (NAPW), liquidity (CACL), company size (CS), premium growth (PG), re-insurance dependency (PCTA), leverage (IEV), technical provision (COE), tangibility of assets, age of the company, growth rate of GDP (GDP), and inflation (INF).

This study found that under writing risk, technical provision, leverage and inflation have negative and significant effect whereas premium growth, age of the company, solvency ratio and GDP have statically positive and significant effect on the profitability of Ethiopian insurance industry. Despite, the study found that liquidity, re-insurance dependency, tangibility of assets and company size have no significant effect on the profitability of insurance industry in Ethiopia. From this it is possible to conclude that the most important determinants of Ethiopian insurance industry are under writing risk, solvency ratio, technical provision, leverage, inflation, premium growth, age of the company and GDP.

The **regression result in the** table 4.3 shows the relationship between profitability as proxied by Return

on Assets (ROA) and age of the company is positive and significant (**p-value of** 0.0000) at 1 percent confidence interval. This is an indication that when the age of the companies increases, its return on assets will also raise. As a result the first hypothesis (H1) that age has a positive and significant effect on profitability of Ethiopian insurance industry's profitability is not rejected. The result of this study is similar with the result of Athanasoglou *et al.* (2005), Yuqi li (2007), Malik (2011), Swiss Re (2008), and Brown (2007) and contradicted with the result of previous study conducted by Hamadan (2008).

It is shown in the table above that size of the company has a positive and statistically significant (p-value of 0.024) effect on the Ethiopian insurance industry's profitability. As a result the second hypothesis that size has positive and significant effect on insurance industry's profitability in Ethiopia is not rejected. The result of this research similar with the previous studies done by Hardwick and Adams (1999), Swiss Re (2008), Malik (2011), Flamini et.al (2009), Athanasoglou *et al.*, (2005), Yuqi Li (2007).

Leverage and return on assets of Ethiopian insurance industry have negative and significant (with pvalue of 0.021) relationship. It was hypothesized that leverage has negative effect on the profitability of Ethiopian insurance industry. Thus the third hypothesis is not rejected showing that while the leverage of companies increased, the profitability of the industry will move to the opposite direction. This result tells us while the insurance companies increase their debt (if the insurance companies operate with huge debt), the
profitability of the industry will significantly falls. This result was also proved by Renbao Chen and Kie Ann Wong (2004), Hamadan Ahamed Ali Al-Shami (2008), Hifza Malik (2011), Sylwester Kozak (2011), Swiss Re (2008) and Flamini et.al (2009) and contradict with the study of Mirie Mwangi and Jane Wanjugu Murigu (2015), which found no relationship between leverage and Kenyan insurance industry's profitability.

The regression in the above table 4.3 indicated that tangibility of assets have negative and insignificant effect on the profitability of insurance industry in Ethiopia. Even if there is a negative relationship between return on assets of insurance industry in Ethiopia and tangibility of assets, their relationship is not significant, hence, the fourth hypothesis is rejected. The result of this study on this variable is consistent with the result of Naveed Ahmed et.al (2011) and Yuqi Li (2007) and contracted with the result of the study conducted by Hafiz Malik (2011).

It is found that liquidity has positive and insignificant effect on ROA of Ethiopian insurance industry. Thus the fifth hypothesis that liquidity and profitability of insurance industry in Ethiopia are negatively related is rejected. This result is similar with (Naveed Ahmed et.al, 2011), Daniel Mehari and Tilahun Aemiro, 2013), Adams and Buckle (2000) and Bilal et al (2013) and is not consistent with empirical results (Chen and Wong, 2004), and Valentina et al, 2009).

The solvency ratio and the return on assets of Ethiopian insurance industry have positive and significant relationship. This indicated that when the solvency of insurance companies is strong, the profitability of the industry will be increased. Thus, the hypothesis that solvency ratio has positive and significant effect on profitability of Ethiopian insurance industry, is not rejected. The result of this study is similar with result of the research previously conducted by Ana-Maria BURCA and Ghiorghe BATRÎNCA (2014), Shiu (2004), (B. Charumathi (2013). This is an indication that financially sound insurance companies are be able to maximaize their profitability.

The premium growth is considered the major determinants of insurance industry profitability everywhere. Similarly, the result of this study indicated that premium growth is positive and significant effect on the profitability of Ethiopian insurance industry and the seventh hypothesis is not rejected. The same result was proved in previous studies conducted by Emine Öner Kaya (2015), and contradicted with the result of Ana-Maria and Ghiorghe (2014) and B. Charumathi (2012) that found negative and significant relationship between profitability of Turkish insurance industry and premium growth.

It is proved from the above regression analysis that the re-insurance dependency doesn't have a significant effect on the profitability of Ethiopian insurance industry. Therefore, the eighth hypothesis is rejected. Basically, purchasing a re-insurance is for the purpose of providing protection against catastrophic losses. However, in Ethiopia, the catastrophic losses are not the headache of insurance companies. Hence, the insurance companies are be able to determine appropriate ceding level and this can lead to а reduction in cost of reinsurance. Due to this, the relationship between re-insurance dependency and ROA of insurance industry in Ethiopia, have no significant relationship. The result of this study is similar with the result conducted by Chen-Ying Lee (2014) and contradicted with the result of Hafiza and Mobeen (214), which found positive and significant relationship between re-insurance dependency and insurance companies' profitability and Hsu-Hua and Chen (2012) and Olajumoke (2012) that found significant but negative relationship between re-insurance dependency and ROA.

As it is found in this study, underwriting risk (CIEP) has negative and significant effect on Ethiopian insurance industry's profitability. Hence, the ninth hypothesis is not rejected. This result is not consistent with the previous study conducted in Kenya by Mirie Mwangi and Jane Wanjugu Murigu (2015), which found no relationship between Kenyans insurance industry's profitability and under writing risk and Chen-Ying Lee (2014) which found positive relationship between underwriting risk and profitability of insurance industry of Taiwan but consistent with Ana-Maria and Ghiorghe (2014).

The regression result of this study revealed that the effect of technical provision on the Ethiopian insurance industry's profitability is significant and negative. Thus, the tenth hypothesis that technical provision has a negative and significant effect on profitability of Ethiopian insurance industry not rejected. Technical provisions is used to cover (for general insurance) provisions for items such as unearned premiums, unexpired risks, claims outstanding (whether or not reported), equalisation. Which means insurance companies set aside some sort of funds for contingencies or companies' solvency, which is not used for immediate operation of the companies. Therefore, this can have a negative effect on the profitability of insurance companies, which is similarly proved in this study.

As it is indicated in previous studies, the economic growth proxied by GDP is the most important determinants of insurance industry's profitability in the world. Likewise, it is proved in this study that GDP has positive and significant effect on Ethiopian insurance industry's profitability. This shows that the economic growth is a favorable factor for the rise of profitability of insurance industry in Ethiopia. Thus, the eleventh hypothesis is not rejected. The result of this study was similarly proved in previous studies done by Gustina and Abdullah (2012), Beck and Webb (2003), Çelik and Kayali (2009), Redzuan (2011).

In most cases, inflation is a macro-economic challenge for the development and profitability of insurance industry in any country. However, the result of this study indicated that inflation is not a significant factor of Ethiopian insurance industry's profitability and

d) Correlation between ROA and independent variables

Table 4.4 : Correlation coefficients

	roa	ciep	napw	cacl	cs	pg	pcta	coe	tangib~y	age	lev	gdp	jnf
roa	1.0000												
ciep	-0.3244	1.0000											
napw	-0.1051	-0.5289	1.0000										
cacl	-0.0080	-0.4168	0.6844	1.0000									
cs	0.0439	0.0095	-0.0455	-0.0618	1.0000								
pg	0.2326	0.0699	-0.1935	-0.1036	-0.0953	1.0000							
pcta	0.3624	-0.3033	-0.1337	-0.1016	-0.0441	0.1802	1.0000						
coe	-0.0454	0.4077	-0.7125	-0.4921	0.0799	-0.0844	0.2271	1.0000					
tangiblity	0.1981	-0.0688	-0.0282	-0.0102	0.2469	-0.1346	0.1242	0.0791	1.0000				
age	0.2095	0.1492	-0.2184	-0.0165	-0.0085	-0.0597	0.3710	0.2475	0.0921	1.0000			
lev	-0.1096	-0.2748	0.0177	-0.0440	0.1408	0.0308	0.3947	0.0219	0.0419	0.3272	1.0000		
gdp	0.4963	-0.0555	0.1990	0.2296	-0.0918	0.1123	-0.2239	-0.2483	-0.2765	-0.0002	-0.0662	1.0000	
jnf	- 0.0659	0.1898	-0.2319	-0.2260	- <b>0.074</b> 3	0.2633	0.1108	0.2064	-0.0956	-0.0013	0.0762	-0.3335	1.0000

Sources: author computation

As per the above table, the correlation coefficient between ROA and underwriting risk (CIEP), solvency ratio (NAPW), liquidity (CACL), technical provision (COE), leverage and inflation is negative indicating that an increase in underwriting risk (CIEP), solvency ratio (NAPW), liquidity (CACL), technical provision (COE), leverage and inflation underwriting risk (CIEP), solvency ratio (NAPW), liquidity (CACL), technical provision (COE), leverage and inflation will lead to a decrease in the profitability of insurance companies of Ethiopia. on the other hand, the correlation coefficient between ROA and company size (CS), premium growth (PG), re-insurance dependency (PCTA), age of the company, tangibility of assets and GDP is positive showing an increase in company size (CS), premium growth (PG), re-insurance dependency (PCTA), age of the company, tangibility of assets and GDP company size (CS), premium growth (PG), reinsurance dependency (PCTA), age of the company, tangibility of assets and GDP will lead to an increase in the profitability of Ethiopian insurance industries' profitability.

#### V. CONCLUSION AND RECOMMENDATION

Profitability is the major objectives of financial management because one goal of financial management is to maximize the owner's wealth. This study attempts to examine the effects of firm specific factors (age of company, size of company, leverage ratio, liquidity ratio, premium growth, technical provision, underwriting risk, solvency, re-insurance dependency and tangibility of assets) and macroeconomic factors

© 2016 Global Journals Inc. (US)

(GDP and Inflation) on profitability of Ethiopian insurance industry. nine insurance companies from the total of 17 companies established before 2008 were included in the study. Secondary data that was obtained from the financial statements of insurance companies; financial publications of NBE are the major sources of the study.

the tenth hypothesis is rejected. This result is similar with

previous studies Chen-Ying Lee, 2014), and Doumpos

et al., 2012), and is not consistent with the previously

conducted research (Doreen, 2013) and Y. Shiu (2000),

between

that found no significant relationship

profitability of insurance industry and inflation.

This study found that under writing risk, technical provision, leverage and inflation have negative and significant effect whereas premium growth, age of the company, solvency ratio and GDP have statically positive and significant effect on the profitability of Ethiopian insurance industry. However, the study found that liquidity, re-insurance dependency, tangibility of assets and company size have no significant effect on the profitability of insurance industry in Ethiopia.

This study suggests that insurance companies should critically consider underwriting risk and should minimize the accumulation used for technical provision and the level of leverage.

#### References Références Referencias

- 1. Aburime T., (2008). Determinant of Bank Profitability: Company Level Evidence from Nigeria, Working Paper, Department of Banking and Finance, University of Nigeria.
- 2. Adams, M., & Buckle, M., (2000). The determinants of operational performance in the Bermuda insurance market: European business management school working paper.
- 3. Ana-Maria BURCA and Ghiorghe BATRÎNCA (2014). The Determinants of Financial Performance in the

Romanian Insurance Market. International Journal of Academic Research in Accounting, Finance and Management Sciences, Vol. 4, No.1, January 2014, pp. 299–308, E-ISSN: 2225-8329, P-ISSN: 2308-0337

- Athanasoglous P. *Et Al.*, (2005). Bank-Specific, Industry- Specific and Macroeconomic Determinants of Bank Profitability, Bank of Greece Working Paper, No. 25
- 5. B. Charumathi (2013). The determinants of solvency margin of Indian general insurers.
- 6. B. Charumathi (2012). On the Determinants of Profitability of Indian Life Insurers – An Empirical Study. *Proceedings of the World Congress on Engineering 2012 Vol I. WCE 2012, July 4 - 6, 2012, London, U.K.*
- Bilal, J., Khan S. &Tufail S. A., (2013). Determinants of profitability panel data evidence from insurance sector of Pakistan: *Elixir International Journal*, Available at www.elixirpublishers.com accessed on March 23, 2013
- Beck, T., & Webb, I. (2003). Economic, Demographic, and Institutional Determinants of Life Insurance Consumption Across Countries. World Bank Economic Review, vol. 17 No 1.
- 9. Chen-Ying Lee (2014). The effects of firm specific factors and macroeconomics on profitability of property-liability insurance industry in Taiwan. *Asian Economic and Financial Review, 2014, 4(5):* 681-691
- Daniel Mehari and Tilahun Aemiro, (2013). Firm specific factors that determine insurance companies' performance in Ethiopia. European Scientific Journal, vol.9, No.10 ISSN: 1857 – 7881
- 11. Desheng Wu Z., Sandra V. & Lianga, (2007). Simultaneous Analysis of Production and Invest ment Performance of Canadian Life and Healt h Insurance Companies", Computers and Operations Research, Vol. 34, Issue. 1.
- 12. Doreen Umotho Murungi (2013). Relationship between macroeconomic variables and financial performance of insurance companies in Kenya
- 13. Emine Öner Kaya (2015). The Effects of Firm-Specific Factors on the Profitability of Non-Life Insurance Companies in Turkey. *International Journal of Financial Studies, ISSN 2227-7072. www.mdpi.com/journal/ijfs*
- 14. Gatzlaff, K., (2009). Dimensions of property-liability insurer performance. Florida State University Doctoral Dissertation.
- 15. Gustina & Abdullah (2012). Analysis of Demand for Family Takaful and Life Insurance: A Comparative Study in Malaysia *Journal of Islamic Economics*, *Banking and Finance*, Vol. 8 No. 4
- Hamadan Ahamed Ali Al-Shami, (2008). Determinats of insurance companies' profitability in UAE, Universiti Utara, Malasiya.

- 17. Hafiza and Mobeen (214). Empirical analysis of reinsurance utilisation and dependence with respect to its impact on the performance of domestic nonlife stock insurance companies operating in the private sector of Pakistan. *International Journal of Financial Services Management 01/2014; 7(2):95 - 112. DOI: 10.1504/IJFSM.2014.063946*
- 18. Harrington C. (2005). The Effect of Competitive structure on the relationship between leverage and profitability.
- 19. Hendon Redzuan(2011). Analysis Of The Demand For Life Insurance And Family Takaful, *Universiti Teknologi Mara Doctor of Philosophy*
- 20. Hifza, M., (2011). Determinants of insurance companies' profitability: An analysis of insurance sector of Pakistan, vol. 1, no. 3
- Hsu-Hua and Chen (2012). An Analysis of Reinsurance and Firm Performance: Evidence from the Taiwan Property-Liability Insurance Industry. Graduate Institute of Management Sciences, No. 151, Yi ngzhuan Rd., Tamsui Dist., New Taipei City, 886, Taiwan R.O.C.
- 22. Jay Angoff Roger Brown (2007). An Analysis of the Profitability and Performance of the Michigan Auto Insurance Market. *McCarty Street Jefferson City, MO* 65101
- Kozak S., (2011). Determinants of Profitability of Non-Life Insurance Companies in Poland during Integration with the European Financial System, Volume 14, Issue 1. Available Online: Http://Www.EJPAU.Media.Pl.
- 24. Lee, H.H. and C.Y. Lee, (2012). An analysis of reinsurance and firm performance: Evidence from the Taiwan property-liability insurance industry. *The Geneva Papers on Risk and Insurance-Issues and Practice*, 37(3): 467-484.
- 25. Michael Doumpos, Chrysovalantis Gaganis and Fotios Pasiouras, (2012). Estimating and Explaining the Financial Performance of Property and Casualty Insurers: A Two-Stage Analysis. *The Business and Economics Research Journal, Volume 5, Issue 2,* 155-170.
- Mirie Mwangi and Jane Wanjugu Murigu (2015). The Determinants of financial performance in general insurance companies in Kenya. *European Scientific Journal, vol.*11, No.1 ISSN: 1857 – 7881, e - ISSN 1857-7431
- 27. Naveed, A., Zulfqar, A., Ahmad, U., (2011). Determinants of Performance: A Case of Life Insurance Sector of Pakistan, International Research Journal of Finance and Economics, Eurojournals Publishing, Inc.
- Pervan, M. and K.T. Pavá, (2010). Determinants of insurance companies' profitability in Croatia. The Business Review Cambridge, 16(1): 231-238.
- 29. Olajumoke Olaosebikan (2012). The Determinants of the Profitability of Micro-Life insurers in Nigeria.

- 30. Rebao Chen and Ann Wong, (2004). Determinants of Financial Health of Asian Insurance Companies, *the Journal of Risk and Insurance Volume* 71 Number 3
- 31. Swiss Re (2008), Profitability of Non-Life Insurance Industry, Egypt, Available At Www.Insureegypt.Com.
- 32. Sibel Çelik and Mustafa Mesut Kayali (2009). Determinants of demand for life insurance in European countries. Problems and Perspectives in Management, Vol. 7, Issue 3
- 33. William H. Greene and Dan Segal, (2004). Profitability and Efficiency in the U.S Life Insurance Industry, *Journal of Productivity Analysis, Kluwer Academic Publisher,* Netherlands
- 34. Wright, K. M., (1992). The Life Insurance Industry in the United States an Analysis of Economic and Regulatory Issues", *Country Economics Department the World Bank Policy Research Working Paper* 857.
- 35. Valentina Flaminin. Et.Al,. (2009). Determinants of Commercial Bank Profitability In Sub-Saharan Africa, International Monitory Fund (IMF) Working Paper, Wp/09/15
- 36. Yung-Ming Shiu (2004). Determinants of UK general insurance company performance. *British Actuarial Journal*, 10, (5), 1079-1110
- Yuvaraj, S. and Abate G., (2013). A study on the performance of insurance companies In Ethiopia. *International Journal of Marketing, Financial Services* & *Management* ResearchVol.2, No. 7, available at www.indianresearchjournals.com



GLOBAL JOURNAL OF MANAGEMENT AND BUSINESS RESEARCH: C FINANCE Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

### Performance of Monthly Income Scheme in Mutual Fund Industry in India

By N.Dharmalingam & Dr. K. Balanaga Gurunathan

Alliance University

*Abstract-* Mutual funds are an integral part of the stock market. It has become the investment avenue for large number of investors in the past 10 years. Also the stock market volatility is high in these years. Thus in order to analyze the performance of top funds under importance schemes, a research has been done. Ten mutual funds in the income fund category were selected based on their returns. The main focus of this research is to find the risk and return features and study the performance of the funds and to compare it with the market return. The research is limited NAV data for ten income funds available for (2005-2015). The findings will be useful to bring out insight into investment avenues.

*Keywords: mutual fund, risk and return. GJMBR - C Classification : JEL Code: F65* 



Strictly as per the compliance and regulations of:



© 2016. N.Dharmalingam & Dr. K. Balanaga Gurunathan. This is a research/review paper, distributed under the terms of the Creative Commons Attribution-Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

# Performance of Monthly Income Scheme in Mutual Fund Industry in India

N.Dharmalingam  $^{\alpha}$  & Dr. K. Balanaga Gurunathan  $^{\sigma}$ 

Abstract- Mutual funds are an integral part of the stock market. It has become the investment avenue for large number of investors in the past 10 years. Also the stock market volatility is high in these years. Thus in order to analyze the performance of top funds under importance schemes, a research has been done. Ten mutual funds in the income fund category were selected based on their returns. The main focus of this research is to find the risk and return features and study the performance of the funds and to compare it with the market return. The research is limited NAV data for ten income funds available for (2005-2015). The findings will be useful to bring out insight into investment avenues.

Keywords: mutual fund, risk and return.

#### I. INTRODUCTION

here are a number of investment opportunities available to an investor. Each of these investments has its own risk and return features. An investor must learn to analyze and measure the risk and return of the portfolio. The mutual fund industry plays a significant role in the development of the economy. Its buoyant growth leads to lower intermediate costs, more efficient financial markets, and increased vibrancy of the capital markets and higher local ownership of financial assets. If retail investment is directed through the mutual fund route, it will lead to greater wealth creation in the long run. Thus, the industry can be one of the causative factors for a healthy economy. The Indian mutual funds business is expected to grow significantly in the coming years due to a high degree of transparency and disclosure standards comparable to anywhere in the world, through there are many challenges that need to be addressed to increase net mobilization of funds in the sector.

#### II. LITERATURE REVIEW

Michael C. Jensen, Harvard Business School (2002), in his paper a risk adjusted measure of portfolio performance that estimates how much a manger's forecasting ability contributes to the funds return. The measure is based on the theory of the pricing of Capital assets by Sharpe (1964) Lintner (1965) and Treynor. Timotej Jagric, Boris Podobnik, Sebastjan, Strasek and Vita Jagric (2007), studied the mutual fund industry and

Author o: M.Com., MBA., M.Phil., Ph.D., Professor in Finance, Alliance School of Business, Alliance University, Bangalore - 562 106 e-mail: balanagagurunathan@yahoo.com

apply various test to evaluate the Performance capacity of mutual funds. They used performance measure to evaluate funds and also they rank them according to the results. Arnod L. Redman, N.S. Gullet and Herman Manakyan (2000), examines the risk adjusted returns using Sharpe Index, Treynors Index, Jensens Alpha for five portfolios of International mutual funds and for three time period: 1985-1994, 1985-1989 and 1990-94. The bench mark for comparison was the US market provided by the Vanguand Index 500 mutual funds and a portfolio of funds that invest solely in US stocks. J. Cal, KC Chan and T. Yamada, 'The performance of Japanese mutual funds', analyze the performance of Japanese open-type stock mutual funds for the 1981-1992 period. David Blake (2003), reviewed the extensive empirical literature on mutual fund performance and also conducted an empirical analysis of the performance of a large sample of UK unit trusts. S.P. Kothari, Jerold B, Warner (2005), indicates standard model fund performance measures, using simulated funds whose characteristics mimic actual funds. Shanmugham (2000) conducted a survey of 201 individual investors to study the information sourcing by investors, their perceptions of various investment strategy dimensions and the factors motivating share investment decisions, and reports that among the various factors, psychological and sociological factors dominated the economic factors in share investment decisions. Madhusudhan V Jambodekar (1996)conducted a study to assess the awareness of MFs among investors, to identify the information sources influencing the buying decision and the factors influencing the choice of a particular fund. The study reveals among other things that Income Schemes and Open Ended Schemes are more preferred than Growth Schemes and Close Ended Schemes during the then prevalent market conditions. Investors look for safety of Principal, Liquidity and Capital appreciation in the order of importance; Newspapers and Magazines are the first source of information through which investors get to know about MFs/Schemes and investor service is a major differentiating factor in the selection of Mutual Fund Schemes. Sujit Sikidar and Amrit Pal Singh (1996) carried out a survey with an objective to understand the behavioural aspects of the investors of the North Eastern region towards equity and mutual funds investment portfolio. The survey revealed that the salaried and self-employed formed the major investors

Author a: Research Scholar, Bharathiyar University, Coimbatore. e-mail: dharma1405@gmail.com

in mutual fund primarily due to tax concessions. UTI and SBI schemes were popular in that part of the country then and other funds had not proved to be a big hit during the time when survey was done. Syama Sunder (1998) conducted a survey to get an insight into the mutual fund operations of private institutions with special reference to Kothari Pioneer. The survey revealed that awareness about Mutual Fund concept was poor during that time in small cities like Visakhapatnam. Agents play a vital role in spreading the Mutual Fund culture; open-end schemes were much preferred then; age and income are the two important determinants in the selection of the fund/scheme; brand image and return are the prime considerations while investing in any Mutual Fund. In India, one of the earliest attempts was made by 64 Pacific Bus. Anjan Chakarabarti and Harsh Rungta (2000) stressed the importance of brand effect in determining the competitive position of the AMCs. Their study reveals that brand image factor, though cannot be easily captured by computable performance measures, influences the investor's perception and hence his fund/ scheme selection. Hirshleifer (2001) categorized different types of cognitive errors that investors make i.e. self-deception, occur because people tend to think that they are better than they really are; heuristic simplification, which occurs because individuals have limited attention, memory and processing capabilities; disposition effect, individuals are prone to sell their winners too quickly and hold on to their losers too long In this paper, an attempt is made by the author, mainly to study preference of investors for mutual funds and their performance evaluation.

#### III. Statement of Problem

Mutual Funds have not only contributed to the India growth story but have also helped families tap into the success of Indian Industry. As information and awareness is rising more and more people are enjoying the benefits of investing in mutual funds. The success of a mutual fund depends upon the confidence of the investors. But most of the investors are lacking in selection of right mutual funds for their regular commitments. All the problems related to the investors are, lack of awareness and poor after sales service to the investors. The investors believed, so far that the mutual funds promoted by regulated bodies and nationalized banks are guaranteed by the Central Govt. The majority of the new investors don't understand the concept, operations and advantages of investment in mutual funds before investing. This research paper has mainly focused how to evaluate the performance of various income scheme mutual funds in India.

#### a) Objectives

The main focus of this research is to analyze the risk, return parameters of the top performing income

funds, rank the funds based on various measures, to compare the performance of the fund returns with the market returns, to analyze the stock selection ability and market timing ability of the fund managers of the top performing funds.

#### IV. Research Methodology

#### a) Treynor's performance index

Treynor (1965) was the first researcher for developing a composite measure of portfolio performance. He measures portfolio risk with beta, and calculates portfolio's market risk premium relative to its beta:

Where:

$$Treynor = \frac{\left(R_P - R_f\right)}{\beta_P}$$

Ti = Treynor's performance index Rp = Portfolio's actual return during a specified time period

Rf = Risk-free rate of return during the same period  $\beta p$  = beta of the portfolio

Whenever Rp> Rf and  $\beta p > 0$  a larger T value means a better portfolio for all investors regardless of their individual risk preferences. In two cases we may have a negative T value: when Rp < Rf or when  $\beta p < 0$ . If T is negative because Rp < Rf, we judge the portfolio performance as very poor. However, if the negativity of T comes from a negative beta, fund's performance is superb. Finally when Rp- Rf, and  $\beta p$  are both negative, T will be positive, but in order to qualify the fund's performance as good or bad we should see whether Rp is above or below the security market line pertaining to the analysis period

#### b) Sharpe's Performance index

Sharpe (1966) developed a composite index which is very similar to the Treynor's measure, the only difference being the use of standard deviation, instead of beta, to measure the portfolio risk, in other words except it uses the total risk of the portfolio rather than just the systematic risk:

Where:

Sharpe = 
$$\frac{\left(R_P - R_f\right)}{\sigma_P}$$

 $Si = Sharpe performance index \sigma p = Portfolio standard deviation$ 

This formula suggests that Sharpe prefers to compare portfolios to the capital market line (CML) rather than the security market line (SML). Sharpe index, therefore, evaluates funds performance based on both rate of return and diversification (Sharpe 1967). For a completely diversified portfolio Treynor's and Sharpe indices would give identical rankings.

#### c) Jensen's Alpha

Jensen (1968), on the other hand, writes the following formula in terms of realized rates of return, assuming that CAPM is empirically valid:

Jp = Rp - Rf = 
$$\alpha$$
 +  $\beta$  (Rm - Rf)  
Jensen =  $\alpha_P = R_P - \left[ R_f + \beta_P \left( R_M - R_f \right) \right]$ 

Jensen uses  $\alpha$  as his performance measure. A superior portfolio manager would have a significant positive  $\alpha$  value because of the consistent positive residuals. Inferior managers, on the other hand, would have a significant negative  $\alpha$ . Average portfolio manager

having no forecasting ability but, still, cannot be considered inferior would earn as much as one could expect on the basis of the CAPM.

Return performance, Return is measured by its 5 to 10 years Sharpe Ratio. For this, initially the unadjusted monthly return has been calculated for each mutual fund through rate of return measure as:

$$Return = [NAVt - NAV(t-1)] / NAV(t-1)$$
(i)

Where,

NAVt = Net Asset Value at time t NAV(t-1) = The corresponding value at time t-1 rit = Return for mutual fund i at time t.

From the above monthly returns, yearly return for each fund has been calculated as:

$$Rit = [(1+r1) \times (1+r2) \times (1+r3) \dots \times (1+rn)] - 1$$
(ii)

Where,

Rit = Yearly return for fund i for year t rn = Monthly return for the fund for month n

 $n = 1, 2, \dots, 12$  the number of months for calculating yearly return and

 $i = 1, 2, \dots$  160, the number of mutual funds taken in the study Then, the risk adjusted performance of each fund has been calculated by using Sharpe performance measure (1966) as explained below:

$$Sp = (Ri - Rf) / \sigma$$
 (iii)

Where,

Ri – mean return on fund i,

Rf - mean risk free rate of return measured by T-364 treasury bill and

 $\sigma i$  – standard deviation of returns for fund i.

d) Analysis

Table 1.1.1 : Return of IDFC Government Securities Fund - Investment Plan - Regular Plan

	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	11.06	10.12	10.52	11.56	13.06	13.89	16.18	17.15	12.94
Total Return (%)	-	-8.47	3.97	9.89	12.97	6.35	16.46	5.99	6.73

From the above table it can be inferred that the IDFC government securities fund earned maximum return of Rs. 16.46 per cent in the year 2014 and the

fund has given minimum return of Rs. -8.47 per cent during the year 2009. The overall return of the IDFC government securities fund was 6.73.

Table 1.1.2 : Performance of evaluation of IDFC Government Securities Fund - Investment Plan - Regular Plan.

	Mean	Std Dev	Beta	Alpha	Sharpe	Treynor's	Jensen's	Sortino
Fund	6.73	4.42	0.69	0.9	1.350962	8.653986	1.304348	0.59

From the above table the performance indicators of IDFC fund is 0.9. The market sensitivity index (Beta) is 0.69. The unsystematic risk of IDFC is 4.42. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 1.35, 8.65, 1.30 and 0.59 respectively.

History	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	10.26	10.82	11.45	125.07	137.4	150.64	169.42	183.86	109.5389
Total Return (%)	-	5.47	5.81	9.25	9.85	9.63	12.47	8.52	7.83375

1.2.1 Return of Birla Sun Life Treasury Optimizer Fund - Regular Plan from 2008 to 2014

The above table shows that the Brila Sun Life Treasury Optimizer fund earned maximum return of Rs. 12.47 per cent in the year 2014 and the fund has given minimum return of Rs. 5.47 per cent during the year 2009. The overall return of the IDFC government securities fund was 7.83.

1.2.2. Performance evaluation of Birla Sun	Life Treasury Optimizer Fund - Regular Plan
--	---

Risk Measures (%)	Mean	Std Dev	Sharpe	Treynors	Jensens	Sortino	Beta	Alpha
Fund	7.83375	1.34	5.801306	18.07849	6.976744	3.65	0.43	3

From the above table that the performance indicators of IDFC fund is 3.0. The market sensitivity index (Beta) is 0.43. The unsystematic risk of IDFC is

1.34. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 5.80, 18.07,6.97 and 3.65 respectively.

	-					
131	Return	of ICICI	Prudential	Banking	& PSU	Debt Fund
1.0.1	riotann	0110101	riadontial	Barnang	0.100	Dobt i ana

History	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	10.6	11.57	12.59	13.72	15.16	16.47	13.35167
Total Return (%)	-	9.12	8.87	8.93	10.5	8.64	9.212

The above table shows that the ICICI Prudential Banking & PSU Debt fund earned maximum return of Rs. 10.5 per cent in the year 2014 and the fund has given minimum return of Rs. 8.64 per cent during the year 2015. The overall return of the IDFC government securities fund was 9.212.

1.3.2 Performance evaluation of ICICI Prudential Banking & PSU Debt Fund

Risk Measures (%)	Mean	Beta	Alpha	Std Dev	Sharpe	Jensens	Treynors	Sortino
Fund	9.212	0.83	2.26	2.21	4.141176	2.722892	11.02651	0.85

The above table shows that the performance indicators for ICICI Prudential Banking & PSU Debt Fund is 2.26. The market sensitivity index (Beta) is 0.83. The unsystematic risk of ICICI is 2.21. The sharpe

performance index, Treynors performance index, Jensen's performance index and sortino ratio are 4.14, 11.02, 2.72 and 0.85 respectively.

1.4.1 Return of HDFC Medium Term Opportunities Fund from 2010 to 2015

History	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	10.35	11.25	12.43	13.37	14.83	16.11	16.35
Total Return (%)	-	8.68	10.49	7.56	10.95	8.61	9.258

The above table shows that the HDFC Medium Term Opportunities fund earned maximum return of Rs. 10.95 per cent in the year 2014 and the fund has given minimum return of Rs. 7.56 per cent during the year 2009. The overall return of the HDFC Medium Term Opportunities fund was 9.258.

1.4.2	Performance	evaluation	of HDFC	Medium <sup>-</sup>	Term	Opportunities	s Fund
-------	-------------	------------	---------	---------------------	------	---------------	--------

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	9.258	2.39	0.92	2.37	3.84854	9.99783	2.57609	0.8

The above table shows that the performance indicators of HDFC Medium Term Opportunities fund is 2.37. The market sensitivity index (Beta) is 0.92. The unsystematic risk of HDFC Medium Term Opportunities

Fund is 2.39. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 3.84, 9.99,2.57 and 0.8 respectively.

History	2011	2012	2013	2014	2015	Average
NAV (Rs)	10.82	11.78	1249.76	1376.12	1490.06	827.708
Total Return (%)	-	8.86	6.11	10.11	8.28	8.34

1.5.1 Return of Religare Invesco Medium Term Bond Fund from 2011 to 2015

The above table shows that the of Religare Invesco Medium Term Bond fund earned maximum return of Rs. 10.11 per cent in the year 2014 and the

fund has given minimum return of Rs. 6.11 per cent during the year 2013. The overall return of the IDFC government securities fund was 8.34.

1.5.2 Performance evaluation of Religare Invesco Medium Term Bond Fund

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	8.34	1.92	0.72	1.61	4.3125	11.5	2.236111	0.58

The above table shows that the performance indicators of of Religare Invesco Medium Term Bond fund is 1.61. The market sensitivity index (Beta) is 0.72. The unsystematic risk of Religare Invesco Medium Term Bond is 1.92. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 4.31, 11.5, 2.23 and 0.58 respectively.

1.6.1	Return	of Tata	Balanced	Fund -	Regular	Plan	from	2006	to	2015
1.0.1	noturn	orraid	Balariooa	i una	riogaiai	i iuii	110111	2000	ιU	2010

History	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	average
NAV (Rs)	49.33	76.64	42.61	74.58	85.67	75.38	98.4	105.82	158.32	169.35	93.61
Total Return (%)	34.95	55.35	-44.4	75.04	14.87	-12.02	30.55	7.54	49.61	6.97	21.846

The above table shows that the of Tata Balanced fund earned maximum return of Rs. 49.61 per cent in the year 2012 and the fund has given minimum return of Rs. -44.4 per cent during the year 2008. The overall return of the Tata Balanced fund was 21.84.

1.6.2 Performance evaluation of Tata Balanced Fund - Regular Plan

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	21.846	12.72	1.01	9.25	1.712736	21.5703	9.158416	1.3

The above table shows that the performance indicators of Tata Balanced fund is 9.25. The market sensitivity index (Beta) is 1.01. The unsystematic risk of Tata Balanced is 12.72. The sharpe performance index,

Treynors performance index, Jensen's performance index and sortino ratio are 1.71, 21.57, 9.15 and 1.3 respectively.

1.7.1 Return HDFC	Childrens Gift Fund	<ul> <li>Investment Plan</li> </ul>	from 2006 to 2015

History	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	24.68	32.45	19.41	32.54	42.94	39.72	50.58	57.13	81.76	83.75	46.496
Total Return (%)	12.4	31.45	- 40.17	67.62	31.96	-7.49	27.32	12.96	43.1	2.44	18.159

The above table shows that the HDFC Childrens Gift fund earned maximum return of Rs. 67.62 per cent in the year 2009 and the fund has given minimum return

of Rs. -7.49 per cent during the year 2011. The overall return of the HDFC Childrens Gift fund was 18.15.

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	18.159	10.61	0.84	7.98	1.705844	21.54643	9.5	1.26

The above table shows that the performance indicators of HDFC Childrens Gift fund is 7.98. The market sensitivity index (Beta) is 0.84. The unsystematic risk of HDFC Childrens Gift is 10.61. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 1.70, 21.54, 9.5 and 1.26 respectively.

History	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	35.37	52.48	29.04	47.73	53.7	41.76	56.39	63.08	90.35	96.99	37.833
Total Return (%)	33.93	48.37	-44.66	64.36	12.51	-22.23	35.03	11.86	43.24	7.36	12.2245

#### 1.8.1 Return of SBI Magnum Balanced Fund from 2006 to 2015

The above table shows that the SBI Magnum Balanced fund earned maximum return of Rs. 64.36 per cent in the year 2009 and the fund has given minimum return of Rs. 7.36 per cent during the year 2009. The overall return of the SBI Magnum Balanced fund was 12.22.

182 Performance	evaluation	of SBI	Magnum	Balanced	Fund
1.0.21 0101110100	cvaluation		magnum	Dalanceu	runu

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	12.2245	11.03	0.85	8.76	1.102856	14.31118	10.30588	1.43

The above table shows that the performance indicators of SBI Magnum Balanced fund is 8.76. The market sensitivity index (Beta) is 0.85. The unsystematic risk of SBI Magnum Balanced fund is 11.03. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 1.10, 14.31,10.30 and 1.43 respectively.

History	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	32.64	46.56	28.72	43.9	50.4	44.15	54.84	58.48	86	90.16	53.585
Total Return (%)	32.83	42.66	-38.32	52.86	14.82	-12.4	24.21	6.64	47.05	4.85	17.52

The above table shows that the Franklin India Balanced fund earned maximum return of Rs. 52.86 per cent in the year 2009 and the fund has given minimum return of Rs. -12.4 per cent during the year 2011. The overall return of the Franklin India Balanced fund was 17.52.

1.9.2 Performance evaluation of Franklin India Balanced Fund

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	17.52	11.54	0.94	8.3	1.512998	18.57447	8.829787	1.48

The above table shows that the performance indicators of Franklin India Balanced Fund is 8.3. The market sensitivity index (Beta) is 0.94. The unsystematic risk of Franklin India Balanced Fund is 11.54. The sharpe performance index, Treynors performance index, Jensen's performance index and sortino ratio are 1.51, 18.57, 8.82 and 1.48 respectively.

1.10.1 Return of L&T Gilt Fund - Investment Plan

History	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	Average
NAV (Rs)	18.62	19.69	24.01	21.68	22.24	23.24	26.47	28.91	33.63	36.15	25.464
Total Return (%)	0.84	5.74	21.97	-9.71	2.6	4.48	13.92	9.19	16.34	7.48	7.285

The above table shows that the L&T Gilt fund earned maximum return of Rs. 21.97 per cent in the year 2008 and the fund has given minimum return of Rs. -9.71 per cent during the year 2009. The overall return of the L&T Gilt fund was 7.28.

1.10.2 Performance evaluation of L&T Gilt Fun	id - Investment Plan
---	----------------------

Risk Measures (%)	Mean	Std Dev	Beta	Alpha	Sharpe	Treynors	Jensens	Sortino
Fund	7.285	4.56	0.74	2.37	1.58443	9.763514	3.202703	0.98

The above table shows that the performance indicators of L&T Gilt fund is 2.37. The market sensitivity index (Beta) is 0.74. The unsystematic risk of L&T Gilt is 4.56. The sharpe performance index, Treynors

performance index, Jensen's performance index and sortino ratio are 1.58, 9.76, 3.20 and 0.98 respectively.

Funds	Sharpe	Treynors	Jensens	Sortino
IDFC Government Securities Fund	1.350962	8.653986	1.304348	0.59
Birla Sun Life Treasury Optimizer Fund	5.801306	18.07849	6.976744	3.65
ICICI Prudential Banking & PSU Debt Fund	3.848536	9.997826	2.576087	0.8
HDFC Medium Term Opportunities Fund	4.141176	11.02651	2.722892	0.85
Religare Invesco Medium Term Bond Fund	4.3125	11.5	2.236111	0.58
Tata Balanced Fund	1.712736	21.5703	9.158416	1.3
HDFC Childrens Gift Fund	1.705844	21.54643	9.5	1.26
SBI Magnum Balanced Fund	1.102856	14.31118	10.30588	1.43
Franklin India Balanced Fund	1.512998	18.57447	8.829787	1.48
L&T Gilt Fund	1.58443	9.763514	3.202703	0.98

Table 1.11 : Performance Index of Funds

From the above table it can be inferred that Birla Sun Life Treasury optimizer fund had the maximum Sharpe ratio of 5.80 and the fund which has the least value is SBI Magnum Balanced fund with the ratio of 1.10. Birla Sun Life fund was out performing fund among the top income fund in India from 2006-2015. As per the Treyners model the fund which has the highest ratio is Franklin India Balanced fund with a ratio of 18.57 and the least fund is IDFC Govt securities fund with a ratio of 8.65. In the Jensens model SBI Magnum has the highest ratio of 10.31 and the least happens to be IDFC Govt. Securities fund with a ratio of 1.30. And in the Sortino model it was found that Birla Sun Life fund had the highest ratio with 3.65 and the least being Religare Invesco Medium term Bond fund with a ratio of 0.58.

Table 1.12 : Performance of Top ten performing Income mutual Funds from 2006	-2015
--	-------

Fund	Sharpe	Treynors	Jensens	Sortino	Weighted Average	RANK
IDFC Government Securities Fund	9	3	10	9	7.75	10
Birla Sun Life Treasury Optimizer	1	10	5	1	4 25	3
ICICI Prudential Banking & PSU Debt Fund	4	8	8	8	7.00	9
HDFC Medium Term Opportunities Fund	3	7	7	7	6.00	6
Religare Invesco Medium Term Bond Fund	2	6	9	10	6.75	7
Tata Balanced Fund	5	1	3	4	3.25	1
HDFC Childrens Gift Fund	6	2	2	5	3.75	2
SBI Magnum Balanced Fund	10	5	1	3	4.75	5
Franklin India Balanced Fund	8	4	4	2	4.50	4
L&T Gilt Fund	7	9	6	6	7.00	8

According to the Sharpe, Tryners, Jensens and Sortino ratio models it was found that Tata Balanced fund out performed every other fund. It was followed by HDFC Children's Gift fund in the second place and in the third place was Birla Sun Life Treasury Optimiser fund. In the fourth and fifth place was Franklin India Balanced fund and SBI Magnum Balanced fund respectively.

#### V. Implications of the Research

IDFC Government Securities Fund has a Alpha value of 0.9 which says that the fund is least performing to bench indices. The Tata Balanced Fund has an Alpha

value of 9.25, which shows that it has outperformed all the other funds. Tata Balanced Fund has a Beta value of 1.01, which shows that the fund is highly volatile among all other funds. Birla Sun Life Treasury Optimizer Fund has a Beta value of 0.43, which shows that the fund is less volatile in the market. Tata Balanced Fund has a standard deviation of 12.72, which means the fund has high risk factor. Birla Sun Life Treasury Optimizer Fund has a standard deviation of 1.34, which means the fund is less risky. Tata Balanced Fund has the highest return of 21.84 and IDFC has the lowest returns of 6.73. This shows that the Tata Balanced Fund has a very good performance over the period of 9 years. Based on the systematic (Beta) and unsystematic (Std. Dev), Performance indicator (Alpha) and Return, Tata Balanced Fund has outperformed the top ten funds.

#### VI. CONCLUSION

The mutual fund industry is gaining importance in the recent years. A large number of plans have come up from different financial resources. With the stock markets soaring the investors are attracted towards these schemes. Still only a small segment of the investors invest in mutual funds due to the risk associated with it. Also there is a greater tendency to invest in fixed deposits due to the security. Such investors can invest in safe funds like debt and balanced funds, with comparatively less risk and earn high returns than fixed deposits. In order to excel and make mutual funds a success, companies still need to create awareness and understand the psyche of the Indian consumer. Performance analysis helps investors as well as the fund manager to study about risk and return relationship and is a useful tool for making proper investment decisions. It acts as a guide for the investors in choosing the schemes which best suits their expected returns and risk tolerance level.

#### **References Références Referencias**

- Arnold L Redman, N.S Gullet and Herman Manakyan (2000), "The Performance of Global and International Mutual Funds" pp 79-84.
- 2. Bollen P., and Busse, J.A., (2005), "Short-term persistence in mutual fund performance", Review of Financial Studies, 18, 569-597.
- 3. Blake, D., and Timmerman, A., (1998), "Mutual fund performance: evidence from the UK", European Finance Review, 57.
- 4. Bhall V.K, (2001) "Investment Management", Sultan Chand & Co., New Delhi, Eight Edition.
- 5. Carhart, M., 1997, "On Persistence in Mutual Fund Performance", Journal of Finance, Vol.52 (1), 57-82.
- 6. Chang, E.C. and W.C. Lewellen, "An Arbitrage Pricing Approach to Evaluating Mutual Fund Performance", Journal of Financial Research, 8, Spring 1985, pp. 15-30.
- 7. David Blake (2003), "Performance Persistence in Mutual Fund", Birbeck College, pp. 18-31.
- Fama, Eugene F. (1968), "Risk, Return and Equilibrium: Some clarifying comments". Journal of Finance, (23), pp 29-40.
- 9. Fisher. E. Donald and Jordan. J. Ronald, "Security Analysis and Portfolio Management", Prentice Hall of India, Sixth edition.
- Friend, Irwin and Douglas Vickers. (1965). "Portfolio Selection and Investment Performance". Journal of Finance (20): pp 391-415.
- 11. Grinblatt, M. and S. Titman (1989), "A Comparison of Measures of Mutual Fund Performance on a

sample of Monthly Mutual Fund Returns", Journal of Business, 62, pp. 383-416.

- Hendricks, D., Patel, J., and Zechhauser, R. (1993), Hot Hands in Mutual Funds: Short-Run Persistence of Relative Performance, Journal of Finance, 48, pp 93-130.
- Horowitz, Ira. (1965). "A Model for Mutual Fund Evolution." Industrial Management Review 6: pp 81-92.
- 14. J. Cai, KC Chan and T Yamada (2002), "The Performance of Japanese Mutual Funds" pp. 10-23.
- Mechael C, Jensen, (2002) "The Performance of Mutual Funds in the period 1945 – 1964" pp.389-416.
- 16. Sharpe, W.F. (1966), "Mutual Fund Performance", Journal of Business, 39, pp. 119-38.
- 17. S.P. Kothari, Jerold B, Werner (2005) "Evaluation Mutual Fund Performance" pp. 1991-2004
- Strong. A.Robert (2001), "Portfolio Management Handbook" Jaico Publishing House, Mumbai, Second Edition.
- Timotej Jagric' Boris Podobnik, Sebastjan Strasek, Vita Jagric (2007) "Risk Adjusted Performance of Mutual Funds.



GLOBAL JOURNAL OF MANAGEMENT AND BUSINESS RESEARCH: C FINANCE Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

# Capital Structure and Financial Performance of Insurance Industries in Ethiopia

### By Mohammed Getahun

#### Jimma University

*Abstract-* An appropriate capital structure is a critical decision for any business organization to be taken by business organization for maximization of shareholders wealth and sustained growth. Thus, the major focus of this study is to investigate empirically firm specific factors such as, firm leverage, growth opportunities, size, risk, tangibility and liquidity were impacts on performance in Ethiopian insurance companies from 2004-2013 annual reports. The results show that firm leverage, Size, tangibility and business risk are significant impact on performance of insurance companies in Ethiopia. While firm growth and liquidity are not clear and statistical proved relationship are obtained from the regression analysis. The results provide strong evidence in support of the pecking order theory of capital structure which asserts that leverage is a significant determinant of firms' performance. A significant negative relationship is established between leverage and performance. From the findings the researcher recommended that the sample of insurance companies in Ethiopia use more equity than debt in financing their business activities, this because if the value of business can be enhanced with debt capital, it is dangerous for the firm.

Keywords: capital structure, performance, Ethiopian insurance industry, Returns on asset.

GJMBR - C Classification : JEL Code: G19



Strictly as per the compliance and regulations of:



© 2016. Mohammed Getahun. This is a research/review paper, distributed under the terms of the Creative Commons Attribution. Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

# Capital Structure and Financial Performance of Insurance Industries in Ethiopia

Mohammed Getahun

Abstract- An appropriate capital structure is a critical decision for any business organization to be taken by business organization for maximization of shareholders wealth and sustained growth. Thus, the major focus of this study is to investigate empirically firm specific factors such as, firm leverage, growth opportunities, size, risk, tangibility and liquidity were impacts on performance in Ethiopian insurance companies from 2004-2013 annual reports. The results show that firm leverage, Size, tangibility and business risk are significant impact on performance of insurance companies in Ethiopia. While firm growth and liquidity are not clear and statistical proved relationship are obtained from the regression analysis. The results provide strong evidence in support of the pecking order theory of capital structure which asserts that leverage is a significant determinant of firms' performance. A significant negative relationship is established between leverage and performance. From the findings the researcher recommended that the sample of insurance companies in Ethiopia use more equity than debt in financing their business activities, this because if the value of business can be enhanced with debt capital, it is dangerous for the firm. Each insurance industry establishes with the aid of professional financial managers, that particular debt-equity mix that maximizes its value and minimizes its weighted average cost of capital.

*Keywords:* capital structure, performance, Ethiopian insurance industry, Returns on asset.

#### I. INTRODUCTION

he capital structure of a firm describes the way in which a firm raised capital needed to establish and expand its business activities. It is a mixture of various types of equity and debt capital a firm maintained resulting from the firms financing decisions. In one way or another, business activity must be financed.

Without finance to support their fixed assets and working capital requirements, business could not exist. In all aspects of capital investment decision, the capital structure decision is the vital one since the profitability of an enterprise is directly affected by such decision. Therefore, proper care and attention need to be given while determining capital structure decision.

The theory of capital structure and its relationship with a firm's value and performance has been a puzzling issue in corporate finance and accounting literature since the Modigliani and Miller (1958) argue that under the perfect capital market assumption that, if there is no bankrupt cost and capital markets are frictionless, if without taxes, the firm's value is independent with the structure of the capital. Debt can reduce the tax to pay, so the best capital structure of enterprise should be one hundred percent of the debt. Since then, several theories have been developed to explain the capital of a firm including the Pecking order theory, Static Trade-off theory and agency cost theory. The firm's decision about its source of capital will affect its competitiveness among its peers. Therefore, firm should use the appropriate mix of debt and equity that will maximize its profitability.

In connection this, financing the firm's needs, the amount of debt to be undertaken is affected by several factors. Capital structure theory, specifically the trade-off model suggests that firms with high business risks should use less debt than lower risk firms. This because the higher the risk the higher probability that the firm will face financial distress. Furthermore, firms that have tangible asset should use more debt than firms that have more intangible assets since only tangible assets can be used as collateral. Besides, when financial distress occurs, intangible assets will most likely to lose value. It also stated that firms that are paying taxes at higher rates should take more debt since its bankruptcy risks is lesser than the lower taxpayer firms (Brigham, Gapenski and Ehrhardt, 1999).

Pecking order theory that has been introduced by Myers (1977) is also relevant to deviation of capital structure. It states that firms have a preferred hierarchy for financing decisions. The highest preference is to use internal financing before resorting to any form of external fund.

The Agency cost theory lastly states that an optimal capital structure is attainable by reducing the costs resulting from the conflicting between the managers and the owners. Jensen and Meckling (1976) argued that leverage level can be used to monitor the managers to pursue the overall firm's objectives and theirs. By so doing, cost is reduced leading to efficiency which shall eventually enhance firm performance (Buferna et.al, 2005).

Furthermore, capital structure and its impact on performance have been investigated for many years, but researchers have found different results with different contexts. Accordingly, there is no specific result, which can be generalizes on the extent of the relationship

Author: (MSc), Department of Accounting & Finance, college Business and Economics, Jimma University, Ethiopia. e-mail: mamegetahun@gmail.com

between capital structure and firm performance, thus there is a constant for new research in different context for achieving a more complete understanding for the dynamics of the capital structure and firm performance interchange.

The issues of capital structure are commonly, not given attention in developing countries, such as Ethiopia. The primary reason is that firms in those countries face major financing constraints, such as undeveloped bond markets and ineffective bank lending. It is important for developing countries to better understanding their financial institutions and the nature of their funding sources. The financial managers very important to know issue capital structure decision in these institutions. To them in fulfilling their goals, it is important to provide them with knowledge that relates to various determinants of financing. It would help financial managers to improve their financing decisions regarding their financing mix. By taking into account some key variables that affect their capital structure, financial managers can better achieve their overall performance goals. Therefore, the researcher attempt to clarify someof the key firm characteristics that managers need to consider when setting their optimal capital structure. Thus, the researcher goal is to understand and isolate the effects of firm characteristics on the performance ofinsurance companies in Ethiopia.

a) Objectives of the study

The general objective of this study will be to determine the effect of capital structure to the company's financial performance of some selected insurance companies institutions in Ethiopia. The Specific Objectives of the study are:

- 1. To investigate the effect of leverage on performance of insurance companies in Ethiopia
- 2. To determine growth opportunitieson performance of insurance companies in Ethiopia
- 3. To examine the effect of bank size on performance of insurance companies in Ethiopia
- 4. To determine the effect tangibility, business risk and *liquidity* on performance of insurance companies in Ethiopia

#### b) Justification of the Study

The main objective of this study was the impact capital structure on the performance of Ethiopian insurance industry. It provides the applicable and practical teaching to anyone who wishes to understand the topic. In general, this study will cover many aspects of the topic but specifically it has been tried to determine the relationship between of capital structure and performance of the firm. This study especially will help the managers to take the financing decision for their firms. The creditors can also take the benefit to minimize their risk, in funding a specific sector firms. This study will be beneficial to Ethiopian insurance company's management and investors in making clear decisions on capital structure. In addition to the above, a lot of work is written because of the endless argument on capital structure theories.

This study is another contribution to the existing work on the study of the impact of capital structure on performance of Ethiopian insurance companies.

#### II. LITERATURE REVIEW

#### a) Theories of Capital Structure

Capital structure theory, as known today, originates from the work of Modigliani and Miller, hereafter named M&M, who published their famous article in 1958. Many, if not all business and finance academics have heard and know about M&M's capital structure irrelevance proposition and several textbooks within corporate finance begin their explanations of capital structure and cost of capital with the work of M&M. In addition M&M Myers (2002) indicates that the capital structure theories and empirical evidences focus mainly on financing strategy as well as the selection of an optimal debt ratio for a certain type of firm that operates in a distinct institutional environment. According to Myers (2002), these theories are credible not because they do a perfect job highlighting the differences in total debt ratios, but because the costs and benefits that drive the theories at work in financing strategies can beobserved. While there is no universal theory of capital structure, there are however, some relevant conditional theories and these theories can be distinguished in their relative focus on the factors that could significantly impact the right mix of debt and equity. These factors comprise taxes, agency costs, and differences in information, institutional or regulatory constraints and a whole lot more (Myers, 2002). The same author stressed that each of these factors could be very significant for some firms and for other firms they could be highly unimportant. The leading theories are given below. Majority of these theories overlap and a blend of these theories help in explaining capital structure.

Capital structure theory still provides the foundation for many other theories suggested by other researches.

#### b) Trade-Off Theory

The tradeoff theory model originated from the debate over the M&M's theorem. When corporate tax was added to the original irrelevance proposition of M&M, a benefit for debt is observed that serves to shield earnings from taxes. This theory states that the optimal capital structure is the trade-off between the benefits of debt (i. e., the interest tax shields) and the costs of debt (i. e., the financial distress and agency costs) (Brigham and Houston, 2004).



Source: (Brigham & Houston 2004)

Study made by Wippern(1966) investigated relationship between financial leverage and firm performance. In his study he used debt to equity ratio as financial leverage indicator and earning to market value of common stock as performance indicator. His results indicated that leverage has positive effects on firm performance.

Capon et al. (1990) conducted a meta- analysis from 320 published studies related financial performance, and found a positive relationship between usage of leverage levels and financial performance. In 1995 Roden and Lewellen analyzed the impact of capital structure on performance for 48 US based firms with a leveraged buyout during the period 1981 through 1990, using multinomial logit models. Their results indicate a positive relationship between firm performance and its leverage policy based on tax considerations. Their findings were consistent with the trade-off theory.

#### c) Pecking Order Model

Unlike the trade-off theory, the pecking order theory does not assume an optimal level of capital structure. It states that companies prioritize their source of financing, from internal financing to equity financing, according to the principle of the least resistance, preferring to raise equity as a financing means of last resort. So, the pecking order theory claims that internal funds are used first and only when all internal finances have been depleted, firms will optimum for debt. When it is not sensible to issue any more debt, they will eventually turn to equity as a last financing resource.

Study made by kester (1986) recorded a negative link between capital structure and firm performance in the U.s and Japan. Similar results,

negative relationship between capital structure and financial performance, were reported for US firms by friend and Lang (1988) as by Titman and wessels(1988). According to the study Rajan and Zingales(1995) used data from F7 countries and recorded a negative relationship between firm leverage and firm performance.

According the studies of Fama and French also tested the pecking order and the trade-off theories on more than 3000 firms in their publication of 2002. Their study covered the period 1965 to 1999. Their models were based on both cross-section and time series methods in order to check for robustness of their results. They support the pecking order theory by documenting a negative relationship between a firm's leverage and its performance.

According to Minton and Wruck (2001) examined domestic financial conservative firms and their capital structure over the period of 1974 to 1998 and they concluded that the performance of low leverage firms outweigh the performance of high level firms. This thus indicates that there is a negative relationship between leverage and a firm's performance.

#### d) Agency Cost Theory

The next important theory mentioned in the literature is the agency cost theory. Jensen and Meckling developed this theory in their 1976 publications. This theory considered debt to be a necessary factor that creates conflict between equity holders and managers. Both scholars used this theory to argue that the probability distribution of cash flows provided by the firm is not independent of its ownership structure and that this fact may be used to explain optimal capital structure. Jensen and Meckling recommended that, given increasing agency costs with both the equity-holders and debt-holders, there would be an optimum combination of outside debt and equity to reduce total agency costs.

#### e) Capital structure determinants and performance

#### i. Firms Leverage

The pecking theory of capital structure shows that if a firm is profitable, then it is more likely that financing would be from internal sources rather than external sources. In other words, firms tend to use internally generated funds first and then resort to external financing. This implies that profitable firms will have less amount of leverage (Myers and Majluf, 1984). By this, profitable firms that have access to retained profits can rely on them as opposed to depending on outside sources (debt).

Most studies found a negative relationship between profitability and capital structure Friend and Lang, (1988); Barton et al., (1989); Van der Wijst and Thurik, (1993); Chittendenet al., 1996; Jordan et al., (1998); Shyam-Sunder and Myers, (1999); Mishra and McConaughy, (1999);. Cassar and Holmes (2003), and Hall et al. (2004) also suggest negative relationships between profitability and both long-term debt and shortterm debt ratios.

#### ii. Growth opportunities

According to Brush, Bromiley, &Hendricks, (2000) in the light of free cash flow hypothesis, they conducted in Maryland-USA found a strong positive relationship between sales growth and a firm's financial performance in terms of stockholders' returns and return on assets.

Additionally, for the top 500 Australian companies. In addition of this Hutchinson and Gul, (2006) they found that firms with high investment opportunities are associated with lower agency costs and better return on equity.

#### iii. Firm's size

According to the studies (Orser, Hogarth-Scott, & Riding 2000), using Canadian firms using changes in gross revenue to reflect performance. They find a positive effect for a firm's size support the arguments that size reflects greater diversification, economies of scale production, greater access to new technology and cheaper sources of funds.

#### f) Asset structure (tangibility)

#### i. Firm's liquidity

According to the researcher knowledge apart from (Wang, 2002) there is no studies address this relationship. But, (Wang, 2002) and, who addresses the liquidity management. He investigates the liquidity management and its relationship with performance and corporate value using data of Taiwan and Japan. Furthermore, he observes that the cash conversion cycle (CCC) has a negative relationship with the financial performance measured by returns on assets (ROA) or returns on equity (ROE) and this relationship is sensitive to industry factors. Furthermore, he finds that aggressive liquidity management enhances operating performance.

#### ii. Firm's business risk

Many studies investigate the relationship between risk and profitability. Among others (Shergill & Sarkaria 1999) using the data of Indian firms, they confirm the positive relationship between a firm's risk and financial Performance,(Dewan, Shi, &Gurbaxani 2007) using the Fortune 1000 and the total firm value to reflect performance,( Loudon 2006) for 15 markets, comprising a mix of developed and emerging markets using equity returns.

#### iii. Conceptual Frame Work

After careful study of literature review, the following conceptual model is formulated to illustrate the effects of capital structure on performance. The conceptualization model from figure below shows the effects of capital structure on profitability of Ethiopian insurance companies.

#### III. METHODOLOGY

This section stresses the methodology employed for this work. The process of research usually entails problem identification, making hypothetical statements, collecting relevant data, analyzing the data using the relevant and appropriate statistical tools of analysis.

This paper is based on secondary data collection. The sources of data for this study are Balance sheets and Income Statements of companies over 10 years period from 2004 still 2013, which are mainly extracted from National Bank of Ethiopia, which can provide comprehensive database for all insurance companies. Time series and cross sectional data has been used in this study where 9 commercial banks out of 17 insurance companies have been included in the study in Ethiopia. However, the remaining insurance companies did not have the required period information. Due to this reason, the year service below Ten years is not included in sample frame to make panel data model structured.

#### a) Model Specification

The multiple regression models used to establish the relationship between capital structure and financial performance was of the specific form;

## $$\begin{split} \text{ROA} &= \beta 0 - \beta 1 \text{LEVit} + \beta 2 \text{ GRit} + \beta 3 \text{ SIZE it} + \beta 5 \\ \text{TANGit} + \beta 6 \text{LQit} + \beta 4 \text{Brit} + \text{eit.} \end{split}$$

Where:

ROA - Return on Asset (performance of the firm) β0 - Constant coefficient  $\beta 1 - \beta 6$  = Regression coefficients for measuring independent variables LV = Firm Leverage GR = growth opportunities Size = firm size Tang = tangibility of fixed asset

LQ = liquidity of the firm

Br = business risk.

 $\label{eq:Uit} \mbox{Uit} = \mbox{Error component showing unobserved factor}$ 

#### b) Operationalization of Variables

The description of each variable and their expected signs are given below in the following tables.

Variables	Variable Measurements	Some References	Expected Signs
Firm's leverage	Total liabilities Total assets	Kyereboah-Coleman (2007), Abor (2005), Titman and Wessels (1988), king and santor, 2008	(-)
Growth opportunities	Change in the total assets	(Degryse, Goeij, &Kappert, 2010), Hovakimian, Opler and Titman, 2001	(+)
Business risk	Standard deviation of operating income/Total Asset	Wald (1999),Kim & Limpaphayom,(1998) and Allen & Mizunot, (1989)	( +)
Size	Natural logarithm of total assets	Holmes, 2003; Panno, 2003; Deesomsak 2004; King and Santor (2008)).	(+)
Asset tangibility	Total fixed assets Total assets	Titman &Wessels 1988; Gaud <i>et al.</i> ,2005, Fattouh, Scaramozzino, & Harris 2005	( +)
Liquidity	Current assets Current liabilities	Kila and Mansoor (2009), Ozkan 2001, Laitinen 200	(+)
μit	Are the error terms		
Dependent variable			
Return on Asset	Profit after tax Total assets	(Bistrova, Lace, &Peleckienė, 2011 Mehran (1995), Ang, Cole and Line (2000	

#### IV. Results and Discussion

#### a) Descriptive statistics

Variable	Obs	Mean	Std. Dev.	Min	Max
ROA	90	.0783043	.123769	10886	.921629
Lev	90	.520138	.1843834	.02007	.902047
Grow.Opp	90	.352805	1.418099	9800652	1 3.16158
Size	90	18.95876	1.090104	16.30014	21.22304
Та	90	.1410642	.0998923	.000258	.465749
Lq	90	2.633622	1.829073	.103773	11.24678
Br	90	.1602669	.183787	.019253	1.48693

#### Source: computed from financial statement of Ethiopian insurance companies

As presented in table above, the average value of the performance ratios measured by ROA, sample Ethiopian insurance industry is 7.8 percent (0.0783043), this implies sample Ethiopian insurance companies on average earned a net income of 7.8 percent of total asset with a maximum and minimum value of 0.921629 and -.10886. The standard deviation is 12.4 percent from the average value, which reflects the presence of moderate variation among across the sampled insurance companies.

#### b) Regression result

Regression analysis is a statistical technique used to test the relationship between one dependent variable and one or several independent (predictor) variables.

ROA	Coef.	Std. Err.	t	P> t	[95% Conf. Inte	erval]
Lev	1673747	.0758574	-2.21	0.030*	3184903	016259
Grow.Opp	.0038993	.0082162	0.47	0.636	0124683	.0202669
Size	.0886285	.0211508	4 .19	0.000**	.0464938	.1307631
Та	3100963	.1231772	-2.52	0.014*	5554778	0647148
Lq	0160876	.0126096	-1.28	0.206	0412072	.0090321
Br	.3995292	.0728456	5 .48	0.000**	.2544132	.5446451
_cons	-1.494222	.4170297	-3.58	0.001**	-2.324988	6634563
Sigma u	.09529538					
Sigma e	.0989344					
rho	48127097					
R2	0.3720					
No.obs	90					

Regression Result: Fixed effect regression model

Note \* Significant at 1% level, \*\* significant at 5% level

 $ROA = \beta 0 - \beta 1 LEVit + \beta 2 GRit + \beta 3 SIZEit + \beta 4Brit + \beta 5 TANGit + \beta 6LQit + eit.$ 

ROA=-1.494-0.167LVit + 0.0039Grit + 0.089SIZE it - 0.31TANGit - 0.016LQit + 0.40Briteit.

(0.759) (0.0082) (0.211) (0.073) (0.0123) (0.126)

R2 from the table 4.8, 37.2% variations in the dependent variable can be accounted for by the independent variables. This means 37.2% of variations in the performance of selected Ethiopian insurance companies are explained by independent variable. This showed that the independent variable values have at least 37% significant influence on performance of the Ethiopian insurance companies. This also indicates that there are other variables that influence the variations in the level of performance of the firms.

#### c) Firm leverage

As presented in table above, panel data results for the analysis method of fixed effects model results show a negative and significant impact on profitability of Ethiopian insurance industry with a regression coefficient of -0.1673747, t-statistic -2.21, P-value of 0.030.

Theoretical prediction yields no conclusion for the relationship between leverage and performance. Trade off models argues that profitable firms have great needs to shield income from corporate tax and should borrow more than less than profitable firms. While pecking order models theory suggests an inverse relationship between leverage and profitability of the firm.

This results has been consistent with Jensen (1986) that if firm leverage acts as a bonding device in terms of forcing managers to commit free cash flows to service debt, then higher debt will lead to lower funds available for managers in profitable investments and then lower performance (Singh & Faircloth 2005).

d) Growth opportunities

The panel fixed effect estimation regression result shows insignificant a positive relationship between growths of sampled Ethiopian insurance companies and their performance ratio with a regression coefficient of 0.0038993, t-statistic of 0.47, p-value of 0.636.

Trade-off theory considers growth opportunities as an indicator for the firm success; these firms are stronger to face financial distress. Firms with good opportunities have a good reputation in getting funds, easier access to the finance markets and reflected in better performance for these firms. According to the agency theory perspective, firms with high growth opportunities have lower agency costs. These firms might have lower debt ratios due to the fear of debt holders those firms may forgo valuable investment opportunities and expropriate wealth to their benefit, and this outcome would be reflected in lower agency costs (Hutchinson & Gul 2006).

#### e) Firm size

The panel fixed effect estimation result reveals there is significant positive relationship between size and performance of sampled Ethiopian insurance companies with a regression coefficient of .0886285, tstatistic of4.19, and P-value of 0.000. The significance of firm size on firm performance indicates that large firms canearn higher returns compared to smaller firms, most probably as a result of diversification of investment and economies of scale.

#### f) Asset tangibility

The panel fixed effect estimation result, in this study, shows a statistical significant negative relationship between tangibility of assets and

performance of Ethiopian insurance companies with a regression coefficient of -.3100963, t-statistic-2.52 and p-value of 0.014. This means that a sampled Ethiopian insurance company with high ratio of fixed assets to total asset leads lower performance of the companies, because in Ethiopia lending financial institutions not require fixed assets as collateral to provide debt to those of insurance companies. The other reason is the fixed asset of Ethiopian insurance companies not able to generate revenue. This shows that firms with high ratio of tangibility have a lower performance ratio. However, the negative relationship between firm's asset tangibility and performance is consistent with similar findings of previous researchers Osuji & Odita, A (2012).

#### g) Firm Liquidity

A result from fixed effects models shows a negative and insignificant relationship between firm liquidity and performance of Ethiopian insurance industries. Specifically, fixed effect estimation with a coefficient of -0.0160876, t- statistic -1.28 and p-value of 0.206 confirmed a negative relationship between liquidity and performance ratio.

Pecking order theory suggesting that the more liquid firm would use external financing due to their ability of paying back liabilities while trade of theory suggesting that high liquidity position for the firm indicates that this firm is strong enough to face any short or long-term financial problems and this strong firm can perform better than a weak firm which has weak liquidity position in its financial statements.

#### h) Business risk

Result shows in this study, shows a statistical significant positive relationship between business risk and performance ratio with a regression coefficient 0.3995292, t-statistic-3.58 and p-value of 0.001, which statistical significant positive on performance of Ethiopian insurance companies.

The reason for such relationship due the theoretical prediction of the agency theory; the required rate return from investors should be suitable to their risk in the firm. Shareholders will require high return in order to hold the risk related to the bankruptcy and financial distress since the debt holders have the priority in the case of bankruptcy.

#### V. Limitation of the Study

- The study consists of only ten years of data that might not be sufficient to establish the relation in a very significant manner.
- The study considers only secondary data but not primary (i.e., interaction with the executives in finance department would close picture and management style etc. is not considered).

#### VI. CONCLUSION AND RECOMMENDATIONS

#### a) Conclusion

Capital structure has been a much debated topic in the finance field since the Modigliani& Miller proposition in 1958. Capital structure theories, such as the pecking order and the tradeoff theory emerged into the finance field and many have tried to analyze the implications of these theories for firms in the market.

The objective of this study is limited to the impact of capital structure on the performance in the context of Ethiopian insurance industries. This paper has applied the panel data regressions for nine insurance companies in Ethiopia during the period 2004 to 2013. All insurance companies included in the study if they had the specified period of time, audited financial statements of ten years. This thesis examined empirically the implication of theory of capital structure in Ethiopian insurance companies. The results of regression analysis disclose that firm leverage, growth opportunities, size, business risk, tangibility of assets and liquidity as independent variable while the profitability the firm (ROA) is dependent variable. The results show that firm leverage, Size, tangibility and business risk are significant effects on performance of Ethiopian insurance companies.

b) Recommendation

The result proves that with the increase in leverage negatively affects the performance Ethiopian insurance industry. Therefore, the researcher recommends that managers shall not use excessive amount of leverage in their capital structure, they must try to finance their projects with retained earnings and use leverage as a last option. Managers must work to achieve the optimal capital structure level to maximize the firm's performance and try to maintain it as much as possible.

In generally, the variable that significant direct relationship between the impacts of capital structure on performance of the firm, the managers should devote their time and efforts on those variables in order to minimize the weighted average cost of capital and consequently maximize the welfare of shareholders.

#### c) Recommendations for Further Research

The study has laid some ground work to explore the effects of capital structure on performance of Ethiopian insurance industries. Further work is required to develop new hypotheses and design new variables to reflect the firm specific factors to influence on firm performance related with theory of capital structure.

#### References Références Referencias

1. Brigham, E. F. & Ehrhardt, M. C. 1999, "Financial management: theory and practice", 11th ed., International student ed. edn, South-Western M, ason, Ohio.

- 2. Brigham, E. F., Foster, E., & Houston, J. F. 2004, "Fundamental of Financial Management",10th ed., edn
- Brush, T. H., Bromiley, P., & Hendrickx, M. 2000, "The free cash flow hypothesis for sales growth and firm performance", Strategic Management Journal, vol. 21, no. 4, p. 455.
- 4. Buferna, F., Bangassa, K. & Hodgkinson, L. (2005). Determinants of Capital Structure: Evidence from Libya. Research Paper Series.
- 5. Capon, N., Farley, J., & Hoenig, S. (1990). Determinants of financial performance: a metaanalysis. Management Science.
- Cassar, G. & Holmes, S. 2003, "Capital structure and financing of SMEs: Australian evidence", Accounting & Finance, vol. 43, no. 2, pp. 123-147.
- Chittenden F, Hall G. and Hutchinson P. (1996). Small Firm Growth, Access to Capital Markets and Financial Structure: Review of Issues and an Empirical Investigation. Small Business Economics 8, 59-67.
- 8. Fama, E., & Fench, K. (2002). Testing trade-off theory and pecking order predictions about dividends and debt. Review of Financial Studies.
- Hall, G., Hutchinson, P., and Michaelas, N. (2004). Determinants of the Capital Structures of European SMEs, Journal of Business Finance & Accounting 31, 711-728.
- Hutchinson, M. & Gul, A. 2006, "The effects of executive share options and investment opportunities on firms accounting performance: Some Australian evidence", British Accounting Review, vol. 38, no. 3, pp. 277-297.
- 11. Jensen, M. & Meckling, W., 1976. Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure. Journal of Financial Economics, 3, p. 305-60.
- Kester, C. W. 1986. Capital and ownership structure: A comparison of United States and Japanese manufacturing corporations. Financial Management 15(1): 5-16.
- 13. Loudon, G. F. 2006, "Is the risk return relation positive? Further evidence from a stochastic

volatility in mean approach", Applied Financial Economics, vol. 16, no. 13, pp.

- 14. Minton, B., & Wruck, K. (2001). Financial Conservatism: Evidence on Capital Structure from low leverage firms.
- Modigliani, F., and Miller, M. H. (1958,), —The Cost of Capital, Corporate Finance and the Theory of Investmentl, American Economic Review, 48, 261-97.
- Myers, S. 1977. Determinants of Corporate Borrowing, Journal of Financial Economics, 5: 147-176.
- Orser, B. J., Hogarth-Scott, S., & Riding, A. L. 2000, "Performance, Firm Size, and Management Problem Solving", Journal of Small Business Management, vol. 38, no. 4, pp. 42-58.
- Osuji & Odita Anthony (2012), Arabian Journal of Business and Management Review, Vol.1, No.12; July 2012.
- Rajan, R. and Zingales. L. (1995). what do we know about capital structure? Some evidence from international data. Journal of Finance. 50. 1421-1460.
- 20. Shergill, J. & Sarkaria, M. 1999," Impact of Industry Type and Firm Characteristics on Firm-Financial Performance-- Evidence from Indian Industry level"", Journal of Entrepreneurship, vol. 8, no. 1.
- Shyam-Sanders, L. and Myers, S.C. (1999). ITesting Static trade-off against pecking order Models of Capital structure", Journal of Financial Economics, (51): pp.1421-1460.
- 22. Van der Wijst N. and Thurik R. (1993). Determinants of Small Firm Debt Ratios: An analysis of Retail Panel Data, Small Business Economics 5, 55-65.
- 23. Wald, J. (1999). How firm characteristics affect capital structure: an international comparison. Journal of Financial Research, 22, 161-87.
- 24. Wing, C. C. K. & Yiu, M. F. K. 1997, "Firm size and performance of manufacturing enterprises P.R. China: The case of Shanghai's.", Small Businesses conomics, vol. 9, no, 3, p. 287.
- 25. Wippern, R. (1996). Financial Structure and the Value of the Firm. The Journal of Finance.

#### Appendix

Appendix for Hausman test

			Coeffic	ients	
	(b)	(B)		(b-B) sqrt(d	diag (V b-V B))
		fixed	random	Difference	S.E.
-	lv	1673747	0884367	078938	.0334373
	gr	.0038993	.002613	.0012863	.0008566
	size	.0886285	.0464443	.0421841	.0157781
	ta	3100963	1414094	1686869	.0288326
	lq	0160876	0054207	0106668	.0094598
	br	.3995292	.3139315	.0855977	.0238584

b = consistent under Ho and Ha; obtained from xtreg
 B = inconsistent under Ha, efficient under Ho; obtained from xtreg
 Test: Ho: difference in coefficients not systematic

chi2 (6) = (b-B)'[(V\_b-V\_B)^(-1)](b-B) = 62.20Prob>chi2 = 0.0000 (V b-V B is not positive definite)

Appendix for fixed effects regression analysis

Fixed-e	effects (v	vithin) regres	Numbe	er of Obs =	90		
Group	variable	: company	Number of groups $=$ 9				
R-sq:	within =	= 0.3720		Obs per	group: min =	10	
E	Between	= 0.3310			avg = 10	.0	
	Overall	= 0.2383			max = 10	0	
		Date:	08 /05/1	4 Time: 9:28	5		
C C	Sample2	004 to 2010		F (6, 75)	= 7.4	40	
corr(	u_i, Xb)	= -0.7622		Prob >	F = 0	.0000	
 roa	Coef.	Std. Err.	t	P >  t	[95% Con	if. Interval]	
IV  16	673747	.0758574	-2.21	0.030**	3184903	016259	
gr   .003	8993	.0082162	0.47	0.636	0124683	.0202669	
Size   .08	86285	.0211508	4.19	0.000*	.0464938	.1307631	
ta  31	00963	.1231772	-2.52	0.014*	5554778	0647148	
lq  016	60876	.0126096	-1.28	0.206	0412072	.0090321	
br   .39	95292	.0728456	5.48	0.000**	.2544132	.5446451	
_cons   -1.	494222	.4170297	-3.58	0.001**	-2.324988	6634563	

sigma\_u | .09529538 sigma\_e | .0989344 rho | .48127097 (fraction of variance due to u\_i)

# This page is intentionally left blank



GLOBAL JOURNAL OF MANAGEMENT AND BUSINESS RESEARCH: C FINANCE Volume 16 Issue 7 Version 1.0 Year 2016 Type: Double Blind Peer Reviewed International Research Journal Publisher: Global Journals Inc. (USA) Online ISSN: 2249-4588 & Print ISSN: 0975-5853

### Factors Affecting Loan Repayment Performance of Small Scale Enterprises Financed by Micro Finance Institutions: Study on Private Borrowers around Wolaita and Dawuro Zone

By Mesele Kebede, Tesfahun Tegegn & Tadele Tafese

Wolaita Sodo University

*Abstract*- Micro finance involves the provision of micro-credit, savings, and other services to the poor that are excluded by the commercial banks for collateral and other reasons. Microfinance is relatively new to Ethiopia and came to existence during 1994-95. Out of which Wolaita zone Micro Finance Institution (WZMFI) is one among 31 Micro Finance Institutions (MFIs) to serve needy people in Ethiopia. Based on this researchers' intended to study major socio- economic factors and loan related factors that determines loan repayment performance of borrowers in WZMFI. In fact, the identifying and analyzing such determining factors of loan repayment rate is vital in the achievement of profitability and sustainability of MFIs. In this connection, researchers' collected data from primary and secondary resources and analyzed by using two limit Tobit model is used. Through the study 15 determinants' are selected for evaluation, out of which 6 variables are significant and remaining insignificant are found.

Keywords: loan repayment, performance, small scale enterprise, micro finance institution, private borrowers.

GJMBR - C Classification : JEL Code: G20



Strictly as per the compliance and regulations of:



© 2016. Mesele Kebede, Tesfahun Tegegn & Tadele Tafese. This is a research/review paper, distributed under the terms of the Creative Commons Attribution-Noncommercial 3.0 Unported License http://creativecommons.org/licenses/by-nc/3.0/), permitting all non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

# Factors Affecting Loan Repayment Performance of Small Scale Enterprises Financed by Micro Finance Institutions: Study on Private Borrowers around Wolaita and Dawuro Zone

Mesele Kebede  $^{\alpha},$  Tesfahun Tegegn  $^{\sigma}$  & Tadele Tafese  $^{\rho}$ 

Abstract- Micro finance involves the provision of micro-credit, savings, and other services to the poor that are excluded by the commercial banks for collateral and other reasons. Microfinance is relatively new to Ethiopia and came to existence during 1994-95. Out of which Wolaita zone Micro Finance Institution (WZMFI) is one among 31 Micro Finance Institutions (MFIs) to serve needy people in Ethiopia. Based on this researchers' intended to study major socio- economic factors and loan related factors that determines loan repayment performance of borrowers in WZMFI. In fact, the identifying and analyzing such determining factors of loan repayment rate is vital in the achievement of profitability and sustainability of MFIs. In this connection, researchers' collected data from primary and secondary resources and analyzed by using two limit Tobit model is used. Through the study 15 determinants' are selected for evaluation, out of which 6 variables are significant and remaining insignificant are found. Based on the analysis, researchers are recommended that proper training should be provided. reasonable amount of loan which should be useful to their business. Further, more age people and well business experience people can able to repay their loan amount' timely to micro finance institution.

*Keywords: loan repayment, performance, small scale enterprise, micro finance institution, private borrowers.* 

#### I. INTRODUCTION AND JUSTIFICATION

n developing countries like Ethiopia where the farming system is at its traditional level and the industrial and service sectors are at their infant stage, the role of small scale enterprises (SSEs) is significant in terms of their employment generation capacity, quick production response, adaptation to weak infrastructure, use of local resources and as a means of developing indigenous entrepreneurial and managerial skills for a sustained growth need (Aryeetey, 2004 in BFasika and Daniel, 2007). For small-scale enterprises to grow up to medium and large-scale level, the need for formal credit source is indispensable because formal financial sector have the financial capacity to meet their growing credit demand, which the informal sector is incapable to supply.

Despite their importance, many of them do not have sufficient access to credit from formal financial institutes. Their major source of finance, especially at the start up stage, is the informal sector (i.e. from friends, relatives and local money lenders). This poor credit access from formal financial source, based on the experience of some developing countries, arises partly from biased government policy, due to the operational practices and procedures of the formal financial institutions and the internal problems of small scale enterprises themselves. (Asrat, 2009). Solving the major financial constraint of this important sub-sector of the economy is an important step towards achieving the national development objective of a country. For this to succeed, the problem of high default risk associated with them, which made the financial institutes reluctant to extend loan, has to be solved. However, the majority of potentially viable SSEs still couldn't get credit access from the formal financial market. High transaction cost, complex bureaucratic lending procedures, elaborate paper work, high collateral requirements and delays are some of the factors which militate against effective utilization of the existing banking facilities (Dejene, 2003). Because of this only limited number of SSEs could be eligible for credit from the banking sector.

Similarly, MFI was one of government owned financial institute that passed through the lending policies mentioned earlier. Its major task has been extending medium and long term credit small and medium-scale development projects.

After 1991 like other financial institutes, MFI diverted its attention towards the private sector whose share never exceeded 11% during the socialist period increased to more than 77% in 2011/12 (MFI, 1970/71-2011/12). Credit access to small scale private enterprises was also improved although it didn't match with the need of customers. With the removal of restrictions imposed by the government, the bank has been given autonomy to pass its own lending decision on the basis of purely commercial criteria. Together with this there is no government guarantee unlike before in

Author α: Department of Accounting and Finance, Wolaita Sodo University, Ethiopia. e-mail: mesiyek2002@yahoo.com

Author σ: Department of Management, Wolaita Sodo University, Ethiopia. e-mail: tesfahuntegegn@gmail.com

Author p: Department of Economics, Wolaita Sodo University, Ethiopia. e-mail: tadeleth@yahoo.com

case of default. That is, the bank is required to meet its development objective keeping at the same time its financial position safe. Its success/failure of development financing as well as its financial position therefore relies on its loan recovery performance.

The problem of loan default reduces the lending capacity of a financial institution. It also denies new applicants access to credit as the bank's cash flow management problems augment in direct proportion to the increasing default problem. In other words, it may disturb the normal inflow and outflow of fund a financial institution has to keep staying in sustainable credit market.

The effect of default problem experienced in MFI as mentioned earlier has been reflected on its financial position. For instance, as of June 30, 2012 the MFI's debt equity ratio was 6:1 as opposed to the internationally recognized ratio of 4:1. During the same period its current ratio (i.e. the ratio of current assets to current liabilities) stood at 0.59:1 implying that the MFI is in severe liquidity constraint, that is its current asset is not in a position to cover its current liability. The repayment problem could arise either from the demand side, supply side, and both or other external factors.

The supply side problems include change in the structure of the bank, change in the lending policy, failure in properly appraising the project document (i.e. in assessing the background of the promoter, technical capability, marketability, financial and economic viability of the project) and lack of responsibility and accountability of the staff members of the MFI.

Concerning MFI there has been no significant change introduced on the general lending policy of the MFI except shifting its attention towards loan collection than loan disbursement, which in fact arisen from severe liquidity problem it has faced. Therefore the problem on the supply side relies more on implementation of the rules and regulations of the MFI and on the MFI's efficiency of making proper credit assessment.

Studies conducted so far were on Micro enterprises (Mengistu, 2007; Birhanu, 2009; Teferri, 2000) and on manufacturing firms' case (relatively medium and large scale ones) located in Addis Ababa (Mengistu, 2009). However, these studies don't specifically touch the case of small-scale private enterprises. This study therefore tried to narrow the research gap paying attention to this sector of the economy. Studies done on micro enterprises are meant to evaluate the institutional sustainability of the credit scheme. However, this study will focuses on identifying factors behind the poor loan repayment performance that SSEs are associated with.

#### II. Objectives of the Study

The general objective of the study is to analyze and identify the major factors that determine loan repayment performance of the small scale enterprises and to identify the major challenges of the MFI's in the wolaita and Dawuro area.

- a) Specific objectives
- To identify the major socio-economic factors that influence loan repayment rate of the borrowers of micro finance institution.
- To examine the businesses and loan related factors influence the repayment performance of the Private borrowers.
- To investigate the major problems faced by the borrowers and lenders in the repayment process in micro finance institution.

#### III. METHODOLOGY

#### a) Research Design

The study employed explanatory research design with quantitative and qualitative methods. The quantitative aspect of the data focused on description of socioeconomic variables, loan and related variables, and business related variables and analysis of relationship among the dependent and explanatory variables of WZMFI for the study.

#### b) Data Sources

The study employed both primary and secondary sources. Primary data sources are the sample loan borrowers of both defaulters and non defaulters from each branch. In support of primary sources, secondary data sources were obtained from both head office, and branches' managers concerned other officers and unpublished works also.

#### c) Sampling Techniques

For this study multi-stage probability sampling techniques were used. At the first stage, the WOLAITA ZONE micro finance institution was selected purposively due to so far there is no scientific studies has been taken regarding to loan repayment performance in this study area. So, researchers motivated to identify and analyze the determinants of loan repayment performances of borrowers in WZMFI. A stratified sampling technique would used to select the respondents. At the outset, the respondents were stratified into two categories, i.e. defaulters and nondefaulters. All borrowers of the MFI's credit that would have repaid their loans when the due date are classified as non-defaulters while those who did not repay their loan three months after the due date are classified as defaulters.

Stratified sampling would be used where borrowers can be divided in to two different strata's: defaulters and non-defaulters. In each stratum, simple random sampling will be used to get the required number of respondents. According to 2006 E.C record of both Wolaita and Dawuro zone finance and economy development office, there are a total of 108,672 private borrowers were listed on the both Wolaita and Dawuro area chart of loan account out of which the repayment date for 300 borrowers was not mature, and hence are excluded from the list. sampling respondents were taken from all active clients that have been participated in the program during 2013.

About 300 Sample respondents were selected through using simple random sampling technique. The

#### IV. Result and Discussion

Table 1 : Borrowers characteristics (continuous variables)

Variables	Non -Defaulters				Defaulters				Total sample (N= 261)			
	Mean	St.deviation	minimum	maximum	Mean	St.deviation	minimum	maximum	Mean	St.deviation	minimum	maximum
Age	34.650	8.880	19	60	29.172	8.126	16	56	31.91	8.503	35	116
_									1			
Farming	1.384	5.0266	0	30	.948	5.576	0	40	1.166	5.301	0	70
Expr												
hhsize	6.1970	1.985	1	13	5.655	1.606	2	10	5.256	1.756	3	23
depratio	1.850	1.414	0	8	1.582	1.079	0	5	3.432	1.247	0	13

Sources: survey results, 2014

The average age of the whole respondents was  $32 \pm$  years, ranging between 19 and 60 years old. There was statistically significant (at 5% level) difference between the mean age of defaulter and non-defaulter (Table 1). As we see from the table below, more than half of the respondents were in the first and second age category, showing that most of the borrowers were young age groups. The proportion of youngsters in the defaulter group was a little bit higher than that in the non-defaulter group.

Thus, this indicates that the borrowers at younger stages become more defaulter than at older age. This is because as age of borrowers' increases they became settled and accumulate wealth; acquire experience in business management and credit use than youngsters. Then these and related positive variables enables elder borrowers to be better payers than youngsters.

Therefore, based on the survey result the average family size of non-defaulters is greater than the average family size of defaulters. This indicates that as family size in the household of borrowers' increases then they allocate their business incomes, which was financed by credit loan, to cover different household's expenses. As a result, this impacts the borrowers' loan repayment performance negatively. The significance value is .000, which is less than .05; therefore based on this researchers can say that there is a significant difference between defaulters at 5% significance level (table 1).

Variables	Ν	lon -Defau	lters (N=)			Defaulte	Total sample ( $N = 261$ )			
	Mean	Standard	minimum	maximum	Mean	Standard	minimum	maximum	Mean	Standard
		deviation				deviation				deviation
Total live stock ownership	4.10699	38.19087	0	544.44	.8186207	2.3631	0	9.4	2.4628	20.276985
Value of equipment	12247.41	54085.63	0	759,550	2818.1	5460.217	0	22,800	7532.755	29772.9235
Land size	4.623498	2.557656	0.99	16	4.396897	3.158024	1	20	4.5101975	2.85784
Off farm income	1289.468	4389.592	0	48,900	267.7586	983.0369	0	5000	778.6133	2686.31445

Table 2 : Borrowers characteristics (continuous variables)

Sources: Survey results, 2014, Significant at 5% level of confidence

Next to land, livestock is the most important asset for rural households in Ethiopia. It is used as a source of food, draft power, income and energy. Moreover, livestock is an index of wealth and prestige in rural community. All the sample households reared livestock, which consisted of cattle, small ruminants, back animals and poultries.

Total livestock ownership (LIVSTKNO) is, as expected, positively related to the dependent Variable (significant at 5% level). Each additional TLO increases the probability being non defaulter by the mean value of 4.10695. The minimum number of livestock maintained was 0 whereas the maximum was 544.44 Credit users possessed relatively more livestock unit than non-users. The mean difference between the two groups in owning of livestock was significant at 1% level.

The implication is that, Livestock are sources of cash in rural Ethiopia and serve as security against crop failure. Farmers who owned more livestock are able to repay their loans even when their crops fail due to natural disaster. In addition, as a proxy to oxen ownership the result suggests that farmers who have larger number of livestock have sufficient number of oxen to plough their field timely and as a result obtain high yield and income to repay loans.

Getting income from off-farm activities (OFF-FARM) is another economic factor that was positively and significantly affected loan repayment performance of smallholder farmers. This might be due to the fact that; off-farm activities were additional sources of income for smallholders and the cash generated from these activities could back up the farmers' income to settle their debt even during bad harvesting seasons and when repayment period coincides with low agricultural prices. Each additional unit of Off-farm income increases probability of being non-defaulter by the mean value of 1289.468 and on average increases the rate of loan repayment by 0.1061 for the entire respondents and by 0.131 among non defaulters. However, this result is contrary to Bekele's (2001), findings that, off-farm income was negatively related with loan repayment performance of farmers.

With respect to land size, the average land holding of the sample borrowers was 5.57 hectare. The minimum and maximum holding sizes were 0.99 and 20 hectares, respectively. All respondents owned more than 15 hectares of land. This shows that farming in the area is of subsistent type (Table 2). The average farm sizes of the non-defaulters and defaulters were 4.6 and 4.3 hectares, respectively.

Table 3 : Borrowers characteristics (continuous variables)

Variables	Non -Defaulters (N=)				Defaulters(N=)				Total sample (N= 261)			
	Mean Mean	St.deviation	minimum	maximum	Mean	Standard deviation	minimum	maximum	Mean	Standard deviation	minimum	maximum
Loan amount	22545.32	29583.82	0	544.44	7412.069	14017.07	1500	100,000	26251.3545	30,291.855	1500	100544.44
Loan frequency	2.221675	1.474242	1	8	1.12069	.65098	1	5	1.6711825	1.062611	2	13
Days pro	84.2957	75.1213	10	365	8.275862	34.08647	0	180	46.285781	54.603885	10	545
Portion of loan repaid	.6157656	.3028648	.0312891	1	0	0	0	0.3078828	.3078828	.1514324	.0312891	1.3078828

#### Source: Survey results, 2014

Despite the fact that majority of respondents have responded the loan amount approved as insufficient to their planned or current engagement, the mean amount of approval deviation is birr 29583.064 and 14,017 for good non defaulter borrowers and defaulters respectively. This implies that majority of those who have defaulted were granted a loan much lower than their request in relation to those of non defaulter borrowers.

Most borrowers request below sufficient amount and are granted even below their request. This condition leads to lower amount of investment on business, unable to hold all the necessary stocks demanded by the market and minimal return from business activity. As noted on table 3 this was the main reason cited by borrowers for lower return.

However, due to the factors indicated in the beginning, the amount applied in the first place is influenced by the credit officers' advice of what amount would possibly be approved with the given status of the borrower, irrespective of his/her demand.

With respect to loan frequency, on an average, respondents obtained the loan credits for 1.671 rounds with the standard deviation of 1.062. It was found that non-defaulters had credit 2.22 rounds while defaulters had 1.120 rounds with a standard deviation of 1.474 and 0.6509 respectively. Moreover, the mean difference between defaulters and non defaulters was statistically significant at 5 % level (table 3). This implies that if client borrow loan for the number of rounds, then they aware obligation and responsibility on loan usage as well as

repayment more than those who are the first time borrowers.

With respect to portion of loan repaid, according to loan repayment status of respondents, on an average it was found that 0.6157 of non-defaulters had fully repaid on maturity time, and only .3028 of them paid it fully but too late. The fully repayment of loan enables non-defaulters to gate the advantages of next higher loan like, having good relationship with the lending institution, keeping their socially status in the society, realizing their freedom from any penalty. Whereas, according to defaulters group, on an average, none of respondents had paid loan partially on maturity period and none of the same group had paid too late.

Marital status		Defaulters		Non d	Non defaulters		ample 261)	X <sup>2</sup> -Value
		Ν	Р	Ν	Р	Ν	Р	X <sup>2</sup> 11 0007
	Single	40	15.33	89	34.10	129	49.43	n = 11.3907
	Married	18	6.9	114	43.68	132	50.7	P=0.001***
Sex	Male	32	12.26	143	54.79	175	67.05	x <sup>2</sup> =4.7617
	Female	26	9.96	60	22.99	86	32.95	P=0.029**
educational	Illiterate	6	2.3	13	4.98	19	7.28	X <sup>2</sup> 4 0000
status	Literate	190	72.80	52	19.92	242	92.72	P= 0.308*

Table 4 : Socio-Economic Characteristics of Respondents Based on Discrete Variables

Source: Survey results, 2014 \* significant association \*\* Not significantly associated

N= number of respondents, P = number of respondents' percentage

As regard to marital status, from the total sample respondents 50.7%, 49.43%, were married and single, respectively. The marital statuses of defaulters were also married and single, with the percentage of 6.9, 15.33, respectively. Whereas the marital statuses of non-defaulters were also married and single, with the percentage of 43.68, 34.10, at the same order. Statistically, it was found that the percentage differences between the two groups were insignificant (Table 4). This indicates that being single, married, divorced, and widowed have the same status either to repay or not to repay.

As regards to sex composition, 86(32.95%) were female respondents, whereas, 175(67.05%) were male respondents. The proportion of non-defaulters was 60(22.99%) for females, whereas, 143(54.79%) for male counter parts. This reveals that from their respective sex

composition, males' respondents were found having more repayment performance than female respondents. However, the chi-square result shows that the association between sex and loan repayment is significant ( $X^2 = 4.7617$  P= 0.029) table 4. This indicates that being either sex does determine loan repayment rate.

As regards to the educational status of the respondents, the survey results also revealed that 7.28 percent of the sample household heads were illiterate, whereas 92.72 percent of the household's heads were literate (Table 4). Of the total sample respondents, 4.9 percent of the non-defaulters and 2.3 percent of defaulters were illiterate respectively. There was no significant difference between defaulters and non defaulters in terms of their literacy level (Table 4).

Variables		Non- defaulters	Defaulters	Total sample (n=261)		x <sup>2</sup> & p values
				N	P	
Loan issued	No	62(23.75 %)	55(21.07%)	117	44.83	$\chi^2 = 75.3781$
timely	Yes	141(54.02)	3(1.15%)	144	55.17	P= 0.000
	Purchase of industrial products	18(6.9%)	2(.77%)	20	7.66	
	Construction of diary	12(4.6%)	0(.0%)	12	4.6	$\chi^2 = 27.9488$
Loan purpose	Purchase of cross bre	15(5.75%)	0(.0%)	15	5.75	P= 0.000
	Fill family	21(8.05)	0(.0%)	21	8.05	
	Settle debts	3(1,15)	0(.0%)	3	1.15	-
	Growing crops	11(4.21)	0(.0%)	11	4.21	-
	Other	123(47.13)	56(22.22%)	179	47.13	
	Unqualified loan officers	51(19.54%)	2(.77%)	53	20.31	$X^2 = 221.9382$
	Less speedy procedure	41(15.71%)	1(.38%)	42	16.09	P= 0.000
Why long loan	Long procedure	66(25.29%)	3(1.15%)	69	26.44	1
application	High no of	26(9.96)	0	26	9.96	

Table 5 : Characteristics of Lending Scheme, Purpose, Usage, and Size of Loan

	applicants					
	Non willing officers	18(6.9%)	0	18	6.90	
	Others	1(0.38%)	52(19.92%)	53	20.31	
Repayment	No	32(12.26%)	54(20.69%)	86	32.95	$\chi^2 = 122 1336$
suitable	Yes	171(65.52%)	4(1.53%)	175	67.05	P = 0.000
Repaying loan	No	0	54(20.69%)	54	20.69	$\chi^2 = 238,3043$
	Yes	203(77.78%)	4(1.53%)	207	79.31	P = 0.000
Repayment	Fully repaid on time	29(11.11)	0	29	11.11	$\chi^2 - 238.0307$
status	Fuly repaid too late	33(12.64)	0	33	12.64	P = 0.000
	Partially repaid on time	93(35.63)	1(.38%)	94	36.02	
	Partially repaid too late	48(18.39)	3(1.15)	51	19.54	
	Not repaid	0(0.00%)	54(20.69%)	54	20.69	
Receive	No	67(25.67%)	45(17.24%)	112	42.91	x <sup>2</sup> =36.5987
training	Yes	136(52.11%)	13(4.98%)	149	57.09	P= 0.000
Lent in group	No	82(31.42%)	20(7.66%)	102	39.08	$\chi^2 = 0.6621$
	Yes	121(46.36%)	38(14.56%)	159	60.92	P= 0.416

Source: Survey result, 2014. N = number of respondents P = percentage of respondents

With respect to loan disbursement, 55.7% of respondents have been answered that they took loan timely. On the other hand, 44.83% of respondents portrays that loan disbursement was delayed for number of weeks. The table 6 below shows that the higher proportion of non-defaulters 54.02% were found from those respondents who received timely disbursed loan, while only 23.75% of non-defaulter respondents were a group from who have received the loan delayed on disbursement.

According to respondents, this delay of disbursement was due to the absence of qualified loan officers and managers on the work time, 20.31%, Less speedy procedure 16.09%, and taking long procedure to finish precondition to deliver loan service 26.44%, High no of applicants 9.96%, Non willing officers 6.90% and others 20.31%. The chi-square result also shows the presence of strong and significant association between disbursement and dependant variable at 5% significance level ( $X^2$ =75.3781 at P= 0.000).

With respect to the purpose for which loan was taken, we observe that the majority of the borrowers, i.e., 179 (47.13%) took the loan for other purposes like Animal husbandry ,Horticulture ,Weaving and tailoring ,Food processing ,Metal work ,Wood work, Construction ,"Baltina" and petty trading ,Kiosk and shop ,Service provider both in urban rural areas. The next activity for which most of the borrowers took loan is to fill family requirements, 21 (8.05%).

To see if at all purpose of borrowing has some association with loan repayment performance, table 6 is constructed from the survey data. Accordingly only 47.13% of those who borrowed for the other purposes were non-defaulters. The same trend is observed in the rest of the cases. This indicates that purpose of borrowing may not have a notable implication on the loan repayment performance of borrowers. In fact this could be an issue for future research.

According to loan repayment status of respondents, it was found that 11.11% of non-defaulters had fully repaid on maturity time, and only 12.64% of them paid it fully but too late. The fully repayment of loan enables non-defaulters to gate the advantages of next higher loan, having good relationship with the lending institution, keeping their socially status in the society, realizing their freedom from any penalty.

Whereas, according to defaulters group, 0.38% of respondents had paid loan partially on maturity period and 1.15% of the same group had paid too late.

With respect to lending methodology about 60.92% of respondents were engaged in group lending scheme, while 39.08% of respondents were borrowed loan under individual lending scheme. The finding indicates that 14.56% of defaulter's proportions were involved in group lending methodology, whereas the remaining 7.66% of defaulters were categorized under individual lending methodology. In group lending methodology respondents had the chance to gate loan easily without formal collateral and personal guarantee, joint liability of group members used as collateral.

With regard to suitability of repayment period, 67.05% of respondents indicated that the loan repayment period was suitable; on the other hand 32.95% of other respondents revealed it was not. Based on findings, more defaulters' number (20.69%) of respondents was found in the group that replied period

was not suitable, while large numbers of non-defaulters (65.52%) were those who reported period was suitable. For the 20.69%% respondents who disagreed on suitability of period; the main reason was the shortness of grace and repayment period. The chi-square result also shows the presence of strong and significant association between repayment period and dependant variable at 5% significant level ( $x^2$ =122.1336 at P= 0.000).

In regarding to training, majority of respondents 57.9% indicated that they had received some kind of

training on business and about institutional services before receiving loans, while 42.91% responded that they had not received any training before receiving loans. As table 6 shows almost all respondents of nondefaulters were those who took training on business. Hence, the training variable has direct impact on loan repayment performance either to increase or decrease defaulting rate. Statistically, chi-square also confirms the presence of strong and a significant association between training and dependant variable at 5% level of significance (=36.5987 at P= 0.000).

Independent Variable	В	SE	Sig	Exp(B)
Sex of household	0.1305399	0.0763707	0.089	-0.0198839
Educational status of HH	0.223957	0.1352203	0.099	-0.0423806
Marital status of HH	0.075803	0.0837720	0.366	-0.0892004
Age of HH	0.00206	0.0051369	0.689	-0.0080577
Farming experience	0.01549	0.0101005	0.126	-0.004405
HH size	0.015245	0.016997	0.371	-0.0182328
Dependency Ratio	0.043917	0.02588	0.091	-0.0070577
Tropical livestock unit	0.000936	0.0003093	0.003	0.0003262
Off Farm income	-0.002894	0.0107956	0.789	-0.0241573
Value of equipment	0.027837	0.0093389	0.003	0.0094424
Receiving training	0.025597	0.0775924	0.742	-0.1272328
Lend in group	-0.072074	0.0804411	0.371	-0.2305149
Repayment suitability	0.58213	0.0984152	0.0001	0.388286
Loan amount	-0.019143	358382	0.594	-0.0897318
Loan frequency	0.034654	0.0251999	0.17	-0.0149813

Table 6 : Results of two limit Tobit model

Source: Survey result, 2014. B=regression coefficient, Exp (B) = odds ratio Overall, correct prediction = 89.9% Sig. =significance S.E = standard error Log pseudo likelihood = -201.05208 Pseudo R2 = 0.2020

A total of 15 explanatory variables were considered in the econometric model. Out of which **six variables** were found to be significant. These were sex of house hold, Education level, Number of dependants within and out house hold, Tropical live stock unit, Value of equipment, Repayment suitability. The coefficients of these all significant variables were negative and positive.

Sex of house hold: As regards to sex composition, 86(32.95%) were female respondents, whereas, 175(67.05%) were male respondents. The proportion of non-defaulters was 60(22.99%) for females, whereas, 143(54.79%) for male counter parts. This reveals that from their respective sex composition, males' respondents were found having more repayment performance than female respondents. However, the chi-square result shows that the association between sex and loan repayment is significant ( $x^2$ = 4.7617 P= 0.029) table 4. This indicates that being either sex does determine loan repayment rate.

*Education level:* The education level was positively and significantly influencing loan repayment at 1% significance level. An increase in one year schooling increases the probability of the loan repayment rate by 4.23806, ceteris paribus. This figure revels that the borrowers whose educational level increased have the probability of increasing the loan repayment performance four times more than the borrowers who have lesser education level/ illiterates. This suggests that more educated borrower may have access to business information.

Number of dependants within and out household: This variable was found to determine negatively and significantly borrowers' loan repayment performance at 1% significance level. If other variables held constant, having non-dependants or lower number of dependants' decreases the probability of defaulting by the 15.8%.

Livestock Ownership: Next to land, livestock is the most important asset for rural households in Ethiopia. It is

used as a source of food, draft power, income and energy. Moreover, livestock is an index of wealth and prestige in rural community. All the sample households reared livestock, which consisted of cattle, small ruminants, back animals and poultries. Total livestock ownership (LIVSTKNO) is, as expected, positively related to the dependent Variable (significant at 5% level). Each additional livestock ownership unit increases the probability being non defaulter by the mean value of 4.10695. The minimum number of livestock maintained was 0 whereas the maximum was 544.44 Credit users possessed relatively more livestock unit than non-users. The mean difference between the two groups in owning of livestock was significant at 1% level.

*Repayment suitability:* In regard to suitability of repayment period, 67.05% of respondents indicated that the loan repayment period was suitable; on the other hand 32.95% of other respondents revealed it was not. Based on findings, more defaulters' number (20.69%) of respondents was found in the group that replied period was not suitable, while large numbers of non-defaulters (65.52%) were those who reported period was suitable. For the 20.69%% respondents who disagreed on suitability of period; the main reason was the shortness of grace and repayment period. The chi-square result also shows the presence of strong and significant association between repayment period and dependant variable at 5% significant level( $X^2 = 122.1336$  at P= 0.000).

#### V. Conclusion and Recommendation

The finding of this study revealed that the age of respondents negatively and significantly determines the loan repayment performance of borrowers. This indicates that the elder respondents have better repayment performance than youngsters. And the elders were more responsible to repay loan than youngsters.

The researcher not recommends excluding youngsters. However, the care must be taken when starting from applicants' screening to through repayment periods, the special attention for follow up and supervision is necessary.

The education level determines loan repayment positively and significantly. The borrowers who attained higher education level able to pay better than the borrowers who were in lower level schooling and/or illiterates. Therefore, institution should motivate educated people and also easy to provide training.

Time lag between loan application and disbursement should be reduced to increase repayment rate. The complicated loan processing procedures, which might lead to delay in disbursement, further, it will increase default rate.

The supervision made by the loan officers and borrowers ratio should be reduced and it leads to

increase follow-up services. However, it is recommended that institution should compute thoroughly the borrowers' business proposal loan size before approving and sanctioning.

Borrowers who have small number of or no dependants in the household perform better in loan repayment. The borrowers who support large number of dependants also perform well with proper supervision.

Loan diversion was also found as essential and significant determinant of loan repayment rate negatively. This means, diverting loan into non-income generating activities increases default rate. Therefore, it is recommended that the institution should give attention to continuous follow-up on proper loan utilization.

Repayment period is also found to be a significant determinant of loan repayment performance of borrowers. Suitability of loan repayment period for borrowers was found to significantly increase the probability of repaying loan. Therefore, the institution has to give enough time to clients so that they will be able to work with the loans they have borrowed and arrange the time to collect loan that will be suitable for them to sell their business output.

#### References Références Referencias

- 1. Adeyemo, Remi (2004), Loan Delinquency in Multi-Purpose Cooperative Union in Kawara State, Nigeria, savings and development, Vol. VII, No. 3, pp. 267-274.
- Agu Cletus C. (2002), Analysis of the Determinants of the Nigerian Banking System's Profits and Profitability Performance <sup>2</sup>, Savings and Development, Vol. XVI, No. 4, 2002, pp. 353-371.
- Ajayi C.A. (2002) An Analysis of Default Factors in Residential Mortgages of the Federal Mortgage Bank of Nigeria<sup>2</sup>, African Review of Money, Finance and Banking, No. 2, pp.203-215.
- 4. Alemayehu Geda (2009) The Structure and Performance of Ethiopia's Financial Sector in the Pre and Post Reform Period: With Special Focus on Banking, in Alemayehu Geda and Birhanu Nega (ed.). The Ethiopian Economy: Performance and Evaluation: Proceedings of the Eighth Annual Conference on the Ethiopian Economy, pp.133-162.
- Andualem Tegegne (2007) Small Scale Enterprise and Entrepreneurship Development in Ethiopia Concepts, definitions and Major Issues, In Wolday Amha, G.H.R Chipande and Andualem Tegegne (ed.) Proceedings of the Sixth Annual Conference on the Ethiopian Economy.
- 6. Arene C.J (2002) Loan Repayment and Technical Assistance among Small Holder Maize Farmers in Nigeria, African Review of Money, Finance and Banking, No.1, pp.63-73.
- 7. Assefa Admassie (2007) Study of the Factors that Affect the Use of Agricultural Credit among Peasant

farmers in Ethiopia: the Case of Two Districts, Master's Thesis.

- 8. Asrat Tefera (2009) Development Finance for Small and Medium Industries, International Workshop on transfer of Technology for Small Scale and Medium Industries.
- Baker Chester B. and Dia Bernadette (2007) Default Management in Agricultural Lending Program in Ivory Coast, Savings and Development, Vol. XI, No.2, pp.161-177
- Birhanu Lakew 2009) Micro Enterprise Credit and Poverty Alleviation in Ethiopia: The Case of the Project Office for the Creation of Small Scale Business Opportunities (POCSSBO) in Addis Ababa, Master's Thesis, 2009.
- 11. *Bosa George R. (2009)* The Financing of Small and Medium Enterprises in Uganda, *Oxford University Press.*
- 12. Abafita, J. (2003) 'Microfinance and Ioan repayment performance: A Case Study of the Oromia credit and Savings Share Company (OCSSCO) in Kuyu', MSc thesis, Addis Ababa University, Addis Ababa.
- 13. Abreham Gebeyehu (2002), 'Loan repayment and its Determinants in Small-Scale Enterprises Financing in Ethiopia: Case of private borrowers Around Zeway Area', M. Sc. Thesis, AAU.
- Addisu, M. (2006) 'Micro-finance Repayment Problems in the Informal Sector in Addis Ababa', *Ethiopian Journal of Business & Development* Volume 1 Number 2 August 2006.
- 15. Admassie A., Ageba G. & Demeke M. (2005) 'Rural Finance in Ethiopia; Assessement of the financial products of microfinance institution', association of Ethiopian Microfinance Institutions, occasional paper No. 12.
- Amha, W. (2000) 'Review of Microfinance Industry in Ethiopia: Regulatory framework and performance,' Association of Ethiopian Microfinance Institution (AEMI), Addis Ababa, Ethiopia.
- 17. Armendariz, A. and Morduch, J. (2010) 'The Economics of Microfinance', 2nd ed. The MIT Press Cambridge, Massachusetts London, England.
- Asian Development Bank (ADB), 2000, 'Finance for the poor: Microfinance development Strategy', Rural Asia study: beyond the Green revolution. Manila.
- 19. Assefa B. A. (2002) 'Factors influencing loan repayment of rural women in Eastern Ethiopia: the case of Dire Dawa Area', A Thesis presented to the school of graduate studies, Alemaya Univeristy, Ethiopia.
- 20. Auwal, M.A. (1996) 'Promoting micro capitalism in service of the poor: the Grameen model and its cross-cultural adaptation', *Journal of Business Communication* 33.1, 27–49.
- 21. Basu S. (2005) 'Securitization and the Challenges Faced in Micro Finance', Centre for Micro Finance

Research Working Paper Series, Institute for Financial Mangement and Research (IFMR).

22. Basu, A., Blavy, R. and Yulek, M. (2004) 'The Roles of Donors and NGOs: Microfinance in Africa Experience and Lessons from Selected African Countries', IMF Working Paper.

### GLOBAL JOURNALS INC. (US) GUIDELINES HANDBOOK 2016

WWW.GLOBALJOURNALS.ORG

#### Fellows

#### FELLOW OF ASSOCIATION OF RESEARCH SOCIETY IN BUSINESS (FARSB)

Global Journals Incorporate (USA) is accredited by Open Association of Research Society (OARS), U.S.A and in turn, awards "FARSB" title to individuals. The 'FARSB' title is accorded to a selected professional after the approval of the Editor-in-Chief/Editorial Board Members/Dean.



The "FARSB" is a dignified title which is accorded to a person's name viz. Dr. John E. Hall, Ph.D., FARSB or William Walldroff, M.S., FARSB.

FARSB accrediting is an honor. It authenticates your research activities. After recognition as FARSB, you can add 'FARSB' title with your name as you use this recognition as additional suffix to your status. This will definitely enhance and add more value and repute to your name. You may use it on your professional Counseling Materials such as CV, Resume, and Visiting Card etc.

The following benefits can be availed by you only for next three years from the date of certification:



FARSB designated members are entitled to avail a 40% discount while publishing their research papers (of a single author) with Global Journals Incorporation (USA), if the same is accepted by Editorial Board/Peer Reviewers. If you are a main author or co-author in case of multiple authors, you will be entitled to avail discount of 10%.

Once FARSB title is accorded, the Fellow is authorized to organize a symposium/seminar/conference on behalf of Global Journal Incorporation (USA). The Fellow can also participate in conference/seminar/symposium organized by another institution as representative of Global Journal. In both the cases, it is mandatory for him to discuss with us and obtain our consent.





You may join as member of the Editorial Board of Global Journals Incorporation (USA) after successful completion of three years as Fellow and as Peer Reviewer. In addition, it is also desirable that you should organize seminar/symposium/conference at least once.

We shall provide you intimation regarding launching of e-version of journal of your stream time to time. This may be utilized in your library for the enrichment of knowledge of your students as well as it can also be helpful for the concerned faculty members.



© Copyright by Global Journals Inc.(US) | Guidelines Handbook
EARSE YOU

As FARSB, you will be given a renowned, secure and free professional email address with 100 GB of space e.g. johnhall@globaljournals.org. This will include Webmail, Spam Assassin, Email Forwarders, Auto-Responders, Email Delivery Route tracing, etc.

benefit of entire research community.

The FARSB will be eligible for a free application of standardization of their researches. Standardization of research will be subject to

acceptability within stipulated norms as the next step after publishing in a journal. We shall depute a team of specialized research professionals who will render their services for elevating your researches to next higher level, which is worldwide open standardization.

The FARSB can go through standards of OARS. You can also play vital role if you have any suggestions so that proper amendment can take place to improve the same for the

The FARSB member can apply for grading and certification of standards of their educational and Institutional Degrees to Open Association of Research, Society U.S.A. Once you are designated as FARSB, you may send us a scanned copy of all of your credentials. OARS will verify, grade and certify them. This will be based on your academic records, quality of research papers published by you, and some more criteria. After certification of all your credentials by OARS, they will be published on

your Fellow Profile link on website <u>https://associationofresearch.org</u> which will be helpful to upgrade the dignity.

Global RESEARCH RADIC

The FARSB members can avail the benefits of free research podcasting in Global Research Radio with their research documents. After publishing the work, (including published elsewhere worldwide with proper authorization) you can upload your research paper with your recorded voice or you can utilize chargeable

services of our professional RJs to record your paper in their voice on request.

The FARSB member also entitled to get the benefits of free research podcasting of their research documents through video clips. We can also streamline your conference videos and display your slides/ online slides and online research video clips at reasonable charges, on request.











The FARSB is eligible to earn from sales proceeds of his/her researches/reference/review Books or literature, while publishing with Global Journals. The FARSB can decide whether he/she would like to publish his/her research in a closed manner. In this case, whenever readers purchase that individual research paper for reading, maximum 60% of its profit earned as royalty by Global Journals, will be credited to his/her bank account. The entire entitled amount will be credited to

his/her bank account exceeding limit of minimum fixed balance. There is no minimum time limit for collection. The FARSC member can decide its price and we can help in making the right decision.

The FARSB member is eligible to join as a paid peer reviewer at Global Journals Incorporation (USA) and can get remuneration of 15% of author fees, taken from the author of a respective paper. After reviewing 5 or more papers you can request to transfer the amount to your bank account.

## MEMBER OF ASSOCIATION OF RESEARCH SOCIETY IN BUSINESS (MARSB)

The 'MARSB ' title is accorded to a selected professional after the approval of the Editor-in-Chief / Editorial Board Members/Dean.



The "MARSB" is a dignified ornament which is accorded to a person's name viz. Dr. John E. Hall, Ph.D., MARSB or William Walldroff, M.S., MARSB.

MARSB accrediting is an honor. It authenticates your research activities. After becoming MARSB, you can add 'MARSB' title with your name as you use this recognition as additional suffix to your status. This will definitely enhance and add more value and repute to your name. You may use it on your professional Counseling Materials such as CV, Resume, Visiting Card and Name Plate etc.

The following benefitscan be availed by you only for next three years from the date of certification.



MARSB designated members are entitled to avail a 25% discount while publishing their research papers (of a single author) in Global Journals Inc., if the same is accepted by our Editorial Board and Peer Reviewers. If you are a main author or co-author of a group of authors, you will get discount of 10%.

As MARSB, you will be given a renowned, secure and free professional email address with 30 GB of space e.g. johnhall@globaljournals.org. This will include Webmail, Spam Assassin, Email Forwarders, Auto-Responders, Email Delivery Route tracing, etc.





We shall provide you intimation regarding launching of e-version of journal of your stream time to time. This may be utilized in your library for the enrichment of knowledge of your students as well as it can also be helpful for the concerned faculty members.

The MARSB member can apply for approval, grading and certification of standards of their educational and Institutional Degrees to Open Association of Research, Society U.S.A.





Once you are designated as MARSB, you may send us a scanned copy of all of your credentials. OARS will verify, grade and certify them. This will be based on your academic records, quality of research papers published by you, and some more criteria.

It is mandatory to read all terms and conditions carefully.

## AUXILIARY MEMBERSHIPS

## Institutional Fellow of Open Association of Research Society (USA)-OARS (USA)

Global Journals Incorporation (USA) is accredited by Open Association of Research Society, U.S.A (OARS) and in turn, affiliates research institutions as "Institutional Fellow of Open Association of Research Society" (IFOARS).

The "FARSC" is a dignified title which is accorded to a person's name viz. Dr. John E. Hall, Ph.D., FARSC or William Walldroff, M.S., FARSC.

The IFOARS institution is entitled to form a Board comprised of one Chairperson and three to five board members preferably from different streams. The Board will be recognized as "Institutional Board of Open Association of Research Society"-(IBOARS).

The Institute will be entitled to following benefits:



The IBOARS can initially review research papers of their institute and recommend them to publish with respective journal of Global Journals. It can also review the papers of other institutions after obtaining our consent. The second review will be done by peer reviewer of Global Journals Incorporation (USA) The Board is at liberty to appoint a peer reviewer with the approval of chairperson after consulting us.

The author fees of such paper may be waived off up to 40%.

The Global Journals Incorporation (USA) at its discretion can also refer double blind peer reviewed paper at their end to the board for the verification and to get recommendation for final stage of acceptance of publication.





The IBOARS can organize symposium/seminar/conference in their country on seminar of Global Journals Incorporation (USA)-OARS (USA). The terms and conditions can be discussed separately.

The Board can also play vital role by exploring and giving valuable suggestions regarding the Standards of "Open Association of Research Society, U.S.A (OARS)" so that proper amendment can take place for the benefit of entire research community. We shall provide details of particular standard only on receipt of request from the Board.





The board members can also join us as Individual Fellow with 40% discount on total fees applicable to Individual Fellow. They will be entitled to avail all the benefits as declared. Please visit Individual Fellow-sub menu of GlobalJournals.org to have more relevant details.

Journals Research relevant details.

We shall provide you intimation regarding launching of e-version of journal of your stream time to time. This may be utilized in your library for the enrichment of knowledge of your students as well as it can also be helpful for the concerned faculty members.



After nomination of your institution as "Institutional Fellow" and constantly functioning successfully for one year, we can consider giving recognition to your institute to function as Regional/Zonal office on our behalf.

The board can also take up the additional allied activities for betterment after our consultation.

## The following entitlements are applicable to individual Fellows:

Open Association of Research Society, U.S.A (OARS) By-laws states that an individual Fellow may use the designations as applicable, or the corresponding initials. The Credentials of individual Fellow and Associate designations signify that the individual has gained knowledge of the fundamental concepts. One is magnanimous and proficient in an expertise course covering the professional code of conduct, and follows recognized standards of practice.





Open Association of Research Society (US)/ Global Journals Incorporation (USA), as described in Corporate Statements, are educational, research publishing and professional membership organizations. Achieving our individual Fellow or Associate status is based mainly on meeting stated educational research requirements.

Disbursement of 40% Royalty earned through Global Journals : Researcher = 50%, Peer Reviewer = 37.50%, Institution = 12.50% E.g. Out of 40%, the 20% benefit should be passed on to researcher, 15 % benefit towards remuneration should be given to a reviewer and remaining 5% is to be retained by the institution.



We shall provide print version of 12 issues of any three journals [as per your requirement] out of our 38 journals worth \$ 2376 USD.

## Other:

## The individual Fellow and Associate designations accredited by Open Association of Research Society (US) credentials signify guarantees following achievements:

- The professional accredited with Fellow honor, is entitled to various benefits viz. name, fame, honor, regular flow of income, secured bright future, social status etc.
  - © Copyright by Global Journals Inc.(US) | Guidelines Handbook

- In addition to above, if one is single author, then entitled to 40% discount on publishing research paper and can get 10% discount if one is co-author or main author among group of authors.
- The Fellow can organize symposium/seminar/conference on behalf of Global Journals Incorporation (USA) and he/she can also attend the same organized by other institutes on behalf of Global Journals.
- > The Fellow can become member of Editorial Board Member after completing 3yrs.
- > The Fellow can earn 60% of sales proceeds from the sale of reference/review books/literature/publishing of research paper.
- Fellow can also join as paid peer reviewer and earn 15% remuneration of author charges and can also get an opportunity to join as member of the Editorial Board of Global Journals Incorporation (USA)
- This individual has learned the basic methods of applying those concepts and techniques to common challenging situations. This individual has further demonstrated an in-depth understanding of the application of suitable techniques to a particular area of research practice.

## Note :

- In future, if the board feels the necessity to change any board member, the same can be done with the consent of the chairperson along with anyone board member without our approval.
- In case, the chairperson needs to be replaced then consent of 2/3rd board members are required and they are also required to jointly pass the resolution copy of which should be sent to us. In such case, it will be compulsory to obtain our approval before replacement.
- In case of "Difference of Opinion [if any]" among the Board members, our decision will be final and binding to everyone.

The Area or field of specialization may or may not be of any category as mentioned in 'Scope of Journal' menu of the GlobalJournals.org website. There are 37 Research Journal categorized with Six parental Journals GJCST, GJMR, GJRE, GJMBR, GJSFR, GJHSS. For Authors should prefer the mentioned categories. There are three widely used systems UDC, DDC and LCC. The details are available as 'Knowledge Abstract' at Home page. The major advantage of this coding is that, the research work will be exposed to and shared with all over the world as we are being abstracted and indexed worldwide.

The paper should be in proper format. The format can be downloaded from first page of 'Author Guideline' Menu. The Author is expected to follow the general rules as mentioned in this menu. The paper should be written in MS-Word Format (\*.DOC,\*.DOCX).

The Author can submit the paper either online or offline. The authors should prefer online submission.<u>Online Submission</u>: There are three ways to submit your paper:

(A) (I) First, register yourself using top right corner of Home page then Login. If you are already registered, then login using your username and password.

(II) Choose corresponding Journal.

(III) Click 'Submit Manuscript'. Fill required information and Upload the paper.

(B) If you are using Internet Explorer, then Direct Submission through Homepage is also available.

(C) If these two are not convenient, and then email the paper directly to dean@globaljournals.org.

Offline Submission: Author can send the typed form of paper by Post. However, online submission should be preferred.

## PREFERRED AUTHOR GUIDELINES

#### MANUSCRIPT STYLE INSTRUCTION (Must be strictly followed)

Page Size: 8.27" X 11'"

- Left Margin: 0.65
- Right Margin: 0.65
- Top Margin: 0.75
- Bottom Margin: 0.75
- Font type of all text should be Swis 721 Lt BT.
- Paper Title should be of Font Size 24 with one Column section.
- Author Name in Font Size of 11 with one column as of Title.
- Abstract Font size of 9 Bold, "Abstract" word in Italic Bold.
- Main Text: Font size 10 with justified two columns section
- Two Column with Equal Column with of 3.38 and Gaping of .2
- First Character must be three lines Drop capped.
- Paragraph before Spacing of 1 pt and After of 0 pt.
- Line Spacing of 1 pt
- Large Images must be in One Column
- Numbering of First Main Headings (Heading 1) must be in Roman Letters, Capital Letter, and Font Size of 10.
- Numbering of Second Main Headings (Heading 2) must be in Alphabets, Italic, and Font Size of 10.

#### You can use your own standard format also. Author Guidelines:

1. General,

- 2. Ethical Guidelines,
- 3. Submission of Manuscripts,
- 4. Manuscript's Category,
- 5. Structure and Format of Manuscript,
- 6. After Acceptance.

#### 1. GENERAL

Before submitting your research paper, one is advised to go through the details as mentioned in following heads. It will be beneficial, while peer reviewer justify your paper for publication.

#### Scope

The Global Journals Inc. (US) welcome the submission of original paper, review paper, survey article relevant to the all the streams of Philosophy and knowledge. The Global Journals Inc. (US) is parental platform for Global Journal of Computer Science and Technology, Researches in Engineering, Medical Research, Science Frontier Research, Human Social Science, Management, and Business organization. The choice of specific field can be done otherwise as following in Abstracting and Indexing Page on this Website. As the all Global

Journals Inc. (US) are being abstracted and indexed (in process) by most of the reputed organizations. Topics of only narrow interest will not be accepted unless they have wider potential or consequences.

#### 2. ETHICAL GUIDELINES

Authors should follow the ethical guidelines as mentioned below for publication of research paper and research activities.

Papers are accepted on strict understanding that the material in whole or in part has not been, nor is being, considered for publication elsewhere. If the paper once accepted by Global Journals Inc. (US) and Editorial Board, will become the copyright of the Global Journals Inc. (US).

#### Authorship: The authors and coauthors should have active contribution to conception design, analysis and interpretation of findings. They should critically review the contents and drafting of the paper. All should approve the final version of the paper before submission

The Global Journals Inc. (US) follows the definition of authorship set up by the Global Academy of Research and Development. According to the Global Academy of R&D authorship, criteria must be based on:

1) Substantial contributions to conception and acquisition of data, analysis and interpretation of the findings.

2) Drafting the paper and revising it critically regarding important academic content.

3) Final approval of the version of the paper to be published.

All authors should have been credited according to their appropriate contribution in research activity and preparing paper. Contributors who do not match the criteria as authors may be mentioned under Acknowledgement.

Acknowledgements: Contributors to the research other than authors credited should be mentioned under acknowledgement. The specifications of the source of funding for the research if appropriate can be included. Suppliers of resources may be mentioned along with address.

#### Appeal of Decision: The Editorial Board's decision on publication of the paper is final and cannot be appealed elsewhere.

## Permissions: It is the author's responsibility to have prior permission if all or parts of earlier published illustrations are used in this paper.

Please mention proper reference and appropriate acknowledgements wherever expected.

If all or parts of previously published illustrations are used, permission must be taken from the copyright holder concerned. It is the author's responsibility to take these in writing.

Approval for reproduction/modification of any information (including figures and tables) published elsewhere must be obtained by the authors/copyright holders before submission of the manuscript. Contributors (Authors) are responsible for any copyright fee involved.

#### **3. SUBMISSION OF MANUSCRIPTS**

Manuscripts should be uploaded via this online submission page. The online submission is most efficient method for submission of papers, as it enables rapid distribution of manuscripts and consequently speeds up the review procedure. It also enables authors to know the status of their own manuscripts by emailing us. Complete instructions for submitting a paper is available below.

Manuscript submission is a systematic procedure and little preparation is required beyond having all parts of your manuscript in a given format and a computer with an Internet connection and a Web browser. Full help and instructions are provided on-screen. As an author, you will be prompted for login and manuscript details as Field of Paper and then to upload your manuscript file(s) according to the instructions.



To avoid postal delays, all transaction is preferred by e-mail. A finished manuscript submission is confirmed by e-mail immediately and your paper enters the editorial process with no postal delays. When a conclusion is made about the publication of your paper by our Editorial Board, revisions can be submitted online with the same procedure, with an occasion to view and respond to all comments.

Complete support for both authors and co-author is provided.

#### 4. MANUSCRIPT'S CATEGORY

Based on potential and nature, the manuscript can be categorized under the following heads:

Original research paper: Such papers are reports of high-level significant original research work.

Review papers: These are concise, significant but helpful and decisive topics for young researchers.

Research articles: These are handled with small investigation and applications

Research letters: The letters are small and concise comments on previously published matters.

#### **5.STRUCTURE AND FORMAT OF MANUSCRIPT**

The recommended size of original research paper is less than seven thousand words, review papers fewer than seven thousands words also. Preparation of research paper or how to write research paper, are major hurdle, while writing manuscript. The research articles and research letters should be fewer than three thousand words, the structure original research paper; sometime review paper should be as follows:

**Papers**: These are reports of significant research (typically less than 7000 words equivalent, including tables, figures, references), and comprise:

(a)Title should be relevant and commensurate with the theme of the paper.

(b) A brief Summary, "Abstract" (less than 150 words) containing the major results and conclusions.

(c) Up to ten keywords, that precisely identifies the paper's subject, purpose, and focus.

(d) An Introduction, giving necessary background excluding subheadings; objectives must be clearly declared.

(e) Resources and techniques with sufficient complete experimental details (wherever possible by reference) to permit repetition; sources of information must be given and numerical methods must be specified by reference, unless non-standard.

(f) Results should be presented concisely, by well-designed tables and/or figures; the same data may not be used in both; suitable statistical data should be given. All data must be obtained with attention to numerical detail in the planning stage. As reproduced design has been recognized to be important to experiments for a considerable time, the Editor has decided that any paper that appears not to have adequate numerical treatments of the data will be returned un-refereed;

(g) Discussion should cover the implications and consequences, not just recapitulating the results; conclusions should be summarizing.

(h) Brief Acknowledgements.

(i) References in the proper form.

Authors should very cautiously consider the preparation of papers to ensure that they communicate efficiently. Papers are much more likely to be accepted, if they are cautiously designed and laid out, contain few or no errors, are summarizing, and be conventional to the approach and instructions. They will in addition, be published with much less delays than those that require much technical and editorial correction.

The Editorial Board reserves the right to make literary corrections and to make suggestions to improve briefness.

It is vital, that authors take care in submitting a manuscript that is written in simple language and adheres to published guidelines.

#### Format

Language: The language of publication is UK English. Authors, for whom English is a second language, must have their manuscript efficiently edited by an English-speaking person before submission to make sure that, the English is of high excellence. It is preferable, that manuscripts should be professionally edited.

Standard Usage, Abbreviations, and Units: Spelling and hyphenation should be conventional to The Concise Oxford English Dictionary. Statistics and measurements should at all times be given in figures, e.g. 16 min, except for when the number begins a sentence. When the number does not refer to a unit of measurement it should be spelt in full unless, it is 160 or greater.

Abbreviations supposed to be used carefully. The abbreviated name or expression is supposed to be cited in full at first usage, followed by the conventional abbreviation in parentheses.

Metric SI units are supposed to generally be used excluding where they conflict with current practice or are confusing. For illustration, 1.4 I rather than  $1.4 \times 10-3$  m3, or 4 mm somewhat than  $4 \times 10-3$  m. Chemical formula and solutions must identify the form used, e.g. anhydrous or hydrated, and the concentration must be in clearly defined units. Common species names should be followed by underlines at the first mention. For following use the generic name should be constricted to a single letter, if it is clear.

#### Structure

All manuscripts submitted to Global Journals Inc. (US), ought to include:

Title: The title page must carry an instructive title that reflects the content, a running title (less than 45 characters together with spaces), names of the authors and co-authors, and the place(s) wherever the work was carried out. The full postal address in addition with the e-mail address of related author must be given. Up to eleven keywords or very brief phrases have to be given to help data retrieval, mining and indexing.

Abstract, used in Original Papers and Reviews:

Optimizing Abstract for Search Engines

Many researchers searching for information online will use search engines such as Google, Yahoo or similar. By optimizing your paper for search engines, you will amplify the chance of someone finding it. This in turn will make it more likely to be viewed and/or cited in a further work. Global Journals Inc. (US) have compiled these guidelines to facilitate you to maximize the web-friendliness of the most public part of your paper.

#### Key Words

A major linchpin in research work for the writing research paper is the keyword search, which one will employ to find both library and Internet resources.

One must be persistent and creative in using keywords. An effective keyword search requires a strategy and planning a list of possible keywords and phrases to try.

Search engines for most searches, use Boolean searching, which is somewhat different from Internet searches. The Boolean search uses "operators," words (and, or, not, and near) that enable you to expand or narrow your affords. Tips for research paper while preparing research paper are very helpful guideline of research paper.

Choice of key words is first tool of tips to write research paper. Research paper writing is an art.A few tips for deciding as strategically as possible about keyword search:



- One should start brainstorming lists of possible keywords before even begin searching. Think about the most important concepts related to research work. Ask, "What words would a source have to include to be truly valuable in research paper?" Then consider synonyms for the important words.
- It may take the discovery of only one relevant paper to let steer in the right keyword direction because in most databases, the keywords under which a research paper is abstracted are listed with the paper.
- One should avoid outdated words.

Keywords are the key that opens a door to research work sources. Keyword searching is an art in which researcher's skills are bound to improve with experience and time.

Numerical Methods: Numerical methods used should be clear and, where appropriate, supported by references.

Acknowledgements: Please make these as concise as possible.

#### References

References follow the Harvard scheme of referencing. References in the text should cite the authors' names followed by the time of their publication, unless there are three or more authors when simply the first author's name is quoted followed by et al. unpublished work has to only be cited where necessary, and only in the text. Copies of references in press in other journals have to be supplied with submitted typescripts. It is necessary that all citations and references be carefully checked before submission, as mistakes or omissions will cause delays.

References to information on the World Wide Web can be given, but only if the information is available without charge to readers on an official site. Wikipedia and Similar websites are not allowed where anyone can change the information. Authors will be asked to make available electronic copies of the cited information for inclusion on the Global Journals Inc. (US) homepage at the judgment of the Editorial Board.

The Editorial Board and Global Journals Inc. (US) recommend that, citation of online-published papers and other material should be done via a DOI (digital object identifier). If an author cites anything, which does not have a DOI, they run the risk of the cited material not being noticeable.

The Editorial Board and Global Journals Inc. (US) recommend the use of a tool such as Reference Manager for reference management and formatting.

#### Tables, Figures and Figure Legends

Tables: Tables should be few in number, cautiously designed, uncrowned, and include only essential data. Each must have an Arabic number, e.g. Table 4, a self-explanatory caption and be on a separate sheet. Vertical lines should not be used.

*Figures: Figures are supposed to be submitted as separate files. Always take in a citation in the text for each figure using Arabic numbers, e.g. Fig. 4. Artwork must be submitted online in electronic form by e-mailing them.* 

#### Preparation of Electronic Figures for Publication

Even though low quality images are sufficient for review purposes, print publication requires high quality images to prevent the final product being blurred or fuzzy. Submit (or e-mail) EPS (line art) or TIFF (halftone/photographs) files only. MS PowerPoint and Word Graphics are unsuitable for printed pictures. Do not use pixel-oriented software. Scans (TIFF only) should have a resolution of at least 350 dpi (halftone) or 700 to 1100 dpi (line drawings) in relation to the imitation size. Please give the data for figures in black and white or submit a Color Work Agreement Form. EPS files must be saved with fonts embedded (and with a TIFF preview, if possible).

For scanned images, the scanning resolution (at final image size) ought to be as follows to ensure good reproduction: line art: >650 dpi; halftones (including gel photographs) : >350 dpi; figures containing both halftone and line images: >650 dpi.

Color Charges: It is the rule of the Global Journals Inc. (US) for authors to pay the full cost for the reproduction of their color artwork. Hence, please note that, if there is color artwork in your manuscript when it is accepted for publication, we would require you to complete and return a color work agreement form before your paper can be published.

Figure Legends: Self-explanatory legends of all figures should be incorporated separately under the heading 'Legends to Figures'. In the full-text online edition of the journal, figure legends may possibly be truncated in abbreviated links to the full screen version. Therefore, the first 100 characters of any legend should notify the reader, about the key aspects of the figure.

#### 6. AFTER ACCEPTANCE

Upon approval of a paper for publication, the manuscript will be forwarded to the dean, who is responsible for the publication of the Global Journals Inc. (US).

#### **6.1 Proof Corrections**

The corresponding author will receive an e-mail alert containing a link to a website or will be attached. A working e-mail address must therefore be provided for the related author.

Acrobat Reader will be required in order to read this file. This software can be downloaded

(Free of charge) from the following website:

www.adobe.com/products/acrobat/readstep2.html. This will facilitate the file to be opened, read on screen, and printed out in order for any corrections to be added. Further instructions will be sent with the proof.

Proofs must be returned to the dean at dean@globaljournals.org within three days of receipt.

As changes to proofs are costly, we inquire that you only correct typesetting errors. All illustrations are retained by the publisher. Please note that the authors are responsible for all statements made in their work, including changes made by the copy editor.

#### 6.2 Early View of Global Journals Inc. (US) (Publication Prior to Print)

The Global Journals Inc. (US) are enclosed by our publishing's Early View service. Early View articles are complete full-text articles sent in advance of their publication. Early View articles are absolute and final. They have been completely reviewed, revised and edited for publication, and the authors' final corrections have been incorporated. Because they are in final form, no changes can be made after sending them. The nature of Early View articles means that they do not yet have volume, issue or page numbers, so Early View articles cannot be cited in the conventional way.

#### 6.3 Author Services

Online production tracking is available for your article through Author Services. Author Services enables authors to track their article - once it has been accepted - through the production process to publication online and in print. Authors can check the status of their articles online and choose to receive automated e-mails at key stages of production. The authors will receive an e-mail with a unique link that enables them to register and have their article automatically added to the system. Please ensure that a complete e-mail address is provided when submitting the manuscript.

#### 6.4 Author Material Archive Policy

Please note that if not specifically requested, publisher will dispose off hardcopy & electronic information submitted, after the two months of publication. If you require the return of any information submitted, please inform the Editorial Board or dean as soon as possible.

#### 6.5 Offprint and Extra Copies

A PDF offprint of the online-published article will be provided free of charge to the related author, and may be distributed according to the Publisher's terms and conditions. Additional paper offprint may be ordered by emailing us at: editor@globaljournals.org.

You must strictly follow above Author Guidelines before submitting your paper or else we will not at all be responsible for any corrections in future in any of the way.



Before start writing a good quality Computer Science Research Paper, let us first understand what is Computer Science Research Paper? So, Computer Science Research Paper is the paper which is written by professionals or scientists who are associated to Computer Science and Information Technology, or doing research study in these areas. If you are novel to this field then you can consult about this field from your supervisor or guide.

#### TECHNIQUES FOR WRITING A GOOD QUALITY RESEARCH PAPER:

1. Choosing the topic: In most cases, the topic is searched by the interest of author but it can be also suggested by the guides. You can have several topics and then you can judge that in which topic or subject you are finding yourself most comfortable. This can be done by asking several questions to yourself, like Will I be able to carry our search in this area? Will I find all necessary recourses to accomplish the search? Will I be able to find all information in this field area? If the answer of these types of questions will be "Yes" then you can choose that topic. In most of the cases, you may have to conduct the surveys and have to visit several places because this field is related to Computer Science and Information Technology. Also, you may have to do a lot of work to find all rise and falls regarding the various data of that subject. Sometimes, detailed information plays a vital role, instead of short information.

**2. Evaluators are human:** First thing to remember that evaluators are also human being. They are not only meant for rejecting a paper. They are here to evaluate your paper. So, present your Best.

**3. Think Like Evaluators:** If you are in a confusion or getting demotivated that your paper will be accepted by evaluators or not, then think and try to evaluate your paper like an Evaluator. Try to understand that what an evaluator wants in your research paper and automatically you will have your answer.

**4. Make blueprints of paper:** The outline is the plan or framework that will help you to arrange your thoughts. It will make your paper logical. But remember that all points of your outline must be related to the topic you have chosen.

**5.** Ask your Guides: If you are having any difficulty in your research, then do not hesitate to share your difficulty to your guide (if you have any). They will surely help you out and resolve your doubts. If you can't clarify what exactly you require for your work then ask the supervisor to help you with the alternative. He might also provide you the list of essential readings.

6. Use of computer is recommended: As you are doing research in the field of Computer Science, then this point is quite obvious.

7. Use right software: Always use good quality software packages. If you are not capable to judge good software then you can lose quality of your paper unknowingly. There are various software programs available to help you, which you can get through Internet.

8. Use the Internet for help: An excellent start for your paper can be by using the Google. It is an excellent search engine, where you can have your doubts resolved. You may also read some answers for the frequent question how to write my research paper or find model research paper. From the internet library you can download books. If you have all required books make important reading selecting and analyzing the specified information. Then put together research paper sketch out.

9. Use and get big pictures: Always use encyclopedias, Wikipedia to get pictures so that you can go into the depth.

**10.** Bookmarks are useful: When you read any book or magazine, you generally use bookmarks, right! It is a good habit, which helps to not to lose your continuity. You should always use bookmarks while searching on Internet also, which will make your search easier.

11. Revise what you wrote: When you write anything, always read it, summarize it and then finalize it.

**12.** Make all efforts: Make all efforts to mention what you are going to write in your paper. That means always have a good start. Try to mention everything in introduction, that what is the need of a particular research paper. Polish your work by good skill of writing and always give an evaluator, what he wants.

**13.** Have backups: When you are going to do any important thing like making research paper, you should always have backup copies of it either in your computer or in paper. This will help you to not to lose any of your important.

**14. Produce good diagrams of your own:** Always try to include good charts or diagrams in your paper to improve quality. Using several and unnecessary diagrams will degrade the quality of your paper by creating "hotchpotch." So always, try to make and include those diagrams, which are made by your own to improve readability and understandability of your paper.

**15.** Use of direct quotes: When you do research relevant to literature, history or current affairs then use of quotes become essential but if study is relevant to science then use of quotes is not preferable.

**16.** Use proper verb tense: Use proper verb tenses in your paper. Use past tense, to present those events that happened. Use present tense to indicate events that are going on. Use future tense to indicate future happening events. Use of improper and wrong tenses will confuse the evaluator. Avoid the sentences that are incomplete.

**17.** Never use online paper: If you are getting any paper on Internet, then never use it as your research paper because it might be possible that evaluator has already seen it or maybe it is outdated version.

**18.** Pick a good study spot: To do your research studies always try to pick a spot, which is quiet. Every spot is not for studies. Spot that suits you choose it and proceed further.

**19. Know what you know:** Always try to know, what you know by making objectives. Else, you will be confused and cannot achieve your target.

**20.** Use good quality grammar: Always use a good quality grammar and use words that will throw positive impact on evaluator. Use of good quality grammar does not mean to use tough words, that for each word the evaluator has to go through dictionary. Do not start sentence with a conjunction. Do not fragment sentences. Eliminate one-word sentences. Ignore passive voice. Do not ever use a big word when a diminutive one would suffice. Verbs have to be in agreement with their subjects. Prepositions are not expressions to finish sentences with. It is incorrect to ever divide an infinitive. Avoid clichés like the disease. Also, always shun irritating alliteration. Use language that is simple and straight forward. put together a neat summary.

**21.** Arrangement of information: Each section of the main body should start with an opening sentence and there should be a changeover at the end of the section. Give only valid and powerful arguments to your topic. You may also maintain your arguments with records.

**22.** Never start in last minute: Always start at right time and give enough time to research work. Leaving everything to the last minute will degrade your paper and spoil your work.

**23.** Multitasking in research is not good: Doing several things at the same time proves bad habit in case of research activity. Research is an area, where everything has a particular time slot. Divide your research work in parts and do particular part in particular time slot.

24. Never copy others' work: Never copy others' work and give it your name because if evaluator has seen it anywhere you will be in trouble.

**25.** Take proper rest and food: No matter how many hours you spend for your research activity, if you are not taking care of your health then all your efforts will be in vain. For a quality research, study is must, and this can be done by taking proper rest and food.

26. Go for seminars: Attend seminars if the topic is relevant to your research area. Utilize all your resources.

**27. Refresh your mind after intervals:** Try to give rest to your mind by listening to soft music or by sleeping in intervals. This will also improve your memory.

**28. Make colleagues:** Always try to make colleagues. No matter how sharper or intelligent you are, if you make colleagues you can have several ideas, which will be helpful for your research.

29. Think technically: Always think technically. If anything happens, then search its reasons, its benefits, and demerits.

**30.** Think and then print: When you will go to print your paper, notice that tables are not be split, headings are not detached from their descriptions, and page sequence is maintained.

**31.** Adding unnecessary information: Do not add unnecessary information, like, I have used MS Excel to draw graph. Do not add irrelevant and inappropriate material. These all will create superfluous. Foreign terminology and phrases are not apropos. One should NEVER take a broad view. Analogy in script is like feathers on a snake. Not at all use a large word when a very small one would be sufficient. Use words properly, regardless of how others use them. Remove quotations. Puns are for kids, not grunt readers. Amplification is a billion times of inferior quality than sarcasm.

**32.** Never oversimplify everything: To add material in your research paper, never go for oversimplification. This will definitely irritate the evaluator. Be more or less specific. Also too, by no means, ever use rhythmic redundancies. Contractions aren't essential and shouldn't be there used. Comparisons are as terrible as clichés. Give up ampersands and abbreviations, and so on. Remove commas, that are, not necessary. Parenthetical words however should be together with this in commas. Understatement is all the time the complete best way to put onward earth-shaking thoughts. Give a detailed literary review.

**33. Report concluded results:** Use concluded results. From raw data, filter the results and then conclude your studies based on measurements and observations taken. Significant figures and appropriate number of decimal places should be used. Parenthetical remarks are prohibitive. Proofread carefully at final stage. In the end give outline to your arguments. Spot out perspectives of further study of this subject. Justify your conclusion by at the bottom of them with sufficient justifications and examples.

**34. After conclusion:** Once you have concluded your research, the next most important step is to present your findings. Presentation is extremely important as it is the definite medium though which your research is going to be in print to the rest of the crowd. Care should be taken to categorize your thoughts well and present them in a logical and neat manner. A good quality research paper format is essential because it serves to highlight your research paper and bring to light all necessary aspects in your research.

#### INFORMAL GUIDELINES OF RESEARCH PAPER WRITING

#### Key points to remember:

- Submit all work in its final form.
- Write your paper in the form, which is presented in the guidelines using the template.
- Please note the criterion for grading the final paper by peer-reviewers.

#### **Final Points:**

A purpose of organizing a research paper is to let people to interpret your effort selectively. The journal requires the following sections, submitted in the order listed, each section to start on a new page.

The introduction will be compiled from reference matter and will reflect the design processes or outline of basis that direct you to make study. As you will carry out the process of study, the method and process section will be constructed as like that. The result segment will show related statistics in nearly sequential order and will direct the reviewers next to the similar intellectual paths throughout the data that you took to carry out your study. The discussion section will provide understanding of the data and projections as to the implication of the results. The use of good quality references all through the paper will give the effort trustworthiness by representing an alertness of prior workings.

Writing a research paper is not an easy job no matter how trouble-free the actual research or concept. Practice, excellent preparation, and controlled record keeping are the only means to make straightforward the progression.

#### General style:

Specific editorial column necessities for compliance of a manuscript will always take over from directions in these general guidelines.

To make a paper clear

· Adhere to recommended page limits

Mistakes to evade

- Insertion a title at the foot of a page with the subsequent text on the next page
- Separating a table/chart or figure impound each figure/table to a single page
- Submitting a manuscript with pages out of sequence

In every sections of your document

- · Use standard writing style including articles ("a", "the," etc.)
- $\cdot$  Keep on paying attention on the research topic of the paper
- $\cdot$  Use paragraphs to split each significant point (excluding for the abstract)
- · Align the primary line of each section
- · Present your points in sound order
- $\cdot$  Use present tense to report well accepted
- · Use past tense to describe specific results
- · Shun familiar wording, don't address the reviewer directly, and don't use slang, slang language, or superlatives
- · Shun use of extra pictures include only those figures essential to presenting results

#### Title Page:

Choose a revealing title. It should be short. It should not have non-standard acronyms or abbreviations. It should not exceed two printed lines. It should include the name(s) and address (es) of all authors.

#### Abstract:

The summary should be two hundred words or less. It should briefly and clearly explain the key findings reported in the manuscript-must have precise statistics. It should not have abnormal acronyms or abbreviations. It should be logical in itself. Shun citing references at this point.

An abstract is a brief distinct paragraph summary of finished work or work in development. In a minute or less a reviewer can be taught the foundation behind the study, common approach to the problem, relevant results, and significant conclusions or new questions.

Write your summary when your paper is completed because how can you write the summary of anything which is not yet written? Wealth of terminology is very essential in abstract. Yet, use comprehensive sentences and do not let go readability for briefness. You can maintain it succinct by phrasing sentences so that they provide more than lone rationale. The author can at this moment go straight to shortening the outcome. Sum up the study, with the subsequent elements in any summary. Try to maintain the initial two items to no more than one ruling each.

- Reason of the study theory, overall issue, purpose
- Fundamental goal
- To the point depiction of the research
- Consequences, including <u>definite statistics</u> if the consequences are quantitative in nature, account quantitative data; results of any numerical analysis should be reported
- Significant conclusions or questions that track from the research(es)

#### Approach:

- Single section, and succinct
- As a outline of job done, it is always written in past tense
- A conceptual should situate on its own, and not submit to any other part of the paper such as a form or table
- Center on shortening results bound background information to a verdict or two, if completely necessary
- What you account in an conceptual must be regular with what you reported in the manuscript
- Exact spelling, clearness of sentences and phrases, and appropriate reporting of quantities (proper units, important statistics) are just as significant in an abstract as they are anywhere else

#### Introduction:

The **Introduction** should "introduce" the manuscript. The reviewer should be presented with sufficient background information to be capable to comprehend and calculate the purpose of your study without having to submit to other works. The basis for the study should be offered. Give most important references but shun difficult to make a comprehensive appraisal of the topic. In the introduction, describe the problem visibly. If the problem is not acknowledged in a logical, reasonable way, the reviewer will have no attention in your result. Speak in common terms about techniques used to explain the problem, if needed, but do not present any particulars about the protocols here. Following approach can create a valuable beginning:

- Explain the value (significance) of the study
- Shield the model why did you employ this particular system or method? What is its compensation? You strength remark on its appropriateness from a abstract point of vision as well as point out sensible reasons for using it.
- Present a justification. Status your particular theory (es) or aim(s), and describe the logic that led you to choose them.
- Very for a short time explain the tentative propose and how it skilled the declared objectives.

#### Approach:

- Use past tense except for when referring to recognized facts. After all, the manuscript will be submitted after the entire job is done.
- Sort out your thoughts; manufacture one key point with every section. If you make the four points listed above, you will need a least of four paragraphs.

- Present surroundings information only as desirable in order hold up a situation. The reviewer does not desire to read the whole thing you know about a topic.
- Shape the theory/purpose specifically do not take a broad view.
- As always, give awareness to spelling, simplicity and correctness of sentences and phrases.

#### Procedures (Methods and Materials):

This part is supposed to be the easiest to carve if you have good skills. A sound written Procedures segment allows a capable scientist to replacement your results. Present precise information about your supplies. The suppliers and clarity of reagents can be helpful bits of information. Present methods in sequential order but linked methodologies can be grouped as a segment. Be concise when relating the protocols. Attempt for the least amount of information that would permit another capable scientist to spare your outcome but be cautious that vital information is integrated. The use of subheadings is suggested and ought to be synchronized with the results section. When a technique is used that has been well described in another object, mention the specific item describing a way but draw the basic principle while stating the situation. The purpose is to text all particular resources and broad procedures, so that another person may use some or all of the methods in one more study or referee the scientific value of your work. It is not to be a step by step report of the whole thing you did, nor is a methods section a set of orders.

#### Materials:

- Explain materials individually only if the study is so complex that it saves liberty this way.
- Embrace particular materials, and any tools or provisions that are not frequently found in laboratories.
- Do not take in frequently found.
- If use of a definite type of tools.
- Materials may be reported in a part section or else they may be recognized along with your measures.

#### Methods:

- Report the method (not particulars of each process that engaged the same methodology)
- Describe the method entirely
- To be succinct, present methods under headings dedicated to specific dealings or groups of measures
- Simplify details how procedures were completed not how they were exclusively performed on a particular day.
- If well known procedures were used, account the procedure by name, possibly with reference, and that's all.

#### Approach:

- It is embarrassed or not possible to use vigorous voice when documenting methods with no using first person, which would focus the reviewer's interest on the researcher rather than the job. As a result when script up the methods most authors use third person passive voice.
- Use standard style in this and in every other part of the paper avoid familiar lists, and use full sentences.

#### What to keep away from

- Resources and methods are not a set of information.
- Skip all descriptive information and surroundings save it for the argument.
- Leave out information that is immaterial to a third party.

#### **Results:**

The principle of a results segment is to present and demonstrate your conclusion. Create this part a entirely objective details of the outcome, and save all understanding for the discussion.

The page length of this segment is set by the sum and types of data to be reported. Carry on to be to the point, by means of statistics and tables, if suitable, to present consequences most efficiently. You must obviously differentiate material that would usually be incorporated in a study editorial from any unprocessed data or additional appendix matter that would not be available. In fact, such matter should not be submitted at all except requested by the instructor.



Content

- Sum up your conclusion in text and demonstrate them, if suitable, with figures and tables.
- In manuscript, explain each of your consequences, point the reader to remarks that are most appropriate.
- Present a background, such as by describing the question that was addressed by creation an exacting study.
- Explain results of control experiments and comprise remarks that are not accessible in a prescribed figure or table, if appropriate.

• Examine your data, then prepare the analyzed (transformed) data in the form of a figure (graph), table, or in manuscript form. What to stay away from

- Do not discuss or infer your outcome, report surroundings information, or try to explain anything.
- Not at all, take in raw data or intermediate calculations in a research manuscript.
- Do not present the similar data more than once.
- Manuscript should complement any figures or tables, not duplicate the identical information.
- Never confuse figures with tables there is a difference.

#### Approach

- As forever, use past tense when you submit to your results, and put the whole thing in a reasonable order.
- Put figures and tables, appropriately numbered, in order at the end of the report
- If you desire, you may place your figures and tables properly within the text of your results part.

#### Figures and tables

- If you put figures and tables at the end of the details, make certain that they are visibly distinguished from any attach appendix materials, such as raw facts
- Despite of position, each figure must be numbered one after the other and complete with subtitle
- In spite of position, each table must be titled, numbered one after the other and complete with heading
- All figure and table must be adequately complete that it could situate on its own, divide from text

#### Discussion:

The Discussion is expected the trickiest segment to write and describe. A lot of papers submitted for journal are discarded based on problems with the Discussion. There is no head of state for how long a argument should be. Position your understanding of the outcome visibly to lead the reviewer through your conclusions, and then finish the paper with a summing up of the implication of the study. The purpose here is to offer an understanding of your results and hold up for all of your conclusions, using facts from your research and accepted information, if suitable. The implication of result should be visibly described. generally Infer your data in the conversation in suitable depth. This means that when you clarify an observable fact you must explain mechanisms that may account for the observation. If your results vary from your prospect, make clear why that may have happened. If your results agree, then explain the theory that the proof supported. It is never suitable to just state that the data approved with prospect, and let it drop at that.

- Make a decision if each premise is supported, discarded, or if you cannot make a conclusion with assurance. Do not just dismiss a study or part of a study as "uncertain."
- Research papers are not acknowledged if the work is imperfect. Draw what conclusions you can based upon the results that you have, and take care of the study as a finished work
- You may propose future guidelines, such as how the experiment might be personalized to accomplish a new idea.
- Give details all of your remarks as much as possible, focus on mechanisms.
- Make a decision if the tentative design sufficiently addressed the theory, and whether or not it was correctly restricted.
- Try to present substitute explanations if sensible alternatives be present.
- One research will not counter an overall question, so maintain the large picture in mind, where do you go next? The best studies unlock new avenues of study. What questions remain?
- Recommendations for detailed papers will offer supplementary suggestions.

#### Approach:

- When you refer to information, differentiate data generated by your own studies from available information
- Submit to work done by specific persons (including you) in past tense.
- Submit to generally acknowledged facts and main beliefs in present tense.

#### THE ADMINISTRATION RULES

Please carefully note down following rules and regulation before submitting your Research Paper to Global Journals Inc. (US):

Segment Draft and Final Research Paper: You have to strictly follow the template of research paper. If it is not done your paper may get rejected.

- The **major constraint** is that you must independently make all content, tables, graphs, and facts that are offered in the paper. You must write each part of the paper wholly on your own. The Peer-reviewers need to identify your own perceptive of the concepts in your own terms. NEVER extract straight from any foundation, and never rephrase someone else's analysis.
- Do not give permission to anyone else to "PROOFREAD" your manuscript.
- Methods to avoid Plagiarism is applied by us on every paper, if found guilty, you will be blacklisted by all of our collaborated research groups, your institution will be informed for this and strict legal actions will be taken immediately.)
- To guard yourself and others from possible illegal use please do not permit anyone right to use to your paper and files.

## CRITERION FOR GRADING A RESEARCH PAPER (COMPILATION) BY GLOBAL JOURNALS INC. (US)

Please note that following table is only a Grading of "Paper Compilation" and not on "Performed/Stated Research" whose grading solely depends on Individual Assigned Peer Reviewer and Editorial Board Member. These can be available only on request and after decision of Paper. This report will be the property of Global Journals Inc. (US).

Topics	Grades		
	A-B	C-D	E-F
Abstract	Clear and concise with appropriate content, Correct format. 200 words or below	Unclear summary and no specific data, Incorrect form Above 200 words	No specific data with ambiguous information Above 250 words
Introduction	Containing all background details with clear goal and appropriate details, flow specification, no grammar and spelling mistake, well organized sentence and paragraph, reference cited	Unclear and confusing data, appropriate format, grammar and spelling errors with unorganized matter	Out of place depth and content, hazy format
Methods and Procedures	Clear and to the point with well arranged paragraph, precision and accuracy of facts and figures, well organized subheads	Difficult to comprehend with embarrassed text, too much explanation but completed	Incorrect and unorganized structure with hazy meaning
Result	Well organized, Clear and specific, Correct units with precision, correct data, well structuring of paragraph, no grammar and spelling mistake	Complete and embarrassed text, difficult to comprehend	Irregular format with wrong facts and figures
Discussion	Well organized, meaningful specification, sound conclusion, logical and concise explanation, highly structured paragraph reference cited	Wordy, unclear conclusion, spurious	Conclusion is not cited, unorganized, difficult to comprehend
References	Complete and correct format, well organized	Beside the point, Incomplete	Wrong format and structuring

## INDEX

## Α

 $\begin{array}{l} Acquisition \cdot 11, 12, 14 \\ Athanasoglou \cdot 46, 49, 50 \end{array}$ 

## В

Baumeister · 20 Bernstin · 23 Bouzgarrou · 16, 19

## С

Cakici · 2, 9

## D

Dashen · 33, 38, 39 Dawuro · 72, 73, 74

## Ε

Ehrhardt · 43, 63, 69

## F

Figelman · 3, 9

## G

Gapenski · 63 Guerbet · 11, 16 Gurbaxani · 66

## Η

Hetroskedastisty · 48 Hirshleifer · 55

## I

Invesco · 58, 60, 61

## J

Jagric · 54, 62

### L

 $\begin{array}{l} Lakonishok \cdot 1,9 \\ Leuthirch \cdot 20 \\ Lintner \cdot 54 \end{array}$ 

## Μ

Malatesta · 11, 12, 13, 19 Mcguigan · 21, 23, 30 Mehretu · 33, 34, 35, 36, 37, 38 Moskowitz · 1, 2, 3, 9

## Ν

Navatte · 16, 19

### Ρ

Peterkort  $\cdot$  2, 9 Podobnik  $\cdot$  54, 62

## R

Renbao · 45, 46, 50

## S

Sylwester · 46, 50

### Τ

Timotej · 54, 62 Treynors · 54, 56, 57, 58, 59, 60, 61



# Global Journal of Management and Business Research

0

Visit us on the Web at www.GlobalJournals.org | www.JournalofBusiness.Org or email us at helpdesk@globaljournals.org



ISSN 9755853

© Global Journals